

# Magnolia Public Schools

**Regular Board Meeting** 

Date and Time Thursday August 12, 2021 at 6:00 PM PDT

### Location

https://zoom.us/j/93531753860?pwd=TUJDV0FxVmJKd0NqSXNpY0p5WIJYUT09

Meeting ID: 935 3175 3860 Passcode: 383291 One tap mobile: +16699009128,,93531753860# US (San Jose)

All members of the public can participate by calling in using the numbers provided above.

In compliance with the Americans with Disabilities Act (ADA) and upon request, Magnolia Public Schools may furnish reasonable auxiliary aids and services to qualified individuals with disabilities. Members of the public who need special accommodations or translation are strongly encouraged to contact Magnolia Public Schools at least 24 hours in advance of the Board meeting so assistance can be assured.

Any public records relating to an agenda item for an open session which are distributed to all, or a majority of all, of the Board Members shall be available for public inspection. Magnolia Public Schools values public comment during Board meetings. For members of the public who would like to speak, we have speaker cards to be filled out prior to the beginning of the meeting. For teleconference meetings you can email us your request for public comment. We limit individual speakers to three (3) minutes and speakers with interpreters to six (6) minutes. For any questions regarding this meeting email board@magnoliapublicschools.org or call 213-628-3634 Ext. 100.

Board Members: Ms. Sandra Covarrubias, Chair Dr. Umit Yapanel, Vice-Chair Dr. Salih Dikbas Ms. Diane Gonzalez Mr. Mekan Muhammedov

CEO & Superintendent: Mr. Alfredo Rubalcava

# Agenda

	Purpose	Presenter	Time	
I. Opening Items			6:00 PM	
Opening Items				
A. Call the Meeting to Order			1 m	
B. Record Attendance and Guests			1 m	
C. Approval of Agenda	Vote		1 m	
D. Public Comments			5 m	
E. Announcements from CEO & Superintendent and Board Members			5 m	
<b>F.</b> Approval of Minutes from MPS Regular Board Meeting - July 8, 2021	Approve Minutes		1 m	
II. Information/Discussion Items			6:14 PM	
A. Brown Act & Conflict of Interest Training	Discuss	YM&C	90 m	
III. Closed Session			7:44 PM	
A. Public Announcement Of Closed Session	FYI		1 m	
<b>B.</b> Public Employee Performance Evaluation (§ 54957) - Title: CEO & Superintendent			30 m	
C. Report Out Of Closed Session	FYI		1 m	
IV. Action Items			8:16 PM	
A. Approval of CEO Metrics	Vote	M.Muhammedov	5 m	
<b>B.</b> Approval of MSA-1 Mobile Modular Design Contract	Vote	Audit/Facilities Committee	10 m	
<b>C.</b> Approval of MSA-1 Shade Structure Design Contract	Vote	Audit/Facilities Committee	10 m	
<b>D.</b> Approval of Two-Year Extension of Charter School Renewals (Renewal Relief)	Vote	A.Rubalcava	10 m	
<b>E.</b> Approval of Orange County Soccer Club Donation (Mini-Pitch)	Vote	B.Lopez	10 m	
V. Closing Items 9:01 PM				
A. Adjourn Meeting	Vote		1 m	

# **Cover Sheet**

# Announcements from CEO & Superintendent and Board Members

Section: Item: Members Purpose: Submitted by: Related Material: I. Opening Items E. Announcements from CEO & Superintendent and Board

MPS Vision, Mission & Goals.pdf

# Magnolia Public Schools

**Vision:** Graduates of Magnolia Public Schools commit to building a more peaceful and inclusive global society by transforming traditional ideas with creative thinking, effective communication, and the rigor of science.

**Mission:** Magnolia Public Schools provides a safe and nurturing community using a whole-child approach to provide a high-quality, college preparatory STEAM educational experience in an environment that cultivates respect for self and others.

Additional data sources: <u>https://caaspp.cde.ca.gov/sb2017/Search</u> <u>https://www.greatschools.org/</u>

Powered by BoardOnTrack

Magnolia Public Schools - Regular Board Meeting - Agenda - Thursday August 12, 2021 at 6:00 PM

# Magnolia Public Schools 2021-22 Goals

**1.** All students and staff will have access to a safe, secure, healthy, and high-quality learning and working environment. Students will receive high-quality services that are fundamental to academic success, including fully credentialed teachers, standards-aligned and high-quality instructional materials, and appropriately maintained school facilities.

2. All students will have equitable access to a high-quality core curricular and instructional program and make academic progress on the California Content Standards. Students will pursue academic excellence and be college and career-ready.

**3.** All students will have access to a well-rounded education that supports their readiness for college and the global world. Each student will become an independent, innovative scholar by practicing creativity in learning and using technology in transformative ways, and demonstrate high-quality learning outcomes.

**4.** All students, families, staff, and other stakeholders will have access to meaningful engagement opportunities that help cultivate leadership, advocacy, and collaboration in a safe and nurturing environment. Stakeholders will feel a sense of community and connectedness.

Powered by BoardOnTrack

# **Cover Sheet**

# Approval of Minutes from MPS Regular Board Meeting - July 8, 2021

Section: Item:	I. Opening Items F. Approval of Minutes from MPS Regular Board Meeting - July
8, 2021 Purpose: Submitted by:	Approve Minutes
Related Material:	Minutes for Regular Board Meeting on July 8, 2021



# Magnolia Public Schools

# **Minutes**

**Regular Board Meeting** 

# Date and Time Thursday July 8, 2021 at 6:00 PM

APPROVEL

Location https://zoom.us/j/98880640904?pwd=L2I1WGNoamRNRFQ2OHIWdFNhZId5QT09

Meeting ID: 988 8064 0904 Passcode: 494267 One tap mobile: +16699009128,,98880640904# US (San Jose)

Board Members: Ms. Sandra Covarrubias, Chair Dr. Umit Yapanel, Vice-Chair Dr. Salih Dikbas Ms. Diane Gonzalez Mr. Mekan Muhammedov

CEO & Superintendent: Mr. Alfredo Rubalcava

# **Directors Present**

D. Gonzalez (remote), M. Muhammedov (remote), S. Covarrubias (remote), S. Dikbas (remote), U. Yapanel (remote)

# **Directors Absent**

None

# **Directors who arrived after the meeting opened** U. Yapanel

# I. Opening Items

A. Call the Meeting to Order

S. Covarrubias called a meeting of the board of directors of Magnolia Public Schools to order on Thursday Jul 8, 2021 at 6:04 PM.

#### B. Record Attendance and Guests

Refer to attendance information stated above.

### C. Approval of Agenda

S. Covarrubias made a motion to approve the agenda as presented.

M. Muhammedov seconded the motion.

The board **VOTED** to approve the motion.

Roll Call

U. Yapanel Absent

- M. Muhammedov Aye
- S. Dikbas Aye
- D. Gonzalez Aye
- S. Covarrubias Aye

# **D.** Public Comments

No public comment was given at this time.

#### E. Announcements from CEO & Superintendent and Board Members

A.Rubalcava, MPS CEO & Superintendent, updated the Board Members on the MPS schools holding summer school session. He also announced that the Board will be updated on the 2021-22 Budget Act as more information comes. He also announced the passing of Chris Garcia, MPS employee at MSA-7. Words honoring his work ethic and his impact to his students was given and a moment of silence was held. The Board gave their condolences to the Garcia family and asked to be provided information to support his family.

### F. Approval of Minutes from MPS Regular Board Meeting - June 24, 2021

D. Gonzalez made a motion to approve the minutes from Regular Board Meeting on 06-24-21.

M. Muhammedov seconded the motion.

The board **VOTED** to approve the motion.

#### Roll Call

- S. Covarrubias Aye
- D. Gonzalez Aye
- S. Dikbas Aye
- U. Yapanel Absent
- M. Muhammedov Aye

#### **II. Closed Session**

#### A. Public Announcement Of Closed Session

S.Covarrubias, MPS Chair, announced that the Board will be going into Closed Session to discuss public employee performance evaluation (CEO) and public employment and would report out any actions taken.

# B. Public Employee Performance Evaluation (§ 54957) - Title: CEO & Superintendent

This item was discussed in Closed Session. U. Yapanel arrived at 6:17 PM.

### C. Public Employment (§ 54957)

This item was discussed in Closed Session

#### D. Report Out Of Closed Session

S.Covarrubias announced in Open Session at 8:19pm that the Board will further discuss in Open Session at the CEO Evaluation and Contract Renewal action item on the agenda.

# **III. Consent Items**

#### A. Approval of MPS 2021-22 Student/Parent Handbook

D.Yilmaz, MPS Chief Accountability Officer, presented the Student/Parent Handbook. This is an annual item that requires review and approval. He discussed the minor changes and modifications on the Handbook based on feedback from stakeholders and was reviewed by legal counsel.

S. Dikbas made a motion to approve the MPS 2021-22 Student/Parent Handbook.

M. Muhammedov seconded the motion.

The board **VOTED** unanimously to approve the motion.

Roll Call

- M. Muhammedov Aye
- D. Gonzalez Aye
- S. Covarrubias Aye
- U. Yapanel Aye
- S. Dikbas Aye

### **IV. Action Items**

# A. Approval of Updated MPS Health and Safety Policy and Injury and Illness Prevention Program ("IIPP") COVID-19 Addendum

D.Hajmeirza, MPS Human Resources Director, provided updated information on the MPS Health & Safety Policy regarding COVID-19 testing. Legal counsel reviewed the policy. This will be brought to the Board again in August.

D. Gonzalez made a motion to approve the updated MPS Health & Safety Policy alongside the IIPP COVID-19 addendum.

M. Muhammedov seconded the motion.

The board **VOTED** unanimously to approve the motion.

Roll Call

- D. Gonzalez Aye
- S. Dikbas Aye
- S. Covarrubias Aye
- M. Muhammedov Aye
- U. Yapanel Aye

### B. Approval of Teacher Assignment Option: Ed Code 44258.3

D.Hajmeirza, MPS Human Resources Director, provided context regarding the teacher assignment option: Ed Code 44258.3. Traditionally, charter schools did not use Ed Code options, but with the credentialing rule changes and the current shortage of teachers, it has been advised by charter school authorizers to start using Ed Code options as an alternative to assigning teachers to subject areas without them having to earn another credential authorization. Teachers must already have a Preliminary or Clear credential and have completed a teacher preparation program including student teaching to be eligible for this education code. Staff addressed all Board Members questions.

S. Covarrubias made a motion to approve the teacher assignment option: Ed Code 44258.3.

U. Yapanel seconded the motion.

The board **VOTED** unanimously to approve the motion.

- Roll Call
- D. Gonzalez Aye
- S. Covarrubias Aye
- U. Yapanel Aye
- S. Dikbas Aye M. Muhammedov Aye
- Ni. Muhammedov Aye

# C. Approval of Teacher Assignment Option: Ed Code 44258.7(c)(d)

D.Hajmeirza, MPS Human Resources Director, presented Ed Code 44258.7 (c)(d) which allows flexibility for teacher assignment-Elective Classes. She provided context of the specifications of the code and its parameters.

D. Gonzalez made a motion to approve the teacher assignment option: Ed Code 44258.7(c)(d).

S. Covarrubias seconded the motion.

The board **VOTED** unanimously to approve the motion.

Roll Call

- D. Gonzalez Aye
- S. Dikbas Aye
- U. Yapanel Aye
- S. Covarrubias Aye
- M. Muhammedov Aye

# D. Approval of CTC Form CL-500 Declaration of Need (Emergency Permits)

D.Hajmeirza, MPS Human Resources Directors, presented this item. Public schools within California must complete and submit to the CTC the CL-500 form which serves the purpose for schools to anticipate how many emergency permits they will need to cover teacher assignments for each school year. This is on an annual basis. MPS will fill out the form for each of three counties MPS serves (Santa Ana, Los Angeles and San Diego).

D. Gonzalez made a motion to approve the Commission on Teaching Credential ("CTC") Form CL-500 Declaration of Need.

M. Muhammedov seconded the motion.

The board **VOTED** unanimously to approve the motion.

- Roll Call
- S. Dikbas Aye
- M. Muhammedov Aye
- S. Covarrubias Aye
- U. Yapanel Aye
- D. Gonzalez Aye

# E. Adoption of CEO Comparable Compensation Survey

M.Muhammedov, MPS Board Member stated, " As to the next item regarding approval of the CEO/Superintendent's employment contract, I am required by law to indicate, prior to any vote on the contract, that the CEO/Superintendent's contract provides the following: 1. The contract is effective July 1, 2021 and provides an annual salary of **\$236,500**. 2. Standard health benefits as are provided to certificated employees. I would also add that the annual salary is wholly in keeping with the compensation comparability study the Board has received. I am pleased to recommend approval of this contract and call for a vote on it."

M. Muhammedov made a motion to approve the compensation comparability study for the CEO & Superintendent position of Magnolia Public Schools. S. Covarrubias seconded the motion.

The board **VOTED** unanimously to approve the motion.

#### Roll Call

- D. Gonzalez Aye S. Dikbas Aye M. Muhammedov Aye S. Covarrubias Aye U. Yapanel Aye
- F. Approval of CEO Metrics, Evaluation Measures, and Contract Renewal

M. Muhammedov, MPS Board Member, informed that the metrics would be taken for approval at the next regularly scheduled Board Meeting and will move forward with voting on the CEO contract for this meeting.

M. Muhammedov made a motion to approve the CEO & Superintendent contract effective July 1, 2021 and will approve the CEO Metrics at a later Board Meeting. S. Covarrubias seconded the motion.

The board **VOTED** unanimously to approve the motion.

Roll Call

- M. Muhammedov Aye
- D. Gonzalez Aye
- S. Dikbas Aye
- U. Yapanel Aye
- S. Covarrubias Aye

### V. Closing Items

### A. Adjourn Meeting

There being no further business to be transacted, and upon motion duly made, seconded and approved, the meeting was adjourned at 8:48 PM.

Respectfully Submitted, S. Covarrubias

# **Cover Sheet**

# Approval of CEO Metrics

Section: Item: Purpose: Submitted by: Related Material: IV. Action Items A. Approval of CEO Metrics Vote

2021-22 CEO Metrics 8.12.pdf



# CHIEF EXECUTIVE OFFICER AND SUPERINTENDENT JOB DESCRIPTION 2021-2022

# Job Summary:

The Chief Executive Officer ("CEO") and Superintendent is responsible for the leadership, strategic vision and growth of Magnolia Public Schools (MPS) and future Magnolia schools, as well as assisting the Board of Directors ("Board") in fulfilling its responsibilities. The CEO/Superintendent provides daily operational oversight and guidance with regard to safety, academic achievement, fiscal integrity and compliance with all laws and regulations.

The primary responsibility of the CEO/Superintendent is to carry out the strategic plans and policies as established by the Board, including the academic performance and operations of the existing schools and future campuses and fiscal oversight. The CEO/Superintendent reports to the Board.

For the 2021-2022 school year, the two most critical areas are: (1) managing school operations and openings post Covid-19 crisis, (2) Improving academic achievement, meeting and exceeding grade level standards

# Job Duties:

- Provide the leadership, vision, and strategic direction for the school(s);
- Structure the organization of the school(s) to achieve the vision, philosophy and mission;
- Oversee all operations of the school(s) and report to the Board on their progress;
- Assist the Board in the development of governance policies for review and approval;
- Negotiate, on behalf of the Board when duly authorized to do so, all vendor and service contracts, orders, licenses, or other agreements of a special nature unless the signing is expressly limited by the Board;
- Oversee all aspects of the school(s) including personnel, financial matters, the academic program, facilities, and operations;

- Work with the Board Chair to draft the agenda for all Board meetings;
- Recommend an annual budget to the Board, including an accounting of federallyfunded programs as required by applicable guidelines;
- Hire and oversee performance of charter home office staff, and principals whose jobs are to support/lead the operations, academic performance, business, marketing and development work;
- Oversee hiring, supervision, professional development, evaluation and dismissal of all personnel at school(s) and the CMO;
- Oversee the allocation of resources and school model decisions to drive the best balance between financial, operational, and academic concerns;
- Develop an overall human capital strategy and establish career paths within the CMO;
- Build a high performing, consistent data-driven culture at the CMO at all levels of its operations;
- Drive increased performance in the current school(s) and further refine the academic model;
- Determine the scope and sequence of educational programs according to the charter objectives;
- Implement well-researched, creative ways to educate the students in the school(s);
- Serve as primary contact with the State Department of Education;
- Follow all legal mandates from the U.S. Department of Education and the State of California in all aspects related to funding, reporting and regulations associated with charter schools;
- Work with outside counsel to support the legal needs of the organization;
- Assist the Board in seeking donors of time, funds and resources in support of the school(s);
- Manage any required facility acquisition, maintenance and renovation efforts;
- Perform other duties and responsibilities as are customary for the office of the CEO/Superintendent and as may be assigned from time to time by the Board.

# Appendix B -

# Academic Accountability:

- Return to success full instruction and pivot post COVID opening of MPS Schools;
- 100% Four-Year Cohort Graduation Rate and 100% UC/CSU A-G Course Completion Rate across all MPS;

- All student groups across MPS will show growth on the CAASPP-ELA/Literacy and Mathematics assessments by a 3-point or more increase from the prior year as measured by the CA School Dashboard;
- 50% or more of all MPS students will be proficient on the CAASPP-ELA/Literacy assessments and 35% or more will be proficient on the CAASPP-Mathematics assessments, with the ultimate goal of 100% proficiency for all;
- 60% or more of all students will meet their growth targets on the MAP-Reading and Mathematics assessments, with the ultimate goal of 100%;
- All students across MPS will meet their expected growth targets in Reading as measured by the Lexile metrics;
- All MPS sites will organize a STEAM Festival/EXPO, with 100% of students creating and demonstrating a STEAM focused project, experiment, model or demo;
- Average Daily Attendance (ADA) Rate of 97% or more, with the ultimate goal of 100% ADA Rate;
- MPS will develop, align, and improve academic and behavioral resources, programs, supports, and services in addressing the needs of the whole child by utilizing a coherent MTSS framework that engages all systems leading to improved student outcomes;
- School experience survey participation and satisfaction rates of 85% or more for students, parents, teachers, and staff, with the ultimate goal of 100% participation and satisfaction rates;
- All MPS sites will maintain their WASC accreditation;
- All MPS sites will have successful authorizer oversight visits;
- Keep strengthening MPS' standards-based curriculum complete with scope and sequences for all subjects taught at every existing and expected grade level that exceeds national academic standards.

# Personnel and Human Resources (HR):

- Clear and detailed strategy for teacher recruitment and staff retention.
- Expand the salary scale to include a performance-based evaluation for home office staff.
- Delegation of authority the executive staff are required to perform their duties effectively in support of the overall success of the organization. Magnolia Public Schools (MPS) Executive staff will have current job descriptions that include evaluation metrics and the frequency of a performance evaluation. The performance

evaluation will include the Board recommendation for a 6-month evaluation and a year-end evaluation to determine if the performance is acceptable.

- All the personnel paperwork documentation required by the authorizers, state and federal agencies will be ready before the first day of employment for every school year for every staff member. Staff employment paperwork which must be provided within a certain time frame, mandatory training which must be completed within a certain deadline will be completed by each staff member before respective deadlines. Human Resources will continue assisting school sites with authorizer site visits.
- MPS Employee Handbook and MPS Employee Agreements will be revised every year with respect to evolving needs and changing Labor Codes and Laws as well as post Covid-19 changes that need to be made.
- HR will enhance the onboarding process to give extra support to the School Site Leaders.
- HR will provide more features of PayCom (HRIS online software system) to provide more options and keep MPS HR procedures aligned across all school sites.

# Management & Board Oversight:

- Clear, transparent and timely communication between MPS Home Office and MPS Board with timely reporting of any critical development and communication with other local agencies to the board;
- All committees are to meet each academic year according to a schedule set at the beginning of the academic year. Board members are to conduct committee meetings directly with responsible C-team members. All C-team members should present items directly to corresponding committee members in their respective areas with the CEO serving as facilitator of the discussion;
- In conjunction with the Board, continue to develop an effective public relation, marketing, and fundraising plan for 2021-22;
- Secure philanthropic support that directly supports the mission and vision for 2021-22. Report annually to the board on C-Team and school site performance;
- Train and support new board members appointed due to vacancies and in compliance with the terms of authorizer directions;
- Passing reports from authorizer site visits.

# Finance:

- Manage post Covid-19 financial uncertainty, utilize one-time state and federal funds effectively;
- Justify and explain any budget fluctuations over 3% from the June budget. Remain within 3% of the Pl revised budget through the conclusion of the fiscal year;
- Secure private financing for obtaining private facilities for all of the co-located sites;
- Positive net income for combined all schools and Home Office at the end of the year;
- Improving compensation and support for school site staff;
- Unqualified independent audit report;
- Update the five-year capital plan following the bond issuance;
- Continue to provide leadership in the areas of financial planning, budgeting, accounting and management of the organization's financial resources;
- Maintain accurate accounting, payroll, cash management, and insurance systems are maintained.

# Facilities:

- Prepare a strategic plan for all schools at Prop 39 facilities;
- Support all schools in the process of acquiring their own facilities especially concentrate on MSA-5 securing private facility;
- Completion of capital improvement projects funded by grant;
- Implement asset management system to make sure that preventative maintenance action is taken to avoid large unplanned facilities/capital expenditures;
- Oversee, monitor, and maintain current requirements of facility leases;
- Apply for facility grants such as SB 740 and CSFIG;
- Continue overseeing that all physical environment reflects the organization's mission and values as well as enhances learning. Improve fundraising and philanthropic efforts to improve/invest in facilities.

# Stakeholders:

- Increase enrollment and improve efficiency of enrollment process. As well as strengthen enrollment campaign in conjunction with school sites;
- Extensively engage all MPS stakeholders such as parents, students, teachers and the local community in crafting the strategies, implementing academic policies and

engaging in civic participation. School site leaders or MPS Central Office should address any concerns by these stakeholders in a timely manner;

- Continue to strengthen the MPS parent programs across all schools;
- Develop Partnerships to bring in more resources and utilize strategic planning for philanthropic partnerships;
- Optimize resources to the fullest extent to achieve organization efficacy at highest level.

# **Cover Sheet**

# Approval of MSA-1 Mobile Modular Design Contract

Section:IV. Action ItemsItem:B. Approval of MSA-1 Mobile Modular Design ContractPurpose:VoteSubmitted by:Related Material:Design Contract for Modular Restrooms (MSA-1) Board Report.pdf



Agenda Item #:	IV B: Action Item
Date:	August 12, 2021
To:	Magnolia Educational & Research Foundation dba Magnolia Public Schools (" <u>MPS</u> ") Board of Directors
From:	Alfredo Rubalcava, CEO & Superintendent
Staff Lead:	MPS Audit/Facilities Committee
RE:	Approval of MSA-1 Design Contract for Modular Restroom Project

# I. Proposed Board Motion

Staff recommends and moves that the Board approve the selection of Mobile Modular for the design of the modular restroom that is a part of the Magnolia Science Academy—1 ("<u>MSA-1</u>") middle school parking lot site improvement project (the "<u>Project</u>") and the related design contract for \$34,500 with a 20% contingency of \$6,900 for a total fee of \$41,400, attached as <u>Exhibit A</u>, with such changes as MPS Staff may deem advisable or in the best interests of MPS.

# II. Background

MSA-1 filed for and received a zone change for its parking lot parcels from the City of Los Angeles. As a condition to the zone change, MSA-1 agreed to make certain improvements to the parking lot parcel that serves the MSA-1 middle school building. See <u>Exhibit B</u>. They include the following: (1) the fixing of drainage on the middle school parking lot; (2) the creation of an open green space and a concrete pad with a solar shade; (3) fencing surrounding the entire MSA-1 parking lot parcels; (4) the creation of a track; and (5) the installation of modular outdoor restrooms. MSA-1 wishes to undertake these improvements because it will assist the school during the current pandemic to allow for activities to take place outside. Looking to the future, having a good recreation and outdoor space will benefit the students and may in fact attract more students to the campus.

# III. Procurement

Franco Architects worked with Staff to solicit bids for the modular restroom. Staff prepared a Request for Proposals and published it on the MPS website. See <u>Exhibit C</u>. Franco Architects targeted specific vendors to solicit proposals—including, Silver Creek, Williams Scottsman, and Enviroplex. After several rounds of discussions, only Enviroplex remained interested in the Project. However, they referred the team to their sister company Mobile Modular since Enviroplex's focus is Division of State Architect projects. Concurrently, Staff priced out the cost of a custom designed and built restroom. A custom design would require a change order to the Franco Architects contract. Note the language from the March 11, 2021 board report – "[f]or all proposals, all assume that the solar shade structure and the restroom structure will be

210812 - MPS - Report to Facilities Committee re Design Contract for Modular Restrooms

August 9, 2021 Page 2



"design build" or "modular", meaning that a vendor will be identified and they will be responsible for designing and permitting the structure. Each design firm will assist in identifying vendors to provide these services." We further sought input from a general contractor on the cost of building a custom designed restroom. The cost is difficult to determine with the cost escalations in the construction industry. Staff believes that the Mobile Modular design is both attractive (see <u>Exhibit D</u>) and functional and will allow MPS to lock in prices early. The contract fee of \$34,500 will be credited against the final installation price which is currently priced at \$446,670 (See <u>Exhibit E</u> proposal).

# IV. Conclusion & Budget Impact

Staff recommends that MPS sign a design contract for \$34,500 with Mobile Modular for the modular restroom that is part of the Project. Staff further recommends a contingency of \$6,900 (that is, 20% of the contract fee) be approved, for a total of \$41,400.

Staff intends to seek financing for the Project. Until such financing is secured, MSA-1 may need to pay for any expenses, for example, progress payments, out of its operating budget or its reserves. Such expenditure will not have a material impact on either MSA-1's operating budget or its reserves.

Exhibit A	_Mobile Modular Design Contract	.Pg. 3
Exhibit B	Site Improvement Site Plan	Pg. 5
Exhibit C	RFP	
Exhibit D	Mobile Modular Restroom Design	
Exhibit E	Mobile Modular Proposal	Pg. 82



# Exhibit A

# Mobile Modular Design Contract



**MAGNOLIA SCIENCE ACADEMY 1 - MASTERPLANNING** 









23 of 258



# Exhibit B

Site Improvement Site Plan











Exhibit C

RFP



# MAGNOLIA PUBLIC SCHOOLS

Request for Qualifications/Proposals to Provide a Restroom Modular Structure for Magnolia Science Academy 1 | Reseda

> Due Date: May 7, 2021

1

# 1.0 INTRODUCTION

Magnolia Education & Research Foundation doing-business as Magnolia Public Schools ("**MPS**"), a charter school management organization, operates Magnolia Science Academy 1 ("**MSA-1**") located at 18220 Sherman Way and 18238 Sherman Way in the Reseda neighborhood of Los Angeles. The former houses MSA-1's high school and the latter houses MSA-1's middle school. The purpose of this RFP is to solicit bids to design and build a structure consisting of restrooms and changing rooms (the "**Project**") in the existing surface parking lot belonging to the middle school (the "**Project Site**"). The Project is part of a larger site improvement plan for the middle school parking lot wherein recreation and open space will be developed (the "**Site Improvement Plan**"). The location of the Project Site is shown on <u>Exhibit A</u>. The overall Site Improvement Plan is shown on <u>Exhibit B</u>.

Franco Architects is the architect of record for the Site Improvement Plan. The successful respondent will be required to coordinate their efforts with them and the rest of the design team.

Site Tour

A site tour will be facilitated if requested.

Proposals Due

Responses to the RFP are due no later than 5:00 PM (PST), Friday, May 7, 2021, to the following individual:

Patrick Ontiveros Director of Facilities Magnolia Public Schools 250 East 1<sup>st</sup> Street Suite 1500 Los Angeles, CA 90012 pontiveros@magnoliapublicschools.org (213) 628-3634 x 103

Questions regarding this RFP may be directed to the individual identified above via email.

Proposal Format:

One (1) electronic PDF copy (by email) of your proposal must be delivered to the person indicated by the deadline stated above and one (1) hard copy must be delivered to the above address no later than two (2) business days thereafter. Please endeavor to keep any emailed material to a single manageable file size (at or about 10 MBs) so that it may be easily distributed to the Selection Committee.

Respondents are encouraged to only include information pertinent to the Project and the Selection Committee's ability to select the vendor best suited to successfully complete this job.

#### Interviews:

Interviews will be held at the discretion of MPS and MSA-1. Interviews, if any, are expected to be held during the week of May 10th. Location and time are to be determined.

If interviews are held, respondents will meet with the Selection Committee for approximately ½ hour to 1 hour. The interview will be an open, unscripted format. Respondents will have 20 minutes to present any information they feel is pertinent followed by questions and answers and general discussion.

#### Magnolia Public Schools

### Selection Committee:

The Selection Committee will be composed of representatives from MPS, MSA-1, and the Architect.

## 1.1 Timeline

RFP Distributed:	April 23, 2021
Proposals Due:	May 7, 2021
Interviews, if any (exact date and time TBD):	May 10, 11 or 12, 2021
Selection Announced:	May 13, 2021
Contract Execution:	Week of May 17, 2021

## 2.0 PROJECT DESCRIPTION

The Project is the design and construction of a restroom and changing room structure that will support outdoor recreation on the Project Site. The Site Improvement Plan, including the Project, was part of a zone change application that was approved the City of Los Angeles. Conditions of the zone change approval are contained in <u>Exhibit C</u> attached hereto. Respondent is responsible for reviewing the conditions of approval to insure their Project proposal meets the therein stated requirements. <u>Exhibit B-1</u> is a depiction of the Site Improvement Plan, including the Project. <u>Exhibit B-2</u> illustrates different placement of the Project on the Project Site.

A soils report for the immediately adjacent surface parking lot parcel is attached as Exhibit D.

### 3.0 PROPOSAL FORMAT

Respondent shall format its response as set forth below to facilitate timely review and selection. Please be specific to the RFP, and do not include materials not explicitly requested, such as generic marketing materials.

### Your response should include the following:

- Letter of interest
- Name of your company and the individual responsible for the account
- Restate all the requirements of Section 4.0 and provide responses to each

See Section 1.0 for additional proposal format clarifications.

### 4.0 PROPOSAL REQUIREMENTS

# 4.1 Vendor Qualifications and Experience

### 4.1.1 Vendor Description.

Provide a description of your company and why it is qualified to undertake the Project.

## Provide the following:

A minimum of three (3) references, including

- (a) name and scope of the project
- (b) client name and contact information
- (c) contract amount

# 4.1.2 Qualifications and Experience of Key Personnel.

Identify the person(s) that will be principally responsible for working with the MPS and leading this engagement and their qualifications and experience

## 4.1.3 Insurance.

Provide a description of vendor's insurance coverage.

# 4.2 Cost

Respondent's proposal should include an overall not to exceed cost and should be broken down in detail. The proposal should also provide a break-down of any and all other costs and fees including, but not limited to, labor, delivery fees, installation fees, applicable taxes, etc.

## 4.3 **Product Information**

Respondent will confirm its ability to provide the Project as described above in Section 2.2. Where a respondent believes there are multiple options for the Project, the respondent may provide such alternatives explaining the difference in each alternative and describing the cost differences.

# 4.4 Warranty Information

Provide a description of the warranty provided for the Project.

### 4.5 Schedule

MPS and MSA-1 desire to complete construction of the Project and the Site Improvement Plan in time for the 2022-23 school year. Respondents shall provide an affirmative statement of its ability to collaborate with the Architect to submit plans to the Los Angeles Department of Building and Safety ("LADBS") in order to meet the desired delivery date. Respondent shall also provide an affirmative statement describing the construction duration for the Project upon receipt of permits.

Owner and architect believe that the plans for the Project should be ready for submission to LADBS by July 9, 2021 in order to meet the timeline of completing construction by the start of the 2022-23 school year on or about August deadline.

# 4.6 Contract

The successful respondent will be required to sign an agreement with Owner substantially similar to <u>Exhibit</u> <u>E</u> attach hereto. Please provide an affirmative statement of respondent's concurrence or else any changes that respondent desires to make to the form agreement.

# 5.0 CONTACT

Questions to Owner will be accepted via email by the Director of Facilities identified above. Answers to questions will be provided to all participants as available.

## 6.0 RFP/Q EXHIBITS

- Exhibit A: Project Site Location
- Exhibit B: Project
- Exhibit C: Letter of Determination
- **Exhibit D:** Soils Report
- Exhibit E: Form of Contract -- AIA A141-2014 Standard Form of Agreement between Owner and Design-Builder

# 7.0 BID ACCEPTANCE/REJECTION

The Owner reserves the right to reject any or all proposals, or cancel the solicitation process at its sole discretion.

# 8.0 PROPOSAL VALIDITY

RFP responses shall be valid until execution of a contract, which is expected to occur on or about May 2021. No changes to information received within the Respondent's proposal shall be changed or altered without approval by the Owner.

# Exhibit A



# Exhibit B

# Site Improvement Plan



# Exhibit B-2

# Alternative Project Placement






## Exhibit C

# Letter of Determination / Conditions of Approval



SOUTH VALLEY AREA PLANNING COMMISSION

200 North Spring Street, Room 272, Los Angeles, California, 90012-4801, (213) 978-1300 www.planning.lacity.org

### **\*CORRECTED LETTER OF DETERMINATION**

MAILING DATE: APR 2 8 2020

Case No. APCSV-2019-1064-ZC-CDO CEQA: ENV-2019-1065-MND Plan Area: Reseda – West Van Nuys Council District: 3 - Blumenfield

Project Site: 18216 – 18238 West Sherman Way

Applicant: Magnolia Science Academy 1 Representative: Veronica Becerra, Rabuild Commercial Services LLC

At its meeting of **November 14, 2019**, the South Valley Area Planning Commission took the actions below in conjunction with the approval of the following Project:

An outdoor facilities improvement to the Magnolia Science Academy 1 campus. The Project involves the grading of an existing surface parking lot and the construction, use, and maintenance of an open recreation area with a landscaped grass field, a modular restroom and storage facility, and an outdoor covered student activity area with a solar shade structure 25 feet in height. A portion of the parking lot will be re-striped and maintained as a surface parking lot with the addition of landscaping and trees. The Project also includes installation of an eight foot tall concrete masonry unit block wall along the southern property line and a six foot tall security fence along the east and west property lines. The proposed project will not increase the student enrollment capacity of the school.

- 1. **Found**, pursuant to CEQA Guidelines Section 15074(b), after consideration of the whole of the administrative record, including the Mitigated Negative Declaration, No. ENV-2019-1065-MND ("Mitigated Negative Declaration"), and all comments received, with the imposition of the mitigation measures, there is no substantial evidence that the project will have a significant effect on the environment; **Found** the Mitigated Negative Declaration reflects the independent judgment and analysis of the City; **Found** the mitigation measures have been made enforceable conditions on the project; and **Adopt** the Mitigated Negative Declaration Declaration and the Mitigation Monitoring Program prepared for the Mitigated Negative Declaration;
- 2. **Approved** and **recommended** that the City Council adopt, pursuant to Section 12.32 F of the Los Angeles Municipal Code (LAMC), a Zone change from [Q]P-1L-CDO to (T)[Q]C2-1L-CDO to permit the construction and installation of outdoor recreational facilities, a modular restroom a solar shade structure, and landscaping;
- 3. **Approved**, \*<sup>1</sup>pursuant to LAMC Section 13.08 E.3 and the Reseda Central Business District Community Design Overlay, a Design Overlay Plan Approval for the construction,

<sup>&</sup>lt;sup>1</sup> The above-referenced technical correction is not discretionary, therefore, issuance of this corrected letter of determination does not extend or require a new appeal period.

#### APCSV-2019-1064-ZC-CDO

installation, use, and maintenance of a proposed shade structure, modular bathroom, landscaping, and surface parking areas.

- 4. **Adopted** the Conditions of Approval; and
- 5. **Adopted** the Findings.

This action was taken by the following vote:

Moved:	Bishop
Seconded:	Beatty
Ayes:	Mather, Menedjian
Absent:	Dierking

Vote: 4 – 0

Rafael Vega, Commission Executive Assistant

Fiscal Impact Statement: There is no General Fund impact as administrative costs are recovered through fees.

If you seek judicial review of any decision of the City pursuant to California Code of Civil Procedure Section 1094.5, the petition for writ of mandate pursuant to that section must be filed no later than the 90th day following the date on which the City's decision became final pursuant to California Code of Civil Procedure Section 1094.6. There may be other time limits which also affect your ability to seek judicial review.

Attachments: Zone Change Ordinance, Map, Findings

c: JoJo Pewsawang, City Planner Renata Ooms, City Planning Associate ORDINANCE NO. \_\_\_\_\_

An ordinance amending Section 12.04 of the Los Angeles Municipal Code by amending the Zoning map.

THE PEOPLE OF THE CITY OF LOS ANGELES DO ORDAIN AS FOLLOWS:

Section 1. Section 12.04 of the Los Angeles Municipal Code is hereby amended by changing the zone and zone boundaries shown upon a portion of the zone map attached thereto and made a part of Article 2, Chapter 1 of the Los Angeles Municipal Code, so that such portion of the zoning map shall be as follows:



APCSV-2019-1064-ZC-CDO

## [Q] QUALIFIED CLASSIFICATIONS

The subject property shall continue to be subject to the Reseda Central Business District Community Design Overly Permanent [Q] Conditions imposed by Ordinance No. 176558 and adopted on March 1, 2005.

### CONDITIONS FOR EFFECTUATING (T) TENTATIVE CLASSIFICATION REMOVAL

Pursuant to Section 12.32-G of the Municipal Code, the (T) Tentative Classification shall be removed by posting of guarantees through the B-permit process of the City Engineer to secure the following without expense to the City of Los Angeles, with copies of any approval or guarantees provided to the Department of City Planning for attachment to the subject planning case file.

Dedication(s) and Improvement(s). Prior to the issuance of any building permits, the following public improvements and dedications for streets and other rights of way adjoining the subject property shall be guaranteed to the satisfaction of the Bureau of Engineering, Department of Transportation, Fire Department (and other responsible City, regional and federal government agencies, as may be necessary):

#### Responsibilities/Guarantees.

- 1. As part of early consultation, plan review, and/or project permit review, the applicant/developer shall contact the responsible agencies to ensure that any necessary dedications and improvements are specifically acknowledged by the applicant/developer.
- 2. Bureau of Engineering. Prior to issuance of sign offs for final site plan approval and/or project permits by the Department of City Planning, the applicant/developer shall provide written verification to the Department of City Planning from the responsible agency acknowledging the agency's consultation with the applicant/developer. The required dedications and improvements may necessitate redesign of the project. Any changes to project design required by a public agency shall be documented in writing and submitted for review by the Department of City Planning.
  - a. Dedication Required: None
  - b. Improvements Required:
    - i. **Sherman Way** Construct a new full-width concrete sidewalk. Remove and replace bad order concrete curb and gutter along the property frontage.
    - ii. **Alley** Reconstruct the 20-foot wide alley and longitudinal concrete gutter as necessary to provide safe access for students and pedestrian traffic. (Note: The site plan does not show the existing public right-of-way for the Alley turning area. This public right-of-way must be shown on all plans.)

#### Notes:

Broken curb and/or gutter includes segments within existing score lines that are depressed or upraised by more than <sup>1</sup>/<sub>4</sub> inch from the surrounding concrete work or are separated from the main body of the concrete piece by a crack through the entire vertical segment and greater than 1/8 inch at the surface of the section.

Non-ADA compliant sidewalk shall include any sidewalk that has a cross slope that exceeds 2% and/or is depressed or upraised by more than ¼ inch from the surrounding concrete work or has full concrete depth cracks that have separations greater than 1/8

inch at the surface. The sidewalk also includes that portion of the pedestrian path of travel across a driveway.

All new sidewalk curb and gutter shall conform to the Bureau of Engineering Standard Plans S410-2, S410-4, S442-5 and S444-0.

Install tree wells with root barriers and plant street trees satisfactory to the City Engineer and the Urban Forestry Division of the Bureau of Street Services. The applicant should contact the Urban Forestry Division for further information (213) 847-3077.

Street lighting may be required satisfactory to the Bureau of Street Lighting (213) 847-1551.

Department of Transportation may have additional requirements for dedication and improvements.

Regarding any conflicts with power pole matters, contact the Department of Water and Power at (213) 367-2715.

- c. Provide proper site and street drainages for all streets being improved.
- d. Mainline sewer exists in Sherman Way and in the Alley. All Sewerage Facilities Charges and Bonded Sewer Fees are to be paid prior to obtaining a building permit.
- e. Submit a parking area and driveway plan to the Valley District Office of the Bureau of Engineering and the Department of Transportation for review and approval.

#### 3. Department of Transportation.

- a. Access to Sherman Way shall be prohibited. Access to the project site is proposed via the alley.
- b. A driveway apron width of W=30 feet is required for any non-alley access.
- c. A minimum of 20-foot reservoir space is required between any security gate or parking stall and the property line for any non-alley access, to the satisfaction of the Department of Transportation.
- d. A parking area and driveway plan should be submitted to the Citywide Planning Coordination Section of the Department of Transportation for approval prior to submittal of building permit plans for plan check by the Department of Building and Safety. Transportation approvals are conducted at 6262 Van Nuys Blvd., Room 320, Van Nuys, CA 91401.
- e. That permitting fees be paid to the Department of Transportation as required per Ordinance No. 183270 and LAMC Section 19.15 prior to any building permit approval. Note: the applicant may be required to comply with any other applicable fees per this new ordinance.

#### 4. Bureau of Street Lighting

a. IMPROVEMENT CONDITION: No street lighting improvements if no street widening per BOE improvement conditions. Otherwise relocate and upgrade street lights; two (2) on Sherman Way. Notes:

The quantity of street lights identified may be modified slightly during the plan check process based on illumination calculations and equipment selection.

Conditions set: 1) in compliance with a Specific Plan, 2) by LADOT, or 3) by other legal instrument excluding the Bureau of Engineering conditions, requiring an improvement that will change the geometrics of the public roadway or driveway apron may require additional or the reconstruction of street lighting improvements as part of that condition.

#### 5. Urban Forestry Division.

a. Plant street trees and remove any existing trees within dedicated streets or proposed dedicated streets as required by the Urban Forestry Division of the Bureau of Street Services. Parkway tree removals shall be replanted at a 2:1 ratio. All street tree plantings shall be brought up to current standards. When the City has previously been paid for tree plantings, the sub divider or contractor shall notify the Urban Forestry Division (213-847-3077) upon completion of construction to expedite tree planting.

Note: Removal or planting of any tree in the public right-of-way requires approval of the Board of Public Works. Contact Urban Forestry Division at: (213) 847-3077 for permit information. CEQA document must address parkway tree removal.

#### 6. Fire Department.

- a. Access for Fire Department apparatus and personnel to and into all structures shall be required.
- b. One or more Knox Boxes will be required to be installed for LAFD access to project. Location and number to be determined by LAFD Field Inspector. (Refer to FPB Requirement No. 75).
- c. 505.1 Address identification. New and existing buildings shall have approved building identification placed in a position that is plainly legible and visible from the street or road fronting the property.
- d. No building or portion of a building shall be constructed more than 150 feet from the edge of a roadway of an improved street, access road, or designated fire lane.
- e. Fire lane width shall not be less than 20 feet. When a fire lane must accommodate the operation of Fire Department aerial ladder apparatus or where fire hydrants are installed, those portions shall not be less than 28 feet in width.
- f. The width of private roadways for general access use and fire lanes shall not be less than 20 feet, and the fire lane must be clear to the sky.
- g. Fire lanes, where required and dead ending streets shall terminate in a cul-de-sac or other approved turning area. No dead ending street or fire lane shall be greater than 700 feet in length or secondary access shall be required.
- h. Submit plot plans indicating access road and turning area for Fire Department approval.



### Standard Form of Agreement Between Owner and Design-Builder

AGREEMENT made as of the « » day of « » in the year « » (In words, indicate day, month and year.)

#### **BETWEEN** the Owner:

(Name, legal status, address and other information)

~	≫≪	))
«	»	
«	»	
~	»	

and the Design-Builder: (Name, legal status, address and other information)

« »« » « » « » « »

for the following Project: (Name, location and detailed description)

« »

« » « »

The Owner and Design-Builder agree as follows.



added information needed for its completion. The author may also have revised the text of the original AIA standard form. An *Additions* and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

Consultation with an attorney is also encouraged with respect to professional licensing requirements in the jurisdiction where the Project is located.





ELECTRONIC COPYING of any portion of this AIA® Document to another electronic file is prohibited and constitutes a violation of copyright laws as set forth in the footer of this document.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, 1 e-mail copyright@aia.org. User Notes: (2034726772)

#### TABLE OF ARTICLES

- 1 **GENERAL PROVISIONS**
- 2 COMPENSATION AND PROGRESS PAYMENTS
- 3 GENERAL REQUIREMENTS OF THE WORK OF THE DESIGN-BUILD CONTRACT
- 4 WORK PRIOR TO EXECUTION OF THE DESIGN-BUILD AMENDMENT
- WORK FOLLOWING EXECUTION OF THE DESIGN-BUILD AMENDMENT 5
- 6 CHANGES IN THE WORK
- 7 **OWNER'S RESPONSIBILITIES**
- 8 TIME
- 9 PAYMENT APPLICATIONS AND PROJECT COMPLETION
- 10 PROTECTION OF PERSONS AND PROPERTY
- 11 UNCOVERING AND CORRECTION OF WORK
- **COPYRIGHTS AND LICENSES** 12
- 13 **TERMINATION OR SUSPENSION**
- CLAIMS AND DISPUTE RESOLUTION 14
- 15 **MISCELLANEOUS PROVISIONS**
- SCOPE OF THE AGREEMENT 16
- TABLE OF EXHIBITS
- Α **DESIGN-BUILD AMENDMENT**
- В INSURANCE AND BONDS
- С SUSTAINABLE PROJECTS

## ARTICLE 1 GENERAL PROVISIONS

design documents" where appropriate.)

### § 1.1 Owner's Criteria

This Agreement is based on the Owner's Criteria set forth in this Section 1.1. (Note the disposition for the following items by inserting the requested information or a statement such as "not applicable" or "unknown at time of execution." If the Owner intends to provide a set of design documents, and the requested information is contained in the design documents, identify the design documents and insert "see Owner's

#### **§ 1.1.1** The Owner's program for the Project:

(Set forth the program, identify documentation in which the program is set forth, or state the manner in which the program will be developed.)

« »

#### § 1.1.2 The Owner's design requirements for the Project and related documentation:

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, 2 e-mail copyright@aia.org. User Notes:

(Identify below, or in an attached exhibit, the documentation that contains the Owner's design requirements, including any performance specifications for the Project.)

« »

#### § 1.1.3 The Project's physical characteristics:

(Identify or describe, if appropriate, size, location, dimensions, or other pertinent information, such as geotechnical reports; site, boundary and topographic surveys; traffic and utility studies; availability of public and private utilities and services; legal description of the site; etc.)

« »

§ 1.1.4 The Owner's anticipated Sustainable Objective for the Project, if any:

(Identify the Owner's Sustainable Objective for the Project such as Sustainability Certification, benefit to the environment, enhancement to the health and well-being of building occupants, or improvement of energy efficiency. If the Owner identifies a Sustainable Objective, incorporate AIA Document A141<sup>™</sup>–2014, Exhibit C, Sustainable Projects, into this Agreement to define the terms, conditions and Work related to the Owner's Sustainable Objective.)

« »

**§ 1.1.5** Incentive programs the Owner intends to pursue for the Project, including those related to the Sustainable Objective, and any deadlines for receiving the incentives that are dependent on, or related to, the Design-Builder's services, are as follows:

(Identify incentive programs the Owner intends to pursue for the Project and deadlines for submitting or applying for the incentive programs.)

« »

**§ 1.1.6** The Owner's budget for the Work to be provided by the Design-Builder is set forth below: (*Provide total for Owner's budget, and if known, a line item breakdown of costs.*)

#### « »

§ 1.1.7 The Owner's design and construction milestone dates:

.1 Design phase milestone dates:

« »

.2 Submission of Design-Builder Proposal:

« »

.3 Phased completion dates:

« »

.4 Substantial Completion date:

« »

.5 Other milestone dates:

« »



## **§ 1.1.8** The Owner requires the Design-Builder to retain the following Architect, Consultants and Contractors at the Design-Builder's cost:

AIA Document A141<sup>®</sup> - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents<sup>®</sup> Terms of Service. To report copyright violations, e-mail copyright@aia.org. (2034726772)

(List name, legal status, address and other information.)



« »

**§ 1.2.3** The Owner will retain the following consultants and separate contractors: *(List discipline, scope of work, and, if known, identify by name and address.)* 

« »

§ 1.2.4 The Design-Builder identifies the following representative in accordance with Section 3.1.2:

AIA Document Al41" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No. 4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. (2034726772)

(List name, address and other information.)

« » « »

« »

« »

« »

« »

**§ 1.2.5** Neither the Owner's nor the Design-Builder's representative shall be changed without ten days' written notice to the other party.

#### § 1.3 Binding Dispute Resolution

For any Claim subject to, but not resolved by, mediation pursuant to Section 14.3, the method of binding dispute resolution shall be the following:

(Check the appropriate box. If the Owner and Design-Builder do not select a method of binding dispute resolution below, or do not subsequently agree in writing to a binding dispute resolution other than litigation, Claims will be resolved by litigation in a court of competent jurisdiction.)

[ « »

[ « » ] Arbitration pursuant to Section 14.4

[ « »] Litigation in a court of competent jurisdiction

[ « » ] Other: (Specify)

« »

#### § 1.4 Definitions

§ 1.4.1 Design-Build Documents. The Design-Build Documents consist of this Agreement between Owner and Design-Builder and its attached Exhibits (hereinafter, the "Agreement"); other documents listed in this Agreement; and Modifications issued after execution of this Agreement. A Modification is (1) a written amendment to the Contract signed by both parties, including the Design-Build Amendment, (2) a Change Order, or (3) a Change Directive.

**§ 1.4.2 The Contract.** The Design-Build Documents form the Contract. The Contract represents the entire and integrated agreement between the parties and supersedes prior negotiations, representations or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Design-Build Documents shall not be construed to create a contractual relationship of any kind between any persons or entities other than the Owner and the Design-Builder.

§ 1.4.3 The Work. The term "Work" means the design, construction and related services required to fulfill the Design-Builder's obligations under the Design-Build Documents, whether completed or partially completed, and includes all labor, materials, equipment and services provided or to be provided by the Design-Builder. The Work may constitute the whole or a part of the Project.

**§ 1.4.4 The Project.** The Project is the total design and construction of which the Work performed under the Design-Build Documents may be the whole or a part, and may include design and construction by the Owner and by separate contractors.

**§ 1.4.5 Instruments of Service.** Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Design-Builder, Contractor(s), Architect, and Consultant(s) under their respective agreements. Instruments of Service may include, without limitation, studies, surveys, models, sketches, drawings, specifications, digital models and other similar materials.

**§ 1.4.6 Submittal.** A Submittal is any submission to the Owner for review and approval demonstrating how the Design-Builder proposes to conform to the Design-Build Documents for those portions of the Work for which the

Design-Build Documents require Submittals. Submittals include, but are not limited to, shop drawings, product data, and samples. Submittals are not Design-Build Documents unless incorporated into a Modification.

§ 1.4.7 Owner. The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Design-Build Documents as if singular in number. The term "Owner" means the Owner or the Owner's authorized representative.

§ 1.4.8 Design-Builder. The Design-Builder is the person or entity identified as such in the Agreement and is referred to throughout the Design-Build Documents as if singular in number. The term "Design-Builder" means the Design-Builder or the Design-Builder's authorized representative.

§ 1.4.9 Consultant. A Consultant is a person or entity providing professional services for the Design-Builder for all or a portion of the Work, and is referred to throughout the Design-Build Documents as if singular in number. To the extent required by the relevant jurisdiction, the Consultant shall be lawfully licensed to provide the required professional services.

§ 1.4.10 Architect. The Architect is a person or entity providing design services for the Design-Builder for all or a portion of the Work, and is lawfully licensed to practice architecture in the applicable jurisdiction. The Architect is referred to throughout the Design-Build Documents as if singular in number.

§ 1.4.11 Contractor. A Contractor is a person or entity performing all or a portion of the construction, required in connection with the Work, for the Design-Builder. The Contractor shall be lawfully licensed, if required in the jurisdiction where the Project is located. The Contractor is referred to throughout the Design-Build Documents as if singular in number and means a Contractor or an authorized representative of the Contractor.

§ 1.4.12 Confidential Information. Confidential Information is information containing confidential or business proprietary information that is clearly marked as "confidential."

§ 1.4.13 Contract Time. Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, as set forth in the Design-Build Amendment for Substantial Completion of the Work.

§ 1.4.14 Day. The term "day" as used in the Design-Build Documents shall mean calendar day unless otherwise specifically defined.

§ 1.4.15 Contract Sum. The Contract Sum is the amount to be paid to the Design-Builder for performance of the Work after execution of the Design-Build Amendment, as identified in Article A.1 of the Design-Build Amendment.

#### ARTICLE 2 COMPENSATION AND PROGRESS PAYMENTS

§ 2.1 Compensation for Work Performed Prior To Execution of Design-Build Amendment

§ 2.1.1 Unless otherwise agreed, payments for Work performed prior to Execution of the Design-Build Amendment shall be made monthly. For the Design-Builder's performance of Work prior to the execution of the Design-Build Amendment, the Owner shall compensate the Design-Builder as follows:

(Insert amount of, or basis for, compensation, including compensation for any Sustainability Services, or indicate the exhibit in which the information is provided. If there will be a limit on the total amount of compensation for Work performed prior to the execution of the Design-Build Amendment, state the amount of the limit.)

« »

§ 2.1.2 The hourly billing rates for services of the Design-Builder and the Design-Builder's Architect, Consultants and Contractors, if any, are set forth below.

(If applicable, attach an exhibit of hourly billing rates or insert them below.)

« »

Individual or Position

Rate

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," %AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 6 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes (2034726772)

#### § 2.1.3 Compensation for Reimbursable Expenses Prior To Execution of Design-Build Amendment

§ 2.1.3.1 Reimbursable Expenses are in addition to compensation set forth in Section 2.1.1 and 2.1.2 and include expenses, directly related to the Project, incurred by the Design-Builder and the Design-Builder's Architect, Consultants, and Contractors, as follows:

- .1 Transportation and authorized out-of-town travel and subsistence;
- .2 Dedicated data and communication services, teleconferences, Project web sites, and extranets;
- .3 Fees paid for securing approval of authorities having jurisdiction over the Project;
- .4 Printing, reproductions, plots, standard form documents;
- .5 Postage, handling and delivery;
- .6 Expense of overtime work requiring higher than regular rates, if authorized in advance by the Owner;
- .7 Renderings, physical models, mock-ups, professional photography, and presentation materials requested by the Owner;
- .8 All taxes levied on professional services and on reimbursable expenses; and
- .9 Other Project-related expenditures, if authorized in advance by the Owner.

§ 2.1.3.2 For Reimbursable Expenses, the compensation shall be the expenses the Design-Builder and the Design-Builder's Architect, Consultants and Contractors incurred, plus an administrative fee of « » percent ( « » %) of the expenses incurred.

#### § 2.1.4 Payments to the Design-Builder Prior To Execution of Design-Build Amendment

§ 2.1.4.1 Payments are due and payable upon presentation of the Design-Builder's invoice. Amounts unpaid « » ( « » ) days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Design-Builder. (*Insert rate of monthly or annual interest agreed upon.*)

#### « » % « »

**§ 2.1.4.2** Records of Reimbursable Expenses and services performed on the basis of hourly rates shall be available to the Owner at mutually convenient times for a period of two years following execution of the Design-Build Amendment or termination of this Agreement, whichever occurs first.

#### § 2.2 Contract Sum and Payment for Work Performed After Execution of Design-Build Amendment

For the Design-Builder's performance of the Work after execution of the Design-Build Amendment, the Owner shall pay to the Design-Builder the Contract Sum in current funds as agreed in the Design-Build Amendment.

## ARTICLE 3 GENERAL REQUIREMENTS OF THE WORK OF THE DESIGN-BUILD CONTRACT § 3.1 General

**§ 3.1.1** The Design-Builder shall comply with any applicable licensing requirements in the jurisdiction where the Project is located.

§ 3.1.2 The Design-Builder shall designate in writing a representative who is authorized to act on the Design-Builder's behalf with respect to the Project.

**§ 3.1.3** The Design-Builder shall perform the Work in accordance with the Design-Build Documents. The Design-Builder shall not be relieved of the obligation to perform the Work in accordance with the Design-Build Documents by the activities, tests, inspections or approvals of the Owner.

**§ 3.1.3.1** The Design-Builder shall perform the Work in compliance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities. If the Design-Builder performs Work contrary to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities, the Design-Builder shall assume responsibility for such Work and shall bear the costs attributable to correction.

**§ 3.1.3.2** Neither the Design-Builder nor any Contractor, Consultant, or Architect shall be obligated to perform any act which they believe will violate any applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities. If the Design-Builder determines that implementation of any instruction received from the Owner, including those in the Owner's Criteria, would cause a violation of any applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, the Design-Builder determines that implementation of any applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, the Design-Builder shall notify the Owner in

writing. Upon verification by the Owner that a change to the Owner's Criteria is required to remedy the violation, the Owner and the Design-Builder shall execute a Modification in accordance with Article 6.

§ 3.1.4 The Design-Builder shall be responsible to the Owner for acts and omissions of the Design-Builder's employees, Architect, Consultants, Contractors, and their agents and employees, and other persons or entities performing portions of the Work.

§ 3.1.5 General Consultation. The Design-Builder shall schedule and conduct periodic meetings with the Owner to review matters such as procedures, progress, coordination, and scheduling of the Work.

§ 3.1.6 When applicable law requires that services be performed by licensed professionals, the Design-Builder shall provide those services through qualified, licensed professionals. The Owner understands and agrees that the services of the Design-Builder's Architect and the Design-Builder's other Consultants are performed in the sole interest of, and for the exclusive benefit of, the Design-Builder.

§ 3.1.7 The Design-Builder, with the assistance of the Owner, shall prepare and file documents required to obtain necessary approvals of governmental authorities having jurisdiction over the Project.

#### § 3.1.8 Progress Reports

§ 3.1.8.1 The Design-Builder shall keep the Owner informed of the progress and quality of the Work. On a monthly basis, or otherwise as agreed to by the Owner and Design-Builder, the Design-Builder shall submit written progress reports to the Owner, showing estimated percentages of completion and other information identified below:

- Work completed for the period; .1
- .2 Project schedule status;
- .3 Submittal schedule and status report, including a summary of outstanding Submittals;
- .4 Responses to requests for information to be provided by the Owner;
- .5 Approved Change Orders and Change Directives;
- .6 Pending Change Order and Change Directive status reports;
- .7 Tests and inspection reports;
- .8 Status report of Work rejected by the Owner;
- .9 Status of Claims previously submitted in accordance with Article 14;
- .10 Cumulative total of the Cost of the Work to date including the Design-Builder's compensation and Reimbursable Expenses, if any;
- .11 Current Project cash-flow and forecast reports; and
- .12 Additional information as agreed to by the Owner and Design-Builder.

§ 3.1.8.2 In addition, where the Contract Sum is the Cost of the Work with or without a Guaranteed Maximum Price, the Design-Builder shall include the following additional information in its progress reports:

- .1 Design-Builder's work force report;
- .2 Equipment utilization report; and
- .3 Cost summary, comparing actual costs to updated cost estimates.

#### § 3.1.9 Design-Builder's Schedules

§ 3.1.9.1 The Design-Builder, promptly after execution of this Agreement, shall prepare and submit for the Owner's information a schedule for the Work. The schedule, including the time required for design and construction, shall not exceed time limits current under the Design-Build Documents, shall be revised at appropriate intervals as required by the conditions of the Work and Project, shall be related to the entire Project to the extent required by the Design-Build Documents, shall provide for expeditious and practicable execution of the Work, and shall include allowances for periods of time required for the Owner's review and for approval of submissions by authorities having jurisdiction over the Project.

§ 3.1.9.2 The Design-Builder shall perform the Work in general accordance with the most recent schedules submitted to the Owner.

§ 3.1.10 Certifications. Upon the Owner's written request, the Design-Builder shall obtain from the Architect, Consultants, and Contractors, and furnish to the Owner, certifications with respect to the documents and services provided by the Architect, Consultants, and Contractors (a) that, to the best of their knowledge, information and belief, the documents or services to which the certifications relate (i) are consistent with the Design-Build Documents, except to the extent specifically identified in the certificate, and (ii) comply with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities governing the design of the Project; and (b) that the Owner and its consultants shall be entitled to rely upon the accuracy of the representations and statements contained in the certifications. The Design-Builder's Architect, Consultants, and Contractors shall not be required to execute certificates or consents that would require knowledge, services or responsibilities beyond the scope of their services.

#### § 3.1.11 Design-Builder's Submittals

§ 3.1.11.1 Prior to submission of any Submittals, the Design-Builder shall prepare a Submittal schedule, and shall submit the schedule for the Owner's approval. The Owner's approval shall not unreasonably be delayed or withheld. The Submittal schedule shall (1) be coordinated with the Design-Builder's schedule provided in Section 3.1.9.1, (2) allow the Owner reasonable time to review Submittals, and (3) be periodically updated to reflect the progress of the Work. If the Design-Builder fails to submit a Submittal schedule, the Design-Builder shall not be entitled to any increase in Contract Sum or extension of Contract Time based on the time required for review of Submittals.

§ 3.1.11.2 By providing Submittals the Design-Builder represents to the Owner that it has (1) reviewed and approved them, (2) determined and verified materials, field measurements and field construction criteria related thereto, or will do so and (3) checked and coordinated the information contained within such Submittals with the requirements of the Work and of the Design-Build Documents.

§ 3.1.11.3 The Design-Builder shall perform no portion of the Work for which the Design-Build Documents require Submittals until the Owner has approved the respective Submittal.

§ 3.1.11.4 The Work shall be in accordance with approved Submittals except that the Design-Builder shall not be relieved of its responsibility to perform the Work consistent with the requirements of the Design-Build Documents. The Work may deviate from the Design-Build Documents only if the Design-Builder has notified the Owner in writing of a deviation from the Design-Build Documents at the time of the Submittal and a Modification is executed authorizing the identified deviation. The Design-Builder shall not be relieved of responsibility for errors or omissions in Submittals by the Owner's approval of the Submittals.

§ 3.1.11.5 All professional design services or certifications to be provided by the Design-Builder, including all drawings, calculations, specifications, certifications, shop drawings and other Submittals, shall contain the signature and seal of the licensed design professional preparing them. Submittals related to the Work designed or certified by the licensed design professionals, if prepared by others, shall bear the licensed design professional's written approval. The Owner and its consultants shall be entitled to rely upon the adequacy, accuracy and completeness of the services, certifications or approvals performed by such design professionals.

§ 3.1.12 Warranty. The Design-Builder warrants to the Owner that materials and equipment furnished under the Contract will be of good quality and new unless the Design-Build Documents require or permit otherwise. The Design-Builder further warrants that the Work will conform to the requirements of the Design-Build Documents and will be free from defects, except for those inherent in the quality of the Work or otherwise expressly permitted by the Design-Build Documents. Work, materials, or equipment not conforming to these requirements may be considered defective. The Design-Builder's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Design-Builder, improper or insufficient maintenance, improper operation, or normal wear and tear and normal usage. If required by the Owner, the Design-Builder shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

#### § 3.1.13 Royalties, Patents and Copyrights

§ 3.1.13.1 The Design-Builder shall pay all royalties and license fees.

§ 3.1.13.2 The Design-Builder shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner and its separate contractors and consultants harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process or product of a particular manufacturer or manufacturers is required by the Owner, or where the copyright violations are required in the Owner's Criteria. However, if the Design-Builder has reason to believe that the design, process or product required in the Owner's Criteria is an infringement of a copyright or a patent, the Design-Builder shall be responsible for such loss unless such information is promptly furnished to the Owner. If the Owner receives notice from a patent or copyright owner of an

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 9 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes (2034726772)

alleged violation of a patent or copyright, attributable to the Design-Builder, the Owner shall give prompt written notice to the Design-Builder.

#### § 3.1.14 Indemnification

§ 3.1.14.1 To the fullest extent permitted by law, the Design-Builder shall indemnify and hold harmless the Owner, including the Owner's agents and employees, from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work, but only to the extent caused by the negligent acts or omissions of the Design-Builder, Architect, a Consultant, a Contractor, or anyone directly or indirectly employed by them or anyone for whose acts they may be liable. Such obligation shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity that would otherwise exist as to a party or person described in this Section 3.1.14.

§ 3.1.14.2 The indemnification obligation under this Section 3.1.14 shall not be limited by a limitation on amount or type of damages, compensation, or benefits payable by or for Design-Builder, Architect, a Consultant, a Contractor, or anyone directly or indirectly employed by them, under workers' compensation acts, disability benefit acts or other employee benefit acts.

#### § 3.1.15 Contingent Assignment of Agreements

§ 3.1.15.1 Each agreement for a portion of the Work is assigned by the Design-Builder to the Owner, provided that

- assignment is effective only after termination of the Contract by the Owner for cause, pursuant to .1 Sections 13.1.4 or 13.2.2, and only for those agreements that the Owner accepts by written notification to the Design-Builder and the Architect, Consultants, and Contractors whose agreements are accepted for assignment; and
- .2 assignment is subject to the prior rights of the surety, if any, obligated under bond relating to the Contract.

When the Owner accepts the assignment of an agreement, the Owner assumes the Design-Builder's rights and obligations under the agreement.

§ 3.1.15.2 Upon such assignment, if the Work has been suspended for more than 30 days, the compensation under the assigned agreement shall be equitably adjusted for increases in cost resulting from the suspension.

§ 3.1.15.3 Upon such assignment to the Owner under this Section 3.1.15, the Owner may further assign the agreement to a successor design-builder or other entity. If the Owner assigns the agreement to a successor design-builder or other entity, the Owner shall nevertheless remain legally responsible for all of the successor design builder's or other entity's obligations under the agreement.

§ 3.1.16 Design-Builder's Insurance and Bonds. The Design-Builder shall purchase and maintain insurance and provide bonds as set forth in Exhibit B.

#### ARTICLE 4 WORK PRIOR TO EXECUTION OF THE DESIGN-BUILD AMENDMENT § 4.1 General

§ 4.1.1 Any information submitted by the Design-Builder, and any interim decisions made by the Owner, shall be for the purpose of facilitating the design process and shall not modify the Owner's Criteria unless the Owner and Design-Builder execute a Modification.

§ 4.1.2 The Design-Builder shall advise the Owner on proposed site use and improvements, selection of materials, and building systems and equipment. The Design-Builder shall also provide the Owner with recommendations, consistent with the Owner's Criteria, on constructability; availability of materials and labor; time requirements for procurement, installation and construction; and factors related to construction cost including, but not limited to, costs of alternative designs or materials, preliminary budgets, life-cycle data, and possible cost reductions.

#### § 4.2 Evaluation of the Owner's Criteria

§ 4.2.1 The Design-Builder shall schedule and conduct meetings with the Owner and any other necessary individuals or entities to discuss and review the Owner's Criteria as set forth in Section 1.1. The Design-Builder shall thereafter again meet with the Owner to discuss a preliminary evaluation of the Owner's Criteria. The preliminary evaluation shall address possible alternative approaches to design and construction of the Project and include the

Design-Builder's recommendations, if any, with regard to accelerated or fast-track scheduling, procurement, or phased construction. The preliminary evaluation shall consider cost information, constructability, and procurement and construction scheduling issues.

**§ 4.2.2** After the Design-Builder meets with the Owner and presents the preliminary evaluation, the Design-Builder shall provide a written report to the Owner, summarizing the Design-Builder's evaluation of the Owner's Criteria. The report shall also include

- .1 allocations of program functions, detailing each function and their square foot areas;
- .2 a preliminary estimate of the Cost of the Work, and, if necessary, recommendations to adjust the Owner's Criteria to conform to the Owner's budget;
- .3 a preliminary schedule, which shall include proposed design milestones; dates for receiving additional information from, or for work to be completed by, the Owner; anticipated date for the Design-Builder's Proposal; and dates of periodic design review sessions with the Owner; and
  .4 the following:
- the following: (List additional information, if any, to be included in the Design-Builder's written report.)

« »

**§ 4.2.3** The Owner shall review the Design-Builder's written report and, if acceptable, provide the Design-Builder with written consent to proceed to the development of the Preliminary Design as described in Section 4.3. The consent to proceed shall not be understood to modify the Owner's Criteria unless the Owner and Design-Builder execute a Modification.

#### § 4.3 Preliminary Design

**§ 4.3.1** Upon the Owner's issuance of a written consent to proceed under Section 4.2.3, the Design-Builder shall prepare and submit a Preliminary Design to the Owner. The Preliminary Design shall include a report identifying any deviations from the Owner's Criteria, and shall include the following:

- .1 Confirmation of the allocations of program functions;
- .2 Site plan;
- .3 Building plans, sections and elevations;
- .4 Structural system;
- .5 Selections of major building systems, including but not limited to mechanical, electrical and plumbing systems; and
- .6 Outline specifications or sufficient drawing notes describing construction materials.

The Preliminary Design may include some combination of physical study models, perspective sketches, or digital modeling.

**§ 4.3.2** The Owner shall review the Preliminary Design and, if acceptable, provide the Design-Builder with written consent to proceed to development of the Design-Builder's Proposal. The Preliminary Design shall not modify the Owner's Criteria unless the Owner and Design-Builder execute a Modification.

#### § 4.4 Design-Builder's Proposal

§ 4.4.1 Upon the Owner's issuance of a written consent to proceed under Section 4.3.2, the Design-Builder shall prepare and submit the Design-Builder's Proposal to the Owner. The Design-Builder's Proposal shall include the following:

- .1 A list of the Preliminary Design documents and other information, including the Design-Builder's clarifications, assumptions and deviations from the Owner's Criteria, upon which the Design-Builder's Proposal is based;
- .2 The proposed Contract Sum, including the compensation method and, if based upon the Cost of the Work plus a fee, a written statement of estimated cost organized by trade categories, allowances, contingencies, Design-Builder's Fee, and other items that comprise the Contract Sum;
- .3 The proposed date the Design-Builder shall achieve Substantial Completion;
- .4 An enumeration of any qualifications and exclusions, if applicable;
- .5 A list of the Design-Builder's key personnel, Contractors and suppliers; and
- .6 The date on which the Design-Builder's Proposal expires.

**§ 4.4.2** Submission of the Design-Builder's Proposal shall constitute a representation by the Design-Builder that it has visited the site and become familiar with local conditions under which the Work is to be completed.

**§ 4.4.3** If the Owner and Design-Builder agree on a proposal, the Owner and Design-Builder shall execute the Design-Build Amendment setting forth the terms of their agreement.

#### ARTICLE 5 WORK FOLLOWING EXECUTION OF THE DESIGN-BUILD AMENDMENT

#### § 5.1 Construction Documents

§ 5.1.1 Upon the execution of the Design-Build Amendment, the Design-Builder shall prepare Construction Documents. The Construction Documents shall establish the quality levels of materials and systems required. The Construction Documents shall be consistent with the Design-Build Documents.

**§ 5.1.2** The Design-Builder shall provide the Construction Documents to the Owner for the Owner's information. If the Owner discovers any deviations between the Construction Documents and the Design-Build Documents, the Owner shall promptly notify the Design-Builder of such deviations in writing. The Construction Documents shall not modify the Design-Build Documents unless the Owner and Design-Builder execute a Modification. The failure of the Owner to discover any such deviations shall not relieve the Design-Builder of the obligation to perform the Work in accordance with the Design-Build Documents.

#### § 5.2 Construction

§ 5.2.1 Commencement. Except as permitted in Section 5.2.2, construction shall not commence prior to execution of the Design-Build Amendment.

**§ 5.2.2** If the Owner and Design-Builder agree in writing, construction may proceed prior to the execution of the Design-Build Amendment. However, such authorization shall not waive the Owner's right to reject the Design-Builder's Proposal.

**§ 5.2.3** The Design-Builder shall supervise and direct the Work, using the Design-Builder's best skill and attention. The Design-Builder shall be solely responsible for, and have control over, construction means, methods, techniques, sequences and procedures, and for coordinating all portions of the Work under the Contract, unless the Design-Build Documents give other specific instructions concerning these matters.

**§ 5.2.4** The Design-Builder shall be responsible for inspection of portions of Work already performed to determine that such portions are in proper condition to receive subsequent Work.

#### § 5.3 Labor and Materials

**§ 5.3.1** Unless otherwise provided in the Design-Build Documents, the Design-Builder shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services, necessary for proper execution and completion of the Work, whether temporary or permanent, and whether or not incorporated or to be incorporated in the Work.

**§ 5.3.2** When a material or system is specified in the Design-Build Documents, the Design-Builder may make substitutions only in accordance with Article 6.

**§ 5.3.3** The Design-Builder shall enforce strict discipline and good order among the Design-Builder's employees and other persons carrying out the Work. The Design-Builder shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them.

#### § 5.4 Taxes

The Design-Builder shall pay sales, consumer, use and similar taxes, for the Work provided by the Design-Builder, that are legally enacted when the Design-Build Amendment is executed, whether or not yet effective or merely scheduled to go into effect.

#### § 5.5 Permits, Fees, Notices and Compliance with Laws

**§ 5.5.1** Unless otherwise provided in the Design-Build Documents, the Design-Builder shall secure and pay for the building permit as well as any other permits, fees, licenses, and inspections by government agencies, necessary for proper execution of the Work and Substantial Completion of the Project.

AIA Document A141<sup>®</sup> - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft 12 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. (2034726772)

**§ 5.5.2** The Design-Builder shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities, applicable to performance of the Work.

**§ 5.5.3 Concealed or Unknown Conditions.** If the Design-Builder encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Design-Build Documents or (2) unknown physical conditions of an unusual nature that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Design-Build Documents, the Design-Builder shall promptly provide notice to the Owner before conditions are disturbed and in no event later than 21 days after first observance of the conditions. The Owner shall promptly investigate such conditions and, if the Owner determines that they differ materially and cause an increase or decrease in the Design-Builder's cost of, or time required for, performance of any part of the Work, shall recommend an equitable adjustment in the Contract Sum or Contract Time, or both. If the Owner determines that the conditions at the site are not materially different from those indicated in the Design-Builder in writing, stating the reasons. If the Design-Builder disputes the Owner's determination or recommendation, the Design-Builder may proceed as provided in Article 14.

**§ 5.5.4** If, in the course of the Work, the Design-Builder encounters human remains, or recognizes the existence of burial markers, archaeological sites, or wetlands, not indicated in the Design-Build Documents, the Design-Builder shall immediately suspend any operations that would affect them and shall notify the Owner. Upon receipt of such notice, the Owner shall promptly take any action necessary to obtain governmental authorization required to resume the operations. The Design-Builder shall continue to suspend such operations until otherwise instructed by the Owner but shall continue with all other operations that do not affect those remains or features. Requests for adjustments in the Contract Sum and Contract Time arising from the existence of such remains or features may be made as provided in Article 14.

#### § 5.6 Allowances

**§ 5.6.1** The Design-Builder shall include in the Contract Sum all allowances stated in the Design-Build Documents. Items covered by allowances shall be supplied for such amounts, and by such persons or entities as the Owner may direct, but the Design-Builder shall not be required to employ persons or entities to whom the Design-Builder has reasonable objection.

§ 5.6.2 Unless otherwise provided in the Design-Build Documents,

- .1 allowances shall cover the cost to the Design-Builder of materials and equipment delivered at the site and all required taxes, less applicable trade discounts;
- .2 the Design-Builder's costs for unloading and handling at the site, labor, installation costs, overhead, profit, and other expenses contemplated for stated allowance amounts, shall be included in the Contract Sum but not in the allowances; and
- .3 whenever costs are more than or less than allowances, the Contract Sum shall be adjusted accordingly by Change Order. The amount of the Change Order shall reflect (1) the difference between actual costs and the allowances under Section 5.6.2.1 and (2) changes in Design-Builder's costs under Section 5.6.2.2.

**§ 5.6.3** The Owner shall make selections of materials and equipment with reasonable promptness for allowances requiring Owner selection.

#### § 5.7 Key Personnel, Contractors and Suppliers

**§ 5.7.1** The Design-Builder shall not employ personnel, or contract with Contractors or suppliers to whom the Owner has made reasonable and timely objection. The Design-Builder shall not be required to contract with anyone to whom the Design-Builder has made reasonable and timely objection.

**§ 5.7.2** If the Design-Builder changes any of the personnel, Contractors or suppliers identified in the Design-Builder Amendment, the Design-Builder shall notify the Owner and provide the name and qualifications of the new personnel, Contractor or supplier. The Owner may reply within 14 days to the Design-Builder in writing, stating (1) whether the Owner has reasonable objection to the proposed personnel, Contractor or supplier or (2) that the Owner requires

59 of 258

additional time to review. Failure of the Owner to reply within the 14-day period shall constitute notice of no reasonable objection.

§ 5.7.3 Except for those persons or entities already identified or required in the Design-Build Amendment, the Design-Builder, as soon as practicable after execution of the Design-Build Amendment, shall furnish in writing to the Owner the names of persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for each principal portion of the Work. The Owner may reply within 14 days to the Design-Builder in writing stating (1) whether the Owner has reasonable objection to any such proposed person or entity or (2) that the Owner requires additional time for review. Failure of the Owner to reply within the 14-day period-shall constitute notice of no reasonable objection.

**§ 5.7.3.1** If the Owner has reasonable objection to a person or entity proposed by the Design-Builder, the Design-Builder shall propose another to whom the Owner has no reasonable objection. If the rejected person or entity was reasonably capable of performing the Work, the Contract Sum and Contract Time shall be increased or decreased by the difference, if any, occasioned by such change, and an appropriate Change Order shall be issued before commencement of the substitute person or entity's Work. However, no increase in the Contract Sum or Contract Time shall be allowed for such change unless the Design-Builder has acted promptly and responsively in submitting names as required.

#### § 5.8 Documents and Submittals at the Site

The Design-Builder shall maintain at the site for the Owner one copy of the Design-Build Documents and a current set of the Construction Documents, in good order and marked currently to indicate field changes and selections made during construction, and one copy of approved Submittals. The Design-Builder shall deliver these items to the Owner in accordance with Section 9.10.2 as a record of the Work as constructed.

#### § 5.9 Use of Site

The Design-Builder shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, lawful orders of public authorities, and the Design-Build Documents, and shall not unreasonably encumber the site with materials or equipment.

#### § 5.10 Cutting and Patching

The Design-Builder shall not cut, patch or otherwise alter fully or partially completed construction by the Owner or a separate contractor except with written consent of the Owner and of such separate contractor; such consent shall not be unreasonably withheld. The Design-Builder shall not unreasonably withhold from the Owner or a separate contractor the Design-Builder's consent to cutting or otherwise altering the Work.

#### § 5.11 Cleaning Up

§ 5.11.1 The Design-Builder shall keep the premises and surrounding area free from accumulation of waste materials or rubbish caused by operations under the Contract. At completion of the Work, the Design-Builder shall remove waste materials, rubbish, the Design-Builder's tools, construction equipment, machinery and surplus materials from and about the Project.

**§ 5.11.2** If the Design-Builder fails to clean up as provided in the Design-Build Documents, the Owner may do so and Owner shall be entitled to reimbursement from the Design-Builder.

#### § 5.12 Access to Work

The Design-Builder shall provide the Owner and its separate contractors and consultants access to the Work in preparation and progress wherever located. The Design-Builder shall notify the Owner regarding Project safety criteria and programs, which the Owner, and its contractors and consultants, shall comply with while at the site.

#### § 5.13 Construction by Owner or by Separate Contractors

#### § 5.13.1 Owner's Right to Perform Construction and to Award Separate Contracts

**§ 5.13.1.1** The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces; and to award separate contracts in connection with other portions of the Project, or other construction or operations on the site, under terms and conditions identical or substantially similar to this Contract, including those terms and conditions related to insurance and waiver of subrogation. The Owner shall notify the Design-Builder

promptly after execution of any separate contract. If the Design-Builder claims that delay or additional cost is involved because of such action by the Owner, the Design-Builder shall make a Claim as provided in Article 14.

§ 5.13.1.2 When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "Design-Builder" in the Design-Build Documents in each case shall mean the individual or entity that executes each separate agreement with the Owner.

§ 5.13.1.3 The Owner shall provide for coordination of the activities of the Owner's own forces, and of each separate contractor, with the Work of the Design-Builder, who shall cooperate with them. The Design-Builder shall participate with other separate contractors and the Owner in reviewing their construction schedules. The Design-Builder shall make any revisions to the construction schedule deemed necessary after a joint review and mutual agreement. The construction schedules shall then constitute the schedules to be used by the Design-Builder, separate contractors and the Owner until subsequently revised.

§ 5.13.1.4 Unless otherwise provided in the Design-Build Documents, when the Owner performs construction or operations related to the Project with the Owner's own forces or separate contractors, the Owner shall be deemed to be subject to the same obligations, and to have the same rights, that apply to the Design-Builder under the Contract.

#### § 5.14 Mutual Responsibility

§ 5.14.1 The Design-Builder shall afford the Owner and separate contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Design-Builder's construction and operations with theirs as required by the Design-Build Documents.

§ 5.14.2 If part of the Design-Builder's Work depends upon construction or operations by the Owner or a separate contractor, the Design-Builder shall, prior to proceeding with that portion of the Work, prepare a written report to the Owner, identifying apparent discrepancies or defects in the construction or operations by the Owner or separate contractor that would render it unsuitable for proper execution and results of the Design-Builder's Work. Failure of the Design-Builder to report shall constitute an acknowledgment that the Owner's or separate contractor's completed or partially completed construction is fit and proper to receive the Design-Builder's Work, except as to defects not then reasonably discoverable.

§ 5.14.3 The Design-Builder shall reimburse the Owner for costs the Owner incurs that are payable to a separate contractor because of the Design-Builder's delays, improperly timed activities or defective construction. The Owner shall be responsible to the Design-Builder for costs the Design-Builder incurs because of a separate contractor's delays, improperly timed activities, damage to the Work or defective construction.

§ 5.14.4 The Design-Builder shall promptly remedy damage the Design-Builder wrongfully causes to completed or partially completed construction or to property of the Owner or separate contractors as provided in Section 10.2.5.

§ 5.14.5 The Owner and each separate contractor shall have the same responsibilities for cutting and patching the Work as the Design-Builder has with respect to the construction of the Owner or separate contractors in Section 5.10.

#### § 5.15 Owner's Right to Clean Up

If a dispute arises among the Design-Builder, separate contractors and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up and will allocate the cost among those responsible.

#### **ARTICLE 6 CHANGES IN THE WORK**

#### § 6.1 General

§ 6.1.1 Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order or Change Directive, subject to the limitations stated in this Article 6 and elsewhere in the Design-Build Documents.

§ 6.1.2 A Change Order shall be based upon agreement between the Owner and Design-Builder. The Owner may issue a Change Directive without agreement by the Design-Builder.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," %ATA," the ATA Logo, and "ATA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by ATA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 15 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes

**§ 6.1.3** Changes in the Work shall be performed under applicable provisions of the Design-Build Documents, and the Design-Builder shall proceed promptly, unless otherwise provided in the Change Order or Change Directive.

#### § 6.2 Change Orders

A Change Order is a written instrument signed by the Owner and Design-Builder stating their agreement upon all of the following:

- .1 The change in the Work;
- .2 The amount of the adjustment, if any, in the Contract Sum or, if prior to execution of the Design-Build Amendment, the adjustment in the Design-Builder's compensation; and
- .3 The extent of the adjustment, if any, in the Contract Time.

#### § 6.3 Change Directives

§ 6.3.1 A Change Directive is a written order signed by the Owner directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or, if prior to execution of the Design-Build Amendment, the adjustment in the Design-Builder's compensation, or Contract Time. The Owner may by Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, the Contract Sum or, if prior to execution of the Design-Build Amendment, the adjustment in the Design-Builder's compensation, and Contract Time being adjusted accordingly.

§ 6.3.2 A Change Directive shall be used in the absence of total agreement on the terms of a Change Order.

**§ 6.3.3** If the Change Directive provides for an adjustment to the Contract Sum or, if prior to execution of the Design-Build Amendment, an adjustment in the Design-Builder's compensation, the adjustment shall be based on one of the following methods:

- .1 Mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation;
- .2 Unit prices stated in the Design-Build Documents or subsequently agreed upon;
- .3 Cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee; or
- .4 As provided in Section 6.3.7.

**§ 6.3.4** If unit prices are stated in the Design-Build Documents or subsequently agreed upon, and if quantities originally contemplated are materially changed in a proposed Change Order or Change Directive so that application of such unit prices to quantities of Work proposed will cause substantial inequity to the Owner or Design-Builder, the applicable unit prices shall be equitably adjusted.

**§ 6.3.5** Upon receipt of a Change Directive, the Design-Builder shall promptly proceed with the change in the Work involved and advise the Owner of the Design-Builder's agreement or disagreement with the method, if any, provided in the Change Directive for determining the proposed adjustment in the Contract Sum or, if prior to execution of the Design-Builder's compensation, or Contract Time.

**§ 6.3.6** A Change Directive signed by the Design-Builder indicates the Design-Builder's agreement therewith, including adjustment in Contract Sum or, if prior to execution of the Design-Build Amendment, the adjustment in the Design-Builder's compensation, and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.

**§ 6.3.7** If the Design-Builder does not respond promptly or disagrees with the method for adjustment in the Contract Sum or, if prior to execution of the Design-Build Amendment, the method for adjustment in the Design-Builder's compensation, the Owner shall determine the method and the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase, an amount for overhead and profit as set forth in the Agreement, or if no such amount is set forth in the Agreement, a reasonable amount. In such case, and also under Section 6.3.3.3, the Design-Builder shall keep and present, in such form as the Owner may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Design-Build Documents, costs for the purposes of this Section 6.3.7 shall be limited to the following:

- .1 Additional costs of professional services;
- .2 Costs of labor, including social security, unemployment insurance, fringe benefits required by agreement or custom, and workers' compensation insurance;

- .3 Costs of materials, supplies and equipment, including cost of transportation, whether incorporated or consumed;
- .4 Rental costs of machinery and equipment, exclusive of hand tools, whether rented from the Design-Builder or others;
- .5 Costs of premiums for all bonds and insurance, permit fees, and sales, use or similar taxes related to the Work: and
- .6 Additional costs of supervision and field office personnel directly attributable to the change.

§ 6.3.8 The amount of credit to be allowed by the Design-Builder to the Owner for a deletion or change that results in a net decrease in the Contract Sum or, if prior to execution of the Design-Build Amendment, in the Design-Builder's compensation, shall be actual net cost. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.

§ 6.3.9 Pending final determination of the total cost of a Change Directive to the Owner, the Design-Builder may request payment for Work completed under the Change Directive in Applications for Payment. The Owner will make an interim determination for purposes of certification for payment for those costs deemed to be reasonably justified. The Owner's interim determination of cost shall adjust the Contract Sum or, if prior to execution of the Design-Build Amendment, the Design-Builder's compensation, on the same basis as a Change Order, subject to the right of Design-Builder to disagree and assert a Claim in accordance with Article 14.

§ 6.3.10 When the Owner and Design-Builder agree with a determination concerning the adjustments in the Contract Sum or, if prior to execution of the Design-Build Amendment, the adjustment in the Design-Builder's compensation and Contract Time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and the Owner and Design-Builder shall execute a Change Order. Change Orders may be issued for all or any part of a Change Directive.

#### **ARTICLE 7 OWNER'S RESPONSIBILITIES**

#### § 7.1 General

§ 7.1.1 The Owner shall designate in writing a representative who shall have express authority to bind the Owner with respect to all Project matters requiring the Owner's approval or authorization.

§ 7.1.2 The Owner shall render decisions in a timely manner and in accordance with the Design-Builder's schedule agreed to by the Owner. The Owner shall furnish to the Design-Builder, within 15 days after receipt of a written request, information necessary and relevant for the Design-Builder to evaluate, give notice of or enforce mechanic's lien rights. Such information shall include a correct statement of the record legal title to the property on which the Project is located, usually referred to as the site, and the Owner's interest therein.

#### § 7.2 Information and Services Required of the Owner

§ 7.2.1 The Owner shall furnish information or services required of the Owner by the Design-Build Documents with reasonable promptness.

§ 7.2.2 The Owner shall provide, to the extent under the Owner's control and if not required by the Design-Build Documents to be provided by the Design-Builder, the results and reports of prior tests, inspections or investigations conducted for the Project involving structural or mechanical systems; chemical, air and water pollution; hazardous materials; or environmental and subsurface conditions and information regarding the presence of pollutants at the Project site. Upon receipt of a written request from the Design-Builder, the Owner shall also provide surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site under the Owner's control.

§ 7.2.3 The Owner shall promptly obtain easements, zoning variances, and legal authorizations or entitlements regarding site utilization where essential to the execution of the Project.

§ 7.2.4 The Owner shall cooperate with the Design-Builder in securing building and other permits, licenses and inspections.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, 17 e-mail copyright@aia.org. User Notes

§ 7.2.5 The services, information, surveys and reports required to be provided by the Owner under this Agreement, shall be furnished at the Owner's expense, and except as otherwise specifically provided in this Agreement or elsewhere in the Design-Build Documents or to the extent the Owner advises the Design-Builder to the contrary in writing, the Design-Builder shall be entitled to rely upon the accuracy and completeness thereof. In no event shall the Design-Builder be relieved of its responsibility to exercise proper precautions relating to the safe performance of the Work.

§ 7.2.6 If the Owner observes or otherwise becomes aware of a fault or defect in the Work or non-conformity with the Design-Build Documents, the Owner shall give prompt written notice thereof to the Design-Builder.

§ 7.2.7 Prior to the execution of the Design-Build Amendment, the Design-Builder may request in writing that the Owner provide reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Design-Build Documents and the Design-Builder's Proposal. Thereafter, the Design-Builder may only request such evidence if (1) the Owner fails to make payments to the Design-Builder as the Design-Build Documents require; (2) a change in the Work materially changes the Contract Sum; or (3) the Design-Builder identifies in writing a reasonable concern regarding the Owner's ability to make payment when due. The Owner shall furnish such evidence as a condition precedent to commencement or continuation of the Work or the portion of the Work affected by a material change. After the Owner furnishes the evidence, the Owner shall not materially vary such financial arrangements without prior notice to the Design-Builder.

§ 7.2.8 Except as otherwise provided in the Design-Build Documents or when direct communications have been specially authorized, the Owner shall communicate through the Design-Builder with persons or entities employed or retained by the Design-Builder.

§ 7.2.9 Unless required by the Design-Build Documents to be provided by the Design-Builder, the Owner shall, upon request from the Design-Builder, furnish the services of geotechnical engineers or other consultants for investigation of subsurface, air and water conditions when such services are reasonably necessary to properly carry out the design services furnished by the Design-Builder. In such event, the Design-Builder shall specify the services required. Such services may include, but are not limited to, test borings, test pits, determinations of soil bearing values, percolation tests, evaluations of hazardous materials, ground corrosion and resistivity tests, and necessary operations for anticipating subsoil conditions. The services of geotechnical engineer(s) or other consultants shall include preparation and submission of all appropriate reports and professional recommendations.

§ 7.2.10 The Owner shall purchase and maintain insurance as set forth in Exhibit B.

#### § 7.3 Submittals

§ 7.3.1 The Owner shall review and approve or take other appropriate action on Submittals. Review of Submittals is not conducted for the purpose of determining the accuracy and completeness of other details, such as dimensions and quantities; or for substantiating instructions for installation or performance of equipment or systems; or for determining that the Submittals are in conformance with the Design-Build Documents, all of which remain the responsibility of the Design-Builder as required by the Design-Build Documents. The Owner's action will be taken in accordance with the submittal schedule approved by the Owner or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time in the Owner's judgment to permit adequate review. The Owner's review of Submittals shall not relieve the Design-Builder of the obligations under Sections 3.1.11, 3.1.12, and 5.2.3. The Owner's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Owner, of any construction means, methods, techniques, sequences or procedures. The Owner's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

§ 7.3.2 Upon review of the Submittals required by the Design-Build Documents, the Owner shall notify the Design-Builder of any non-conformance with the Design-Build Documents the Owner discovers.

§ 7.4 Visits to the site by the Owner shall not be construed to create an obligation on the part of the Owner to make on-site inspections to check the quality or quantity of the Work. The Owner shall neither have control over or charge of, nor be responsible for, the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, because these are solely the Design-Builder's rights and responsibilities under the Design-Build Documents.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "ATA," the ATA Logo, and "ATA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by ATA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 18 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes

§ 7.5 The Owner shall not be responsible for the Design-Builder's failure to perform the Work in accordance with the requirements of the Design-Build Documents. The Owner shall not have control over or charge of, and will not be responsible for acts or omissions of the Design-Builder, Architect, Consultants, Contractors, or their agents or employees, or any other persons or entities performing portions of the Work for the Design-Builder.

§ 7.6 The Owner has the authority to reject Work that does not conform to the Design-Build Documents. The Owner shall have authority to require inspection or testing of the Work in accordance with Section 15.5.2, whether or not such Work is fabricated, installed or completed. However, neither this authority of the Owner nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Owner to the Design-Builder, the Architect, Consultants, Contractors, material and equipment suppliers, their agents or employees, or other persons or entities performing portions of the Work.

§ 7.7 The Owner shall determine the date or dates of Substantial Completion in accordance with Section 9.8 and the date of final completion in accordance with Section 9.10.

#### § 7.8 Owner's Right to Stop Work

If the Design-Builder fails to correct Work which is not in accordance with the requirements of the Design-Build Documents as required by Section 11.2 or persistently fails to carry out Work in accordance with the Design-Build Documents, the Owner may issue a written order to the Design-Builder to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Design-Builder or any other person or entity, except to the extent required by Section 5.13.1.3.

#### § 7.9 Owner's Right to Carry Out the Work

If the Design-Builder defaults or neglects to carry out the Work in accordance with the Design-Build Documents and fails within a ten-day period after receipt of written notice from the Owner to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to other remedies the Owner may have, correct such deficiencies. In such case, an appropriate Change Order shall be issued deducting from payments then or thereafter due the Design-Builder the reasonable cost of correcting such deficiencies. If payments then or thereafter due the Design-Builder are not sufficient to cover such amounts, the Design-Builder shall pay the difference to the Owner.

#### **ARTICLE 8** TIME

#### § 8.1 Progress and Completion

§ 8.1.1 Time limits stated in the Design-Build Documents are of the essence of the Contract. By executing the Design-Build Amendment the Design-Builder confirms that the Contract Time is a reasonable period for performing the Work.

§ 8.1.2 The Design-Builder shall not, except by agreement of the Owner in writing, commence the Work prior to the effective date of insurance, other than property insurance, required by this Contract. The Contract Time shall not be adjusted as a result of the Design-Builder's failure to obtain insurance required under this Contract

§ 8.1.3 The Design-Builder shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time.

#### § 8.2 Delays and Extensions of Time

§ 8.2.1 If the Design-Builder is delayed at any time in the commencement or progress of the Work by an act or neglect of the Owner or of a consultant or separate contractor employed by the Owner; or by changes ordered in the Work by the Owner; or by labor disputes, fire, unusual delay in deliveries, unavoidable casualties or other causes beyond the Design-Builder's control; or by delay authorized by the Owner pending mediation and binding dispute resolution or by other causes that the Owner determines may justify delay, then the Contract Time shall be extended by Change Order for such reasonable time as the Owner may determine.

§ 8.2.2 Claims relating to time shall be made in accordance with applicable provisions of Article 14.

§ 8.2.3 This Section 8.2 does not preclude recovery of damages for delay by either party under other provisions of the Design-Build Documents.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," %ATA," the ATA Logo, and "ATA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by ATA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the ATA Contract Documents® Terms of Service. To report copyright violations, 19 e-mail copyright@aia.org. User Notes:

#### ARTICLE 9 PAYMENT APPLICATIONS AND PROJECT COMPLETION

#### § 9.1 Contract Sum

The Contract Sum is stated in the Design-Build Amendment.

#### § 9.2 Schedule of Values

Where the Contract Sum is based on a stipulated sum or Guaranteed Maximum Price, the Design-Builder, prior to the first Application for Payment after execution of the Design-Build Amendment shall submit to the Owner a schedule of values allocating the entire Contract Sum to the various portions of the Work and prepared in such form and supported by such data to substantiate its accuracy as the Owner may require. This schedule, unless objected to by the Owner, shall be used as a basis for reviewing the Design-Builder's Applications for Payment.

#### § 9.3 Applications for Payment

§ 9.3.1 At least ten days before the date established for each progress payment, the Design-Builder shall submit to the Owner an itemized Application for Payment for completed portions of the Work. The application shall be notarized, if required, and supported by data substantiating the Design-Builder's right to payment as the Owner may require, such as copies of requisitions from the Architect, Consultants, Contractors, and material suppliers, and shall reflect retainage if provided for in the Design-Build Documents.

§ 9.3.1.1 As provided in Section 6.3.9, Applications for Payment may include requests for payment on account of changes in the Work that have been properly authorized by Change Directives, or by interim determinations of the Owner, but not yet included in Change Orders.

§ 9.3.1.2 Applications for Payment shall not include requests for payment for portions of the Work for which the Design-Builder does not intend to pay the Architect, Consultant, Contractor, material supplier, or other persons or entities providing services or work for the Design-Builder, unless such Work has been performed by others whom the Design-Builder intends to pay.

**§ 9.3.2** Unless otherwise provided in the Design-Build Documents, payments shall be made for services provided as well as materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Design-Builder with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include the costs of applicable insurance, storage and transportation to the site for such materials and equipment stored off the site.

**§ 9.3.3** The Design-Builder warrants that title to all Work, other than Instruments of Service, covered by an Application for Payment will pass to the Owner no later than the time of payment. The Design-Builder further warrants that, upon submittal of an Application for Payment, all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Design-Builder's knowledge, information and belief, be free and clear of liens, claims, security interests or encumbrances in favor of the Design-Builder, Architect, Consultants, Contractors, material suppliers, or other persons or entities entitled to make a claim by reason of having provided labor, materials and equipment relating to the Work.

#### § 9.4 Certificates for Payment

The Owner shall, within seven days after receipt of the Design-Builder's Application for Payment, issue to the Design-Builder a Certificate for Payment indicating the amount the Owner determines is properly due, and notify the Design-Builder in writing of the Owner's reasons for withholding certification in whole or in part as provided in Section 9.5.1.

#### § 9.5 Decisions to Withhold Certification

**§ 9.5.1** The Owner may withhold a Certificate for Payment in whole or in part to the extent reasonably necessary to protect the Owner due to the Owner's determination that the Work has not progressed to the point indicated in the Design-Builder's Application for Payment, or the quality of the Work is not in accordance with the Design-Build Documents. If the Owner is unable to certify payment in the amount of the Application, the Owner will notify the Design-Builder as provided in Section 9.4. If the Design-Builder and Owner cannot agree on a revised amount, the Owner will promptly issue a Certificate for Payment for the amount that the Owner deems to be due and owing. The

Owner may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued to such extent as may be necessary to protect the Owner from loss for which the Design-Builder is responsible because of

- .1 defective Work, including design and construction, not remedied;
- .2 third party claims filed or reasonable evidence indicating probable filing of such claims unless security acceptable to the Owner is provided by the Design-Builder;
- .3 failure of the Design-Builder to make payments properly to the Architect, Consultants, Contractors or others, for services, labor, materials or equipment;
- .4 reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum;
- .5 damage to the Owner or a separate contractor;
- .6 reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay; or
- .7 repeated failure to carry out the Work in accordance with the Design-Build Documents.

§ 9.5.2 When the above reasons for withholding certification are removed, certification will be made for amounts previously withheld.

§ 9.5.3 If the Owner withholds certification for payment under Section 9.5.1.3, the Owner may, at its sole option, issue joint checks to the Design-Builder and to the Architect or any Consultants, Contractor, material or equipment suppliers, or other persons or entities providing services or work for the Design-Builder to whom the Design-Builder failed to make payment for Work properly performed or material or equipment suitably delivered.

#### § 9.6 Progress Payments

§ 9.6.1 After the Owner has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Design-Build Documents.

**§ 9.6.2** The Design-Builder shall pay each Architect, Consultant, Contractor, and other person or entity providing services or work for the Design-Builder no later than the time period required by applicable law, but in no event more than seven days after receipt of payment from the Owner the amount to which the Architect, Consultant, Contractor, and other person or entity providing services or work for the Design-Builder is entitled, reflecting percentages actually retained from payments to the Design-Builder on account of the portion of the Work performed by the Architect, Consultant, Contractor, or other person or entity. The Design-Builder shall, by appropriate agreement with each Architect, Consultant, Contractor, and other person or entity providing services or work for the Design-Builder shall, by appropriate agreement with each Architect, Consultant, Contractor, and other person or entity providing services or work for the Design-Builder shall, by appropriate agreement with each Architect, Consultant, Contractor, and other person or entity providing services or work for the Design-Builder, require each Architect, Consultant, Contractor, and other person or entity providing services or work for the Design-Builder, Builder to make payments to subconsultants and subcontractors in a similar manner.

§ 9.6.3 The Owner will, on request and if practicable, furnish to the Architect, a Consultant, Contractor, or other person or entity providing services or work for the Design-Builder, information regarding percentages of completion or amounts applied for by the Design-Builder and action taken thereon by the Owner on account of portions of the Work done by such Architect, Consultant, Contractor or other person or entity providing services or work for the Design-Builder.

§ 9.6.4 The Owner has the right to request written evidence from the Design-Builder that the Design-Builder has properly paid the Architect, Consultants, Contractors, or other person or entity providing services or work for the Design-Builder, amounts paid by the Owner to the Design-Builder for the Work. If the Design-Builder fails to furnish such evidence within seven days, the Owner shall have the right to contact the Architect, Consultants, and Contractors to ascertain whether they have been properly paid. The Owner shall have no obligation to pay or to see to the payment of money to a Consultant or Contractor, except as may otherwise be required by law.

**§ 9.6.5** Design-Builder payments to material and equipment suppliers shall be treated in a manner similar to that provided in Sections 9.6.2, 9.6.3 and 9.6.4.

**§ 9.6.6** A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work not in accordance with the Design-Build Documents.

§ 9.6.7 Unless the Design-Builder provides the Owner with a payment bond in the full penal sum of the Contract Sum, payments received by the Design-Builder for Work properly performed by the Architect, Consultants, Contractors and

other person or entity providing services or work for the Design-Builder, shall be held by the Design-Builder for the Architect and those Consultants, Contractors, or other person or entity providing services or work for the Design-Builder, for which payment was made by the Owner. Nothing contained herein shall require money to be placed in a separate account and not commingled with money of the Design-Builder, shall create any fiduciary liability or tort liability on the part of the Design-Builder for breach of trust or shall entitle any person or entity to an award of punitive damages against the Design-Builder for breach of the requirements of this provision.

#### § 9.7 Failure of Payment

If the Owner does not issue a Certificate for Payment, through no fault of the Design-Builder, within the time required by the Design-Build Documents, then the Design-Builder may, upon seven additional days' written notice to the Owner, stop the Work until payment of the amount owing has been received. The Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Design-Builder's reasonable costs of shut-down, delay and start-up, plus interest as provided for in the Design-Build Documents.

#### § 9.8 Substantial Completion

§ 9.8.1 Substantial Completion is the stage in the progress of the Work when the Work or designated portion thereof is sufficiently complete in accordance with the Design-Build Documents so that the Owner can occupy or utilize the Work for its intended use. The date of Substantial Completion is the date certified by the Owner in accordance with this Section 9.8.

**§ 9.8.2** When the Design-Builder considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Design-Builder shall prepare and submit to the Owner a comprehensive list of items to be completed or corrected prior to final payment. Failure to include an item on such list does not alter the responsibility of the Design-Builder to complete all Work in accordance with the Design-Build Documents.

**§ 9.8.3** Upon receipt of the Design-Builder's list, the Owner shall make an inspection to determine whether the Work or designated portion thereof is substantially complete. If the Owner's inspection discloses any item, whether or not included on the Design-Builder's list, which is not sufficiently complete in accordance with the Design-Build Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Design-Builder shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Owner. In such case, the Design-Builder shall then submit a request for another inspection by the Owner to determine Substantial Completion.

**§ 9.8.4** Prior to issuance of the Certificate of Substantial Completion under Section 9.8.5, the Owner and Design-Builder shall discuss and then determine the parties' obligations to obtain and maintain property insurance following issuance of the Certificate of Substantial Completion.

**§ 9.8.5** When the Work or designated portion thereof is substantially complete, the Design-Builder will prepare for the Owner's signature a Certificate of Substantial Completion that shall, upon the Owner's signature, establish the date of Substantial Completion; establish responsibilities of the Owner and Design-Builder for security, maintenance, heat, utilities, damage to the Work and insurance; and fix the time within which the Design-Builder shall finish all items on the list accompanying the Certificate. Warranties required by the Design-Build Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion.

**§ 9.8.6** The Certificate of Substantial Completion shall be submitted by the Design-Builder to the Owner for written acceptance of responsibilities assigned to it in the Certificate. Upon the Owner's acceptance, and consent of surety, if any, the Owner shall make payment of retainage applying to the Work or designated portion thereof. Payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Design-Build Documents.

#### § 9.9 Partial Occupancy or Use

**§ 9.9.1** The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Design-Builder, provided such occupancy or use is consented to, by endorsement or otherwise, by the insurer providing property insurance and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Design-Builder have accepted in writing the responsibilities assigned to each of them for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work and

insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Design-Build Documents. When the Design-Builder considers a portion substantially complete, the Design-Builder shall prepare and submit a list to the Owner as provided under Section 9.8.2. Consent of the Design-Builder to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Design-Builder.

§ 9.9.2 Immediately prior to such partial occupancy or use, the Owner and Design-Builder shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.

§ 9.9.3 Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Design-Build Documents.

#### § 9.10 Final Completion and Final Payment

§ 9.10.1 Upon receipt of the Design-Builder's written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Owner will promptly make such inspection. When the Owner finds the Work acceptable under the Design-Build Documents and the Contract fully performed, the Owner will, subject to Section 9.10.2, promptly issue a final Certificate for Payment.

§ 9.10.2 Neither final payment nor any remaining retained percentage shall become due until the Design-Builder submits to the Owner (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work, for which the Owner or the Owner's property might be responsible or encumbered, (less amounts withheld by Owner) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Design-Build Documents to remain in force after final payment is currently in effect, (3) a written statement that the Design-Builder knows of no substantial reason that the insurance will not be renewable to cover the period required by the Design-Build Documents, (4) consent of surety, if any, to final payment, (5) as-constructed record copy of the Construction Documents marked to indicate field changes and selections made during construction, (6) manufacturer's warranties, product data, and maintenance and operations manuals, and (7) if required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts, or releases and waivers of liens, claims security interests, or encumbrances, arising out of the Contract, to the extent and in such form as may be designated by the Owner. If an Architect, a Consultant, or a Contractor, or other person or entity providing services or work for the Design-Builder, refuses to furnish a release or waiver required by the Owner, the Design-Builder may furnish a bond satisfactory to the Owner to indemnify the Owner against such liens, claims, security interests, or encumbrances. If such liens, claims, security interests, or encumbrances remains unsatisfied after payments are made, the Design-Builder shall refund to the Owner all money that the Owner may be compelled to pay in discharging such liens, claims, security interests, or encumbrances, including all costs and reasonable attorneys' fees.

§ 9.10.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Design-Builder or by issuance of Change Orders affecting final completion, the Owner shall, upon application by the Design-Builder, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance for Work not fully completed or corrected is less than retainage stipulated in the Design-Build Documents, and if bonds have been furnished, the written consent of surety to payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Design-Builder to the Owner prior to issuance of payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of claims.

§ 9.10.4 The making of final payment shall constitute a waiver of Claims by the Owner except those arising from

- .1 liens, Claims, security interests or encumbrances arising out of the Contract and unsettled;
- .2 failure of the Work to comply with the requirements of the Design-Build Documents; or
- .3 terms of special warranties required by the Design-Build Documents.

§ 9.10.5 Acceptance of final payment by the Design-Builder shall constitute a waiver of claims by the Design-Builder except those previously made in writing and identified by the Design-Builder as unsettled at the time of final Application for Payment.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 23 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes

#### ARTICLE 10 PROTECTION OF PERSONS AND PROPERTY

#### § 10.1 Safety Precautions and Programs

The Design-Builder shall be responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the performance of the Contract.

#### § 10.2 Safety of Persons and Property

§ 10.2.1 The Design-Builder shall be responsible for precautions for the safety of, and reasonable protection to prevent damage, injury or loss to

- .1 employees on the Work and other persons who may be affected thereby;
- .2 the Work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody or control of the Design-Builder or the Architect, Consultants, or Contractors, or other person or entity providing services or work for the Design-Builder; and
- .3 other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, or structures and utilities not designated for removal, relocation or replacement in the course of construction.

§ 10.2.2 The Design-Builder shall comply with, and give notices required by, applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities, bearing on safety of persons or property, or their protection from damage, injury or loss.

§ 10.2.3 The Design-Builder shall implement, erect, and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations, and notify owners and users of adjacent sites and utilities of the safeguards and protections.

§ 10.2.4 When use or storage of explosives or other hazardous materials or equipment, or unusual methods, are necessary for execution of the Work, the Design-Builder shall exercise utmost care, and carry on such activities under supervision of properly qualified personnel.

§ 10.2.5 The Design-Builder shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Design-Build Documents) to property referred to in Sections 10.2.1.2 and 10.2.1.3, caused in whole or in part by the Design-Builder, the Architect, a Consultant, a Contractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Design-Builder is responsible under Sections 10.2.1.2 and 10.2.1.3; except damage or loss attributable to acts or omissions of the Owner, or anyone directly or indirectly employed by the Owner, or by anyone for whose acts the Owner may be liable, and not attributable to the fault or negligence of the Design-Builder. The foregoing obligations of the Design-Builder are in addition to the Design-Builder's obligations under Section 3.1.14.

§ 10.2.6 The Design-Builder shall designate a responsible member of the Design-Builder's organization, at the site, whose duty shall be the prevention of accidents. This person shall be the Design-Builder's superintendent unless otherwise designated by the Design-Builder in writing to the Owner.

§ 10.2.7 The Design-Builder shall not permit any part of the construction or site to be loaded so as to cause damage or create an unsafe condition.

§ 10.2.8 Injury or Damage to Person or Property. If the Owner or Design-Builder suffers injury or damage to person or property because of an act or omission of the other, or of others for whose acts such party is legally responsible, written notice of the injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding 21 days after discovery. The notice shall provide sufficient detail to enable the other party to investigate the matter.

#### § 10.3 Hazardous Materials

§ 10.3.1 The Design-Builder is responsible for compliance with any requirements included in the Design-Build Documents regarding hazardous materials. If the Design-Builder encounters a hazardous material or substance not addressed in the Design-Build Documents and if reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated biphenyl (PCB), encountered on the site by the Design-Builder, the Design-Builder shall, upon

recognizing the condition, immediately stop Work in the affected area and report the condition to the Owner in writing.

**§ 10.3.2** Upon receipt of the Design-Builder's written notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Design-Builder and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Design-Build Documents, the Owner shall furnish in writing to the Design-Builder the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of such material or substance or who are to perform the task of removal or safe containment of such material or substance. The Design-Builder will promptly reply to the Owner in writing stating whether or not the Design-Builder has reasonable objection to the Owner, the Owner shall propose another to whom the Design-Builder has no reasonable objection. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Design-Builder. By Change Order, the Contract Time shall be extended appropriately and the Contract Sum shall be increased in the amount of the Design-Builder's reasonable additional costs of shut-down, delay and start-up.

**§ 10.3.3** To the fullest extent permitted by law, the Owner shall indemnify and hold harmless the Design-Builder, the Architect, Consultants, and Contractors, and employees of any of them, from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work in the affected area, if in fact the material or substance presents the risk of bodily injury or death as described in Section 10.3.1 and has not been rendered harmless, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease or death, or to injury to, or destruction of, tangible property (other than the Work itself), except to the extent that such damage, loss or expense is due to the fault or negligence of the party seeking indemnity.

**§ 10.3.4** The Owner shall not be responsible under this Section 10.3 for materials or substances the Design-Builder brings to the site unless such materials or substances are required by the Owner's Criteria. The Owner shall be responsible for materials or substances required by the Owner's Criteria, except to the extent of the Design-Builder's fault or negligence in the use and handling of such materials or substances.

§ 10.3.5 The Design-Builder shall indemnify the Owner for the cost and expense the Owner incurs (1) for remediation of a material or substance the Design-Builder brings to the site and negligently handles, or (2) where the Design-Builder fails to perform its obligations under Section 10.3.1, except to the extent that the cost and expense are due to the Owner's fault or negligence.

**§ 10.3.6** If, without negligence on the part of the Design-Builder, the Design-Builder is held liable by a government agency for the cost of remediation of a hazardous material or substance solely by reason of performing Work as required by the Design-Build Documents, the Owner shall indemnify the Design-Builder for all cost and expense thereby incurred.

#### § 10.4 Emergencies

In an emergency affecting safety of persons or property, the Design-Builder shall act, at the Design-Builder's discretion, to prevent threatened damage, injury or loss.

#### ARTICLE 11 UNCOVERING AND CORRECTION OF WORK

#### § 11.1 Uncovering of Work

The Owner may request to examine a portion of the Work that the Design-Builder has covered to determine if the Work has been performed in accordance with the Design-Build Documents. If such Work is in accordance with the Design-Build Documents, the Owner and Design-Builder shall execute a Change Order to adjust the Contract Time and Contract Sum, as appropriate. If such Work is not in accordance with the Design-Builder Documents, the costs of uncovering and correcting the Work shall be at the Design-Builder's expense and the Design-Builder shall not be entitled to a change in the Contract Time unless the condition was caused by the Owner or a separate contractor in which event the Owner shall be responsible for payment of such costs and the Contract Time will be adjusted as appropriate.

Powered by BoardOnTrack

AIA Document Al41<sup>®</sup> - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft yeas produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No. 4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. (2034726772)

#### § 11.2 Correction of Work

**§ 11.2.1 Before or After Substantial Completion.** The Design-Builder shall promptly correct Work rejected by the Owner or failing to conform to the requirements of the Design-Build Documents, whether discovered before or after Substantial Completion and whether or not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for any design consultant employed by the Owner whose expenses and compensation were made necessary thereby, shall be at the Design-Builder's expense.

#### § 11.2.2 After Substantial Completion

**§ 11.2.2.1** In addition to the Design-Builder's obligations under Section 3.1.12, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of an applicable special warranty required by the Design-Build Documents, any of the Work is found not to be in accordance with the requirements of the Design-Build Documents, the Design-Builder shall correct it promptly after receipt of written notice from the Owner to do so unless the Owner has previously given the Design-Builder a written acceptance of such condition. The Owner shall give such notice promptly after discovery of the condition. During the one-year period for correction of the Work, if the Owner fails to notify the Design-Builder and give the Design-Builder and to make a claim for breach of warranty. If the Design-Builder fails to correct nonconforming Work within a reasonable time during that period after receipt of notice from the Owner, the Owner may correct it in accordance with Section 7.9.

**§ 11.2.2.2** The one-year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual completion of that portion of the Work.

**§ 11.2.2.3** The one-year period for correction of Work shall not be extended by corrective Work performed by the Design-Builder pursuant to this Section 11.2.

**§ 11.2.3** The Design-Builder shall remove from the site portions of the Work that are not in accordance with the requirements of the Design-Build Documents and are neither corrected by the Design-Builder nor accepted by the Owner.

§ 11.2.4 The Design-Builder shall bear the cost of correcting destroyed or damaged construction of the Owner or separate contractors, whether completed or partially completed, caused by the Design-Builder's correction or removal of Work that is not in accordance with the requirements of the Design-Build Documents.

**§ 11.2.5** Nothing contained in this Section 11.2 shall be construed to establish a period of limitation with respect to other obligations the Design-Builder has under the Design-Build Documents. Establishment of the one-year period for correction of Work as described in Section 11.2.2 relates only to the specific obligation of the Design-Builder to correct the Work, and has no relationship to the time within which the obligation to comply with the Design-Build Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Design-Builder's liability with respect to the Design-Builder's obligations other than specifically to correct the Work.

#### § 11.3 Acceptance of Nonconforming Work

If the Owner prefers to accept Work that is not in accordance with the requirements of the Design-Build Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

#### ARTICLE 12 COPYRIGHTS AND LICENSES

**§ 12.1** Drawings, specifications, and other documents furnished by the Design-Builder, including those in electronic form, are Instruments of Service. The Design-Builder, and the Architect, Consultants, Contractors, and any other person or entity providing services or work for any of them, shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and shall retain all common law, statutory and other reserved rights, including copyrights. Submission or distribution of Instruments of Service to meet official regulatory requirements, or for similar purposes in connection with the Project, is not to be construed as publication in
derogation of the reserved rights of the Design-Builder and the Architect, Consultants, and Contractors, and any other person or entity providing services or work for any of them.

§ 12.2 The Design-Builder and the Owner warrant that in transmitting Instruments of Service, or any other information, the transmitting party is the copyright owner of such information or has permission from the copyright owner to transmit such information for its use on the Project.

**§ 12.3** Upon execution of the Agreement, the Design-Builder grants to the Owner a limited, irrevocable and non-exclusive license to use the Instruments of Service solely and exclusively for purposes of constructing, using, maintaining, altering and adding to the Project, provided that the Owner substantially performs its obligations, including prompt payment of all sums when due, under the Design-Build Documents. The license granted under this section permits the Owner to authorize its consultants and separate contractors to reproduce applicable portions of the Instruments of Service solely and exclusively for use in performing services or construction for the Project. If the Design-Builder rightfully terminates this Agreement for cause as provided in Section 13.1.4 or 13.2.1 the license granted in this Section 12.3 shall terminate.

**§ 12.3.1** The Design-Builder shall obtain non-exclusive licenses from the Architect, Consultants, and Contractors, that will allow the Design-Builder to satisfy its obligations to the Owner under this Article 12. The Design-Builder's licenses from the Architect and its Consultants and Contractors shall also allow the Owner, in the event this Agreement is terminated for any reason other than the default of the Owner or in the event the Design-Builder's Architect, Consultants, or Contractors terminate their agreements with the Design-Builder for cause, to obtain a limited, irrevocable and non-exclusive license solely and exclusively for purposes of constructing, using, maintaining, altering and adding to the Project, provided that the Owner (1) agrees to pay to the Architect, Consultant or Contractor all amounts due, and (2) provide the Architect, Consultant or Contractor from all costs and expenses, including the cost of defense, related to claims and causes of action asserted by any third person or entity to the extent such costs and expenses arise from the Owner's alteration or use of the Instruments of Service.

**§ 12.3.2** In the event the Owner alters the Instruments of Service without the author's written authorization or uses the Instruments of Service without retaining the authors of the Instruments of Service, the Owner releases the Design-Builder, Architect, Consultants, Contractors and any other person or entity providing services or work for any of them, from all claims and causes of action arising from or related to such uses. The Owner, to the extent permitted by law, further agrees to indemnify and hold harmless the Design-Builder, Architect, Consultants, Contractors and any other person or entity providing services or work for any other person or entity providing services or work for any other person or entity providing services or work for any of them, from all costs and expenses, including the cost of defense, related to claims and causes of action asserted by any third person or entity to the extent such costs and expenses arise from the Owner's alteration or use of the Instruments of Service under this Section 12.3.2. The terms of this Section 12.3.2 shall not apply if the Owner rightfully terminates this Agreement for cause under Sections 13.1.4 or 13.2.2.

#### ARTICLE 13 TERMINATION OR SUSPENSION

§ 13.1 Termination or Suspension Prior to Execution of the Design-Build Amendment

§ 13.1.1 If the Owner fails to make payments to the Design-Builder for Work prior to execution of the Design-Build Amendment in accordance with this Agreement, such failure shall be considered substantial nonperformance and cause for termination or, at the Design-Builder's option, cause for suspension of performance of services under this Agreement. If the Design-Builder elects to suspend the Work, the Design-Builder shall give seven days' written notice to the Owner before suspending the Work. In the event of a suspension of the Work, the Design-Builder shall have no liability to the Owner for delay or damage caused by the suspension of the Work. Before resuming the Work, the Design-Builder shall be paid all sums due prior to suspension and any expenses incurred in the interruption and resumption of the Design-Builder's Work. The Design-Builder's compensation for, and time to complete, the remaining Work shall be equitably adjusted.

**§ 13.1.2** If the Owner suspends the Project, the Design-Builder shall be compensated for the Work performed prior to notice of such suspension. When the Project is resumed, the Design-Builder shall be compensated for expenses incurred in the interruption and resumption of the Design-Builder's Work. The Design-Builder's compensation for, and time to complete, the remaining Work shall be equitably adjusted.

AIA Document Al41" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft 27 was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No. 4655459305 which expires on 01/25/2022, is not for resale, is licensed 27 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. (2034726772)

**§ 13.1.3** If the Owner suspends the Project for more than 90 cumulative days for reasons other than the fault of the Design-Builder, the Design-Builder may terminate this Agreement by giving not less than seven days' written notice.

**§ 13.1.4** Either party may terminate this Agreement upon not less than seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.

**§ 13.1.5** The Owner may terminate this Agreement upon not less than seven days' written notice to the Design-Builder for the Owner's convenience and without cause.

**§ 13.1.6** In the event of termination not the fault of the Design-Builder, the Design-Builder shall be compensated for Work performed prior to termination, together with Reimbursable Expenses then due and any other expenses directly attributable to termination for which the Design-Builder is not otherwise compensated. In no event shall the Design-Builder's compensation under this Section 13.1.6 be greater than the compensation set forth in Section 2.1.

### § 13.2 Termination or Suspension Following Execution of the Design-Build Amendment § 13.2.1 Termination by the Design-Builder

**§ 13.2.1.1** The Design-Builder may terminate the Contract if the Work is stopped for a period of 30 consecutive days through no act or fault of the Design-Builder, the Architect, a Consultant, or a Contractor, or their agents or employees, or any other persons or entities performing portions of the Work under direct or indirect contract with the Design-Builder, for any of the following reasons:

- .1 Issuance of an order of a court or other public authority having jurisdiction that requires all Work to be stopped;
- .2 An act of government, such as a declaration of national emergency that requires all Work to be stopped;
- .3 Because the Owner has not issued a Certificate for Payment and has not notified the Design-Builder of the reason for withholding certification as provided in Section 9.5.1, or because the Owner has not made payment on a Certificate for Payment within the time stated in the Design-Build Documents; or
- .4 The Owner has failed to furnish to the Design-Builder promptly, upon the Design-Builder's request, reasonable evidence as required by Section 7.2.7.

**§ 13.2.1.2** The Design-Builder may terminate the Contract if, through no act or fault of the Design-Builder, the Architect, a Consultant, a Contractor, or their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Design-Builder, repeated suspensions, delays or interruptions of the entire Work by the Owner as described in Section 13.2.3 constitute in the aggregate more than 100 percent of the total number of days scheduled for completion, or 120 days in any 365-day period, whichever is less.

§ 13.2.1.3 If one of the reasons described in Section 13.2.1.1 or 13.2.1.2 exists, the Design-Builder may, upon seven days' written notice to the Owner, terminate the Contract and recover from the Owner payment for Work executed, including reasonable overhead and profit, costs incurred by reason of such termination, and damages.

**§ 13.2.1.4** If the Work is stopped for a period of 60 consecutive days through no act or fault of the Design-Builder or any other persons or entities performing portions of the Work under contract with the Design-Builder because the Owner has repeatedly failed to fulfill the Owner's obligations under the Design-Build Documents with respect to matters important to the progress of the Work, the Design-Builder may, upon seven additional days' written notice to the Owner, terminate the Contract and recover from the Owner as provided in Section 13.2.1.3.

#### § 13.2.2 Termination by the Owner For Cause

§ 13.2.2.1 The Owner may terminate the Contract if the Design-Builder

- .1 fails to submit the Proposal by the date required by this Agreement, or if no date is indicated, within a reasonable time consistent with the date of Substantial Completion;
- .2 repeatedly refuses or fails to supply an Architect, or enough properly skilled Consultants, Contractors, or workers or proper materials;
- .3 fails to make payment to the Architect, Consultants, or Contractors for services, materials or labor in accordance with their respective agreements with the Design-Builder;
- .4 repeatedly disregards applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of a public authority; or
- .5 is otherwise guilty of substantial breach of a provision of the Design-Build Documents.

Powered by BoardOnTrack

§ 13.2.2.2 When any of the above reasons exist, the Owner may without prejudice to any other rights or remedies of the Owner and after giving the Design-Builder and the Design-Builder's surety, if any, seven days' written notice, terminate employment of the Design-Builder and may, subject to any prior rights of the surety:

- Exclude the Design-Builder from the site and take possession of all materials, equipment, tools, and .1 construction equipment and machinery thereon owned by the Design-Builder;
- .2 Accept assignment of the Architect, Consultant and Contractor agreements pursuant to Section 3.1.15; and
- .3 Finish the Work by whatever reasonable method the Owner may deem expedient. Upon written request of the Design-Builder, the Owner shall furnish to the Design-Builder a detailed accounting of the costs incurred by the Owner in finishing the Work.

§ 13.2.2.3 When the Owner terminates the Contract for one of the reasons stated in Section 13.2.2.1, the Design-Builder shall not be entitled to receive further payment until the Work is finished.

§ 13.2.2.4 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Design-Builder. If such costs and damages exceed the unpaid balance, the Design-Builder shall pay the difference to the Owner. The obligation for such payments shall survive termination of the Contract.

#### § 13.2.3 Suspension by the Owner for Convenience

§ 13.2.3.1 The Owner may, without cause, order the Design-Builder in writing to suspend, delay or interrupt the Work in whole or in part for such period of time as the Owner may determine.

§ 13.2.3.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay or interruption as described in Section 13.2.3.1. Adjustment of the Contract Sum shall include profit. No adjustment shall be made to the extent

- that performance is, was or would have been so suspended, delayed or interrupted by another cause for .1 which the Design-Builder is responsible; or
- .2 that an equitable adjustment is made or denied under another provision of the Contract.

#### § 13.2.4 Termination by the Owner for Convenience

§ 13.2.4.1 The Owner may, at any time, terminate the Contract for the Owner's convenience and without cause.

§ 13.2.4.2 Upon receipt of written notice from the Owner of such termination for the Owner's convenience, the Design-Builder shall

- cease operations as directed by the Owner in the notice; .1
- take actions necessary, or that the Owner may direct, for the protection and preservation of the Work; .2 and.
- except for Work directed to be performed prior to the effective date of termination stated in the notice, .3 terminate all existing Project agreements, including agreements with the Architect, Consultants, Contractors, and purchase orders, and enter into no further Project agreements and purchase orders.

§ 13.2.4.3 In case of such termination for the Owner's convenience, the Design-Builder shall be entitled to receive payment for Work executed, and costs incurred by reason of such termination, along with reasonable overhead and profit on the Work not executed.

#### ARTICLE 14 CLAIMS AND DISPUTE RESOLUTION

#### § 14.1 Claims

§ 14.1.1 Definition. A Claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money, or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and Design-Builder arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim.

§ 14.1.2 Time Limits on Claims. The Owner and Design-Builder shall commence all claims and causes of action, whether in contract, tort, breach of warranty or otherwise, against the other, arising out of or related to the Contract in accordance with the requirements of the binding dispute resolution method selected in Section 1.3, within the time

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "ATA," the ATA Logo, and "ATA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by ATA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 29 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes (2034726772)

period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Design-Builder waive all claims and causes of action not commenced in accordance with this Section 14.1.2.

#### § 14.1.3 Notice of Claims

§ 14.1.3.1 Prior To Final Payment. Prior to Final Payment, Claims by either the Owner or Design-Builder must be initiated by written notice to the other party within 21 days after occurrence of the event giving rise to such Claim or within 21 days after the claimant first recognizes the condition giving rise to the Claim, whichever is later.

§ 14.1.3.2 Claims Arising After Final Payment. After Final Payment, Claims by either the Owner or Design-Builder that have not otherwise been waived pursuant to Sections 9.10.4 or 9.10.5, must be initiated by prompt written notice to the other party. The notice requirement in Section 14.1.3.1 and the Initial Decision requirement as a condition precedent to mediation in Section 14.2.1 shall not apply.

§ 14.1.4 Continuing Contract Performance. Pending final resolution of a Claim, except as otherwise agreed in writing or as provided in Section 9.7 and Article 13, the Design-Builder shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Design-Build Documents.

§ 14.1.5 Claims for Additional Cost. If the Design-Builder intends to make a Claim for an increase in the Contract Sum, written notice as provided herein shall be given before proceeding to execute the portion of the Work that relates to the Claim. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Section 10.4.

#### § 14.1.6 Claims for Additional Time

§ 14.1.6.1 If the Design-Builder intends to make a Claim for an increase in the Contract Time, written notice as provided herein shall be given. The Design-Builder's Claim shall include an estimate of cost and of probable effect of delay on progress of the Work. In the case of a continuing delay, only one Claim is necessary.

§ 14.1.6.2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time, could not have been reasonably anticipated, and had an adverse effect on the scheduled construction.

#### § 14.1.7 Claims for Consequential Damages

The Design-Builder and Owner waive Claims against each other for consequential damages arising out of or relating to this Contract. This mutual waiver includes

- damages incurred by the Owner for rental expenses, for losses of use, income, profit, financing, .1 business and reputation, and for loss of management or employee productivity or of the services of such persons; and
- .2 damages incurred by the Design-Builder for principal office expenses including the compensation of personnel stationed there, for losses of financing, business and reputation, and for loss of profit except anticipated profit arising directly from the Work.

This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination in accordance with Article 13. Nothing contained in this Section 14.1.7 shall be deemed to preclude an award of liquidated damages, when applicable, in accordance with the requirements of the Design-Build Documents.

#### § 14.2 Initial Decision

§ 14.2.1 An initial decision shall be required as a condition precedent to mediation of all Claims between the Owner and Design-Builder initiated prior to the date final payment is due, excluding those arising under Sections 10.3 and 10.4 of the Agreement and Sections B.3.2.9 and B.3.2.10 of Exhibit B to this Agreement, unless 30 days have passed after the Claim has been initiated with no decision having been rendered. Unless otherwise mutually agreed in writing, the Owner shall render the initial decision on Claims.

#### § 14.2.2 Procedure

§ 14.2.2.1 Claims Initiated by the Owner. If the Owner initiates a Claim, the Design-Builder shall provide a written response to Owner within ten days after receipt of the notice required under Section 14.1.3.1. Thereafter, the Owner shall render an initial decision within ten days of receiving the Design-Builder's response: (1) withdrawing the Claim

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," %ATA," the ATA Logo, and "ATA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by ATA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the ATA Contract Documents® Terms of Service. To report copyright violations, 30 e-mail copyright@aia.org. User Notes: (2034726772)

in whole or in part, (2) approving the Claim in whole or in part, or (3) suggesting a compromise.

§ 14.2.2.2 Claims Initiated by the Design-Builder. If the Design-Builder initiates a Claim, the Owner will take one or more of the following actions within ten days after receipt of the notice required under Section 14.1.3.1: (1) request additional supporting data, (2) render an initial decision rejecting the Claim in whole or in part, (3) render an initial decision approving the Claim, (4) suggest a compromise or (5) indicate that it is unable to render an initial decision because the Owner lacks sufficient information to evaluate the merits of the Claim.

§ 14.2.3 In evaluating Claims, the Owner may, but shall not be obligated to, consult with or seek information from persons with special knowledge or expertise who may assist the Owner in rendering a decision. The retention of such persons shall be at the Owner's expense.

§ 14.2.4 If the Owner requests the Design-Builder to provide a response to a Claim or to furnish additional supporting data, the Design-Builder shall respond, within ten days after receipt of such request, and shall either (1) provide a response on the requested supporting data, (2) advise the Owner when the response or supporting data will be furnished or (3) advise the Owner that no supporting data will be furnished. Upon receipt of the response or supporting data, if any, the Owner will either reject or approve the Claim in whole or in part.

§ 14.2.5 The Owner's initial decision shall (1) be in writing; (2) state the reasons therefor; and (3) identify any change in the Contract Sum or Contract Time or both. The initial decision shall be final and binding on the parties but subject to mediation and, if the parties fail to resolve their dispute through mediation, to binding dispute resolution.

§ 14.2.6 Either party may file for mediation of an initial decision at any time, subject to the terms of Section 14.2.6.1.

§ 14.2.6.1 Either party may, within 30 days from the date of an initial decision, demand in writing that the other party file for mediation within 60 days of the initial decision. If such a demand is made and the party receiving the demand fails to file for mediation within the time required, then both parties waive their rights to mediate or pursue binding dispute resolution proceedings with respect to the initial decision.

§ 14.2.7 In the event of a Claim against the Design-Builder, the Owner may, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Design-Builder's default, the Owner may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy.

§ 14.2.8 If a Claim relates to or is the subject of a mechanic's lien, the party asserting such Claim may proceed in accordance with applicable law to comply with the lien notice or filing deadlines.

#### § 14.3 Mediation

§ 14.3.1 Claims, disputes, or other matters in controversy arising out of or related to the Contract, except those waived as provided for in Sections 9.10.4, 9.10.5, and 14.1.7, shall be subject to mediation as a condition precedent to binding dispute resolution.

§ 14.3.2 The parties shall endeavor to resolve their Claims by mediation which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of the Agreement. A request for mediation shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of binding dispute resolution proceedings but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration proceeding is stayed pursuant to this Section 14.3.2, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.

§ 14.3.3 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 31 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes

#### § 14.4 Arbitration

**§ 14.4.1** If the parties have selected arbitration as the method for binding dispute resolution in Section 1.3, any Claim subject to, but not resolved by, mediation shall be subject to arbitration which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of the Agreement. A demand for arbitration shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the arbitration. The party filing a notice of demand for arbitration must assert in the demand all Claims then known to that party on which arbitration is permitted to be demanded.

**§ 14.4.1.1** A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the Claim would be barred by the applicable statute of limitations or statute of repose. For statute of limitations or statute of repose, receipt of a written demand for arbitration by the person or entity administering the arbitration shall constitute the institution of legal or equitable proceedings based on the Claim.

**§ 14.4.2** The award rendered by the arbitrator or arbitrators shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction.

**§ 14.4.3** The foregoing agreement to arbitrate, and other agreements to arbitrate with an additional person or entity duly consented to by parties to the Agreement, shall be specifically enforceable under applicable law in any court having jurisdiction thereof.

#### § 14.4.4 Consolidation or Joinder

**§ 14.4.1** Either party, at its sole discretion, may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation, (2) the arbitrations to be consolidated substantially involve common questions of law or fact, and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).

**§ 14.4.4.2** Either party, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.

**§ 14.4.4.3** The Owner and Design-Builder grant to any person or entity made a party to an arbitration conducted under this Section 14.4, whether by joinder or consolidation, the same rights of joinder and consolidation as the Owner and Design-Builder under this Agreement.

#### ARTICLE 15 MISCELLANEOUS PROVISIONS

#### § 15.1 Governing Law

The Contract shall be governed by the law of the place where the Project is located except that, if the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 14.4.

#### § 15.2 Successors and Assigns

§ 15.2.1 The Owner and Design-Builder, respectively, bind themselves, their partners, successors, assigns and legal representatives to the covenants, agreements and obligations contained in the Design-Build Documents. Except as provided in Section 15.2.2, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.

**§ 15.2.2** The Owner may, without consent of the Design-Builder, assign the Contract to a lender providing construction financing for the Project, if the lender assumes the Owner's rights and obligations under the Design-Build Documents. The Design-Builder shall execute all consents reasonably required to facilitate such assignment.

**§ 15.2.3** If the Owner requests the Design-Builder, Architect, Consultants, or Contractors to execute certificates, other than those required by Section 3.1.10, the Owner shall submit the proposed language of such certificates for review at least 14 days prior to the requested dates of execution. If the Owner requests the Design-Builder, Architect,

Consultants, or Contractors to execute consents reasonably required to facilitate assignment to a lender, the Design-Builder, Architect, Consultants, or Contractors shall execute all such consents that are consistent with this Agreement, provided the proposed consent is submitted to them for review at least 14 days prior to execution. The Design-Builder, Architect, Consultants, and Contractors shall not be required to execute certificates or consents that would require knowledge, services or responsibilities beyond the scope of their services.

#### § 15.3 Written Notice

Written notice shall be deemed to have been duly served if delivered in person to the individual, to a member of the firm or entity, or to an officer of the corporation for which it was intended; or if delivered at, or sent by registered or certified mail or by courier service providing proof of delivery to, the last business address known to the party giving notice.

#### § 15.4 Rights and Remedies

§ 15.4.1 Duties and obligations imposed by the Design-Build Documents, and rights and remedies available thereunder, shall be in addition to and not a limitation of duties, obligations, rights and remedies otherwise imposed or available by law.

**§ 15.4.2** No action or failure to act by the Owner or Design-Builder shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach thereunder, except as may be specifically agreed in writing.

#### § 15.5 Tests and Inspections

**§ 15.5.1** Tests, inspections and approvals of portions of the Work shall be made as required by the Design-Build Documents and by applicable laws, statutes, ordinances, codes, rules and regulations or lawful orders of public authorities. Unless otherwise provided, the Design-Builder shall make arrangements for such tests, inspections and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections and approvals. The Design-Builder shall give the Owner timely notice of when and where tests and inspections are to be made so that the Owner may be present for such procedures. The Owner shall bear costs of (1) tests, inspections or approvals that do not become requirements until after bids are received or negotiations concluded, and (2) tests, inspections or approvals where building codes or applicable laws or regulations prohibit the Owner from delegating their cost to the Design-Builder.

**§ 15.5.2** If the Owner determines that portions of the Work require additional testing, inspection or approval not included under Section 15.5.1, the Owner will instruct the Design-Builder to make arrangements for such additional testing, inspection or approval by an entity acceptable to the Owner, and the Design-Builder shall give timely notice to the Owner of when and where tests and inspections are to be made so that the Owner may be present for such procedures. Such costs, except as provided in Section 15.5.3, shall be at the Owner's expense.

**§ 15.5.3** If such procedures for testing, inspection or approval under Sections 15.5.1 and 15.5.2 reveal failure of the portions of the Work to comply with requirements established by the Design-Build Documents, all costs made necessary by such failure shall be at the Design-Builder's expense.

**§ 15.5.4** Required certificates of testing, inspection or approval shall, unless otherwise required by the Design-Build Documents, be secured by the Design-Builder and promptly delivered to the Owner.

**§ 15.5.5** If the Owner is to observe tests, inspections or approvals required by the Design-Build Documents, the Owner will do so promptly and, where practicable, at the normal place of testing.

**§ 15.5.6** Tests or inspections conducted pursuant to the Design-Build Documents shall be made promptly to avoid unreasonable delay in the Work.

#### § 15.6 Confidential Information

If the Owner or Design-Builder transmits Confidential Information, the transmission of such Confidential Information constitutes a warranty to the party receiving such Confidential Information that the transmitting party is authorized to transmit the Confidential Information. If a party receives Confidential Information, the receiving party shall keep the Confidential Information strictly confidential and shall not disclose it to any other person or entity except as set forth in Section 15.6.1.

**§ 15.6.1** A party receiving Confidential Information may disclose the Confidential Information as required by law or court order, including a subpoena or other form of compulsory legal process issued by a court or governmental entity. A party receiving Confidential Information may also disclose the Confidential Information to its employees, consultants or contractors in order to perform services or work solely and exclusively for the Project, provided those employees, consultants and contractors are subject to the restrictions on the disclosure and use of Confidential Information as set forth in this Contract.

#### § 15.7 Capitalization

Terms capitalized in the Contract include those that are (1) specifically defined, (2) the titles of numbered articles or (3) the titles of other documents published by the American Institute of Architects.

#### § 15.8 Interpretation

**§ 15.8.1** In the interest of brevity the Design-Build Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

**§ 15.8.2** Unless otherwise stated in the Design-Build Documents, words which have well-known technical or construction industry meanings are used in the Design-Build Documents in accordance with such recognized meanings.

#### ARTICLE 16 SCOPE OF THE AGREEMENT

§ 16.1 This Agreement is comprised of the following documents listed below:

- .1 AIA Document A141<sup>TM</sup>–2014, Standard Form of Agreement Between Owner and Design-Builder
- .2 AIA Document A141<sup>TM</sup>–2014, Exhibit A, Design-Build Amendment, if executed
- .3 AIA Document A141<sup>TM</sup>–2014, Exhibit B, Insurance and Bonds
- .4 AIA Document A141<sup>TM</sup>–2014, Exhibit C, Sustainable Projects, if completed
- .5 AIA Document E203<sup>™</sup>–2013, Building Information Modeling and Digital Data Exhibit, if completed, or the following:

« »

.6 Other:

« »

This Agreement entered into as of the day and year first written above.

**OWNER** (Signature)

« »« »

(Printed name and title)

#### **DESIGN-BUILDER** (Signature)

« »« »

(Printed name and title)

- i. Adequate off-site public and on-site private fire hydrants may be required. Their number and location to be determined after the Fire Department's review of the plot plan.
- j. All parking restrictions for fire lanes shall be posted and/or painted prior to any Temporary Certificate of Occupancy being issued.
- k. Plans showing areas to be posted and/or painted, "FIRE LANE NO PARKING" shall be submitted and approved by the Fire Department prior to building permit application signoff.
- I. Electric Gates approved by the Fire Department shall be tested by the Fire Department prior to Building and Safety granting a Certificate of Occupancy.
- m. That in order to provide assurance that the proposed common fire lane and fire protection facilities, for the project, not maintained by the City, are properly and adequately maintained, the sub-divider shall record with the County Recorder, prior to the recordation of the final map, a covenant and agreement (Planning Department General Form CP-6770) to assure the following:
  - i. The establishment of a property owners association, which shall cause a yearly inspection to be, made by a registered civil engineer of all common fire lanes and fire protection facilities. The association will undertake any necessary maintenance and corrective measures. Each future property owner shall automatically become a member of the association or organization required above and is automatically subject to a proportionate share of the cost.
  - ii. The future owners of affected lots with common fire lanes and fire protection facilities shall be informed or their responsibility for the maintenance of the devices on their lots. The future owner and all successors will be presented with a copy of the maintenance program for their lot. Any amendment or modification that would defeat the obligation of said association as the Advisory Agency must approve required hereinabove in writing after consultation with the Fire Department.
  - iii. In the event that the property owners association fails to maintain the common property and easements as required by the CC and R's, the individual property owners shall be responsible for their proportional share of the maintenance.
  - iv. Prior to any building permits being issued, the applicant shall improve, to the satisfaction of the Fire Department, all common fire lanes and install all private fire hydrants to be required.
  - v. That the Common Fire Lanes and Fire Protection facilities be shown on the Final Map.
  - vi. The plot plans shall be approved by the Fire Department showing fire hydrants and access for each phase of the project prior to the recording of the final map for that phase. Each phase shall comply independently with code requirements.

Note: The applicant is further advised that all subsequent contact regarding these conditions must be with the Hydrant and Access Unit. This would include clarification, verification of condition compliance and plans or building permit applications, etc., and shall be accomplished <u>by appointment only</u>, in order to assure that you receive service with a minimum amount of waiting please call **(818) 374-4351.** You should advise any consultant representing you of this requirement as well.

### CONDITIONS OF APPROVAL

Pursuant to Section 13.08 of the Los Angeles Municipal Code, the following conditions are hereby imposed upon the use of the subject property:

- 1. **Development.** The use and development of the property shall be in substantial conformance with the plot plan submitted with the application and marked Exhibit "A" except as may be revised as a result of this action.
- 2. Height. The height of the solar shade structure shall not exceed 25 feet.
- 3. **Setbacks.** A 15 foot setback along the southern property line shall be maintained with landscaping and trees.
- 4. Landscape Plan. Landscaping shall be provided in substantial compliance with the submitted Landscape Plan (as seen in Exhibit A). Landscaped areas shall be planted with a variety of plant materials which include shrubs, trees and ground cover, and all plants and trees shall be drought-resistant. Artificial plants are prohibited. All landscaped areas shall be equipped with an automatic sprinkler or drip irrigation system designed to conserve water.
- 5. **Mechanical Equipment**. All mechanical equipment on the roof shall be screened from view. The transformer, if visible from the ally or adjoining properties, shall be screened with landscaping.
- 6. **Lighting.** Lighting shall be installed in the parking lot and within the covered and uncovered recreation areas. Outdoor lighting shall be designed and installed with shielding, such that the light source cannot be seen from adjacent residential properties or the public right-of-way, nor from above.
- 7. **Building Materials.** A gray and blue color scheme as proposed in Exhibit "A" shall be used. No florescent colors shall be used.

#### 8. Fence and Wall Materials.

- a. The fence along the east and west property lines shall be combination wrought iron and masonry wall. Chain link fencing is prohibited.
- b. Any masonry walls along any property line should be constructed from decorative brick, stone, split face concrete block, or other decorative material. Masonry block walls should be finished with a masonry cap.
- c. The use of barbed wire, razor wire, and concertina wire is prohibited.
- 9. **Maintenance**. The subject property (including all trash storage areas, parking facilities, sidewalks, yard areas, parkways, and exterior walls and fences along the property lines) shall be maintained in an attractive condition and shall be kept free of trash and debris.
- 10. **Solar Panels.** Solar panels shall be installed on the rooftop of the shade structure to be connected to the school building's electrical system. The entirety of the roof area shall be reserved for the installation of a solar photovoltaic system, to be installed prior to the issuance of a certificate of occupancy, in substantial conformance with the plans stamped "Exhibit A".

#### Administrative Conditions of Approval

- 11. **Approval, Verification and Submittals.** Copies of any approvals, guarantees or verification of consultations, review or approval, plans, etc., as may be required by the subject conditions, shall be provided to the Department of City Planning for placement in the subject file.
- 12. **Code Compliance.** Area, height and use regulations of the (T)(Q)RD3-1 Zone classification of the subject property shall be complied with, except where herein conditions are more restrictive.
- 13. **Covenant.** Prior to the issuance of any permits relative to this matter, an agreement concerning all the information contained in these conditions shall be recorded in the County Recorder's Office. The agreement shall run with the land and shall be binding on any subsequent property owners, heirs or assign. The agreement must be submitted to the Department of City Planning for approval before being recorded. After recordation, a copy bearing the Recorder's number and date shall be provided to the Department of City Planning for attachment to the file.
- 14. **Definition.** Any agencies, public officials or legislation referenced in these conditions shall mean those agencies, public officials, legislation or their successors, designees or amendment to any legislation.
- 15. **Enforcement.** Compliance with these conditions and the intent of these conditions shall be to the satisfaction of the Department of City Planning and any designated agency, or the agency's successor and in accordance with any stated laws or regulations, or any amendments thereto.
- 16. **Building Plans.** Page 1 of the grants and all the conditions of approval shall be printed on the building plans submitted to the Department of City Planning and the Department of Building and Safety.
- 17. **Corrective Conditions.** The authorized use shall be conducted at all time with due regards to the character of the surrounding district, and the right is reserved to the North Valley Area Planning Commission, or the Director pursuant to Section 12.27.1 of the Municipal Code to impose additional corrective conditions, if in the Commission's or Director's opinion such conditions are proven necessary for the protection of persons in the neighborhood or occupants of adjacent property.
- 18. **Expediting Processing Section.** Prior to the clearance of any conditions, the applicant shall show that all fees have been paid to the Department of City Planning Expedited Processing Section.

#### 19. Indemnification and Reimbursement of Litigation Costs.

Applicant shall do all of the following:

a. Defend, indemnify and hold harmless the City from any and all actions against the City relating to or arising out of, in whole or in part, the City's processing and approval of this entitlement, including <u>but not limited to</u>, an action to attack, challenge, set aside, void or otherwise modify or annul the approval of the entitlement, the environmental review of the entitlement, or the approval of subsequent permit decisions or to claim personal property damage, including from inverse condemnation or any other constitutional claim.

- b. Reimburse the City for any and all costs incurred in defense of an action related to or arising out of, in whole or in part, the City's processing and approval of the entitlement, including but not limited to payment of all court costs and attorney's fees, costs of any judgments or awards against the City (including an award of attorney's fees), damages and/or settlement costs.
- c. Submit an initial deposit for the City's litigation costs to the City within 10 days' notice of the City tendering defense to the Applicant and requesting a deposit. The initial deposit shall be in an amount set by the City Attorney's Office, in its sole discretion, based on the nature and scope of action, but in no event shall the initial deposit be less than \$50,000. The City's failure to notice or collect the deposit does not relieve the Applicant from responsibility to reimburse the City pursuant to the requirement in paragraph (b).
- d. Submit supplemental deposits upon notice by the City. Supplemental deposits may be required in an increased amount from the initial deposit if found necessary by the City to protect the City's interests. The City's failure to notice or collect the deposit does not relieve the Applicant from responsibility to reimburse the City pursuant to the requirement (b).
- e. If the City determines it necessary to protect the City's interests, execute an indemnity and reimbursement agreement with the City under terms consistent with the requirements of this condition.

The City shall notify the applicant within a reasonable period of time of its receipt of any action and the City shall cooperate in the defense. If the City fails to notify the applicant of any claim, action or proceeding in a reasonable time, or if the City fails to reasonably cooperate in the defense, the applicant shall not thereafter be responsible to defend, indemnify or hold harmless the City.

The City shall have the sole right to choose its counsel, including the City Attorney's office or outside counsel. At its sole discretion, the City may participate at its own expense in the defense of any action, but such participation shall not relieve the applicant of any obligation imposed by this condition. In the event the Applicant fails to comply with this condition, in whole or in part, the City may withdraw its defense of the action, void its approval of the entitlement, or take any other action. The City retains the right to make all decisions with respect to its representations in any legal proceeding, including its inherent right to abandon or settle litigation.

For purposes of this condition, the following definitions apply:

"City" shall be defined to include the City, its agents, officers, boards, commission, committees, employees and volunteers.

"Action" shall be defined to include suits, proceedings (including those held under alternative dispute resolution procedures), claims or lawsuits. Actions includes actions, as defined herein, alleging failure to comply with any federal, state or local law.

Nothing in the definitions included in this paragraph are intended to limit the rights of the City or the obligations of the Applicant otherwise created by this condition.

### FINDINGS

#### General Plan/Charter Findings

#### 1. General Plan Land Use Designation.

General Plan Land Use Designation. The subject property is located within the Reseda – West Van Nuys Community Plan area which was updated by the City Council on November 17, 1999 and designates the subject property for Community Commercial land uses corresponding to the CR, C2, C4, RAS3, RAS4, P and PB Zones. The site is zoned with the [Q]P-1L-CDO Zone. The proposed C2 zone is a corresponding zone for the Community Commercial land use designation. The proposed outdoor recreation facilities, bathroom with storage area, and solar shade structure is consistent with development permitted in the proposed (T)[Q]C2-1L-CDO Zone and is consistent with the regulations of the qualified Q condition. Therefore, the project is in substantial conformance with the purposes, intent and provisions of the General Plan as reflected in the adopted Framework Element and Community Plan.

#### 2. General Plan Text

a. The **Framework Element** of the General Plan (Framework Element) was adopted by the City of Los Angeles in December 1996 and re-adopted in August 2001. The Framework Element provides guidance regarding policy issues for the entire City of Los Angeles, including the project site. The Framework Element also sets forth a Citywide comprehensive long-range growth strategy and defines Citywide polices regarding such issues as land use, housing, urban form, neighborhood design, open space, economic development, transportation, infrastructure, and public services.

The Framework Element Long Range Land Use Diagram for the San Fernando Valley identifies Sherman Way in Reseda as a "Community Center." Community Centers are a focal point for surrounding residential neighborhoods and contain a diversity of uses such as small offices and overnight accommodations, cultural and entertainment facilities, and schools and libraries, in addition to neighborhood oriented services. The proposed Zone Change to a C2 zone is consistent with a "Community Center" and the proposed school recreation facilities are consist with the long range vision for the area.

The Framework Element includes the following Goals, Objectives and Policies relevant to the request:

GOAL 9N: Public schools that provide a quality education for all of the City's children, including those with special needs, and adequate school facilities to serve every neighborhood in the City so that students have an opportunity to attend school in their neighborhoods.

Objective 9.32 Work constructively with LAUSD to promote the siting and construction of adequate school facilities phased with growth.

Policy 9.32.2: Explore creative alternatives for providing new school sites in the City, where appropriate.

The proposed Zone Change from [Q]P-1L-CDO to (T)[Q]C2-1L-CDO will allow for the development of a recreational facility where an underutilized surface parking is currently located. The project is in conjunction with Magnolia Science Academy 1, an LAUSD

Charter School, which has been conducting student lunch, student recreation and dropoff/pick-up actives incidental to the school, in the surface parking lot through Zone Variances since 2006 (ZA-2005-3787-ZV, ZA-2008-0748-ZV, ZA-2014-0995-ZV). The Zone Change is consistent with Framework Policy 9.32.2 as it would allow an underutilized site to be developed into a permanent school facility.

Therefore, the Zone Change is consistent with the Distribution of Land Use goals, objectives and policies of the General Plan Framework Element.

b. **Reseda – West Van Nuys Community Plan.** The Community Plan text includes the following relevant land use Objectives and Policies:

GOAL 4: PUBLIC SCHOOLS THAT PROVIDE A QUALITY EDUCATION FOR ALL OF THE CITY'S CHILDREN, INCLUDING THOSE WITH SPECIAL NEEDS, AND ADEQUATE SCHOOL FACILITIES TO SERVE EVERY NEIGHBORHOOD IN THE CITY.

Objective 4-1: Work constructively with LAUSD to promote the siting and construction of adequate school facilities phased with growth.

Policy 4-1.1: Explore creative alternatives for providing new school sites in the City, where appropriate.

The proposed Zone Change from [Q]P-1L-CDO to (T)[Q]C2-1L-CDO will allow for the redevelopment of an underutilized surface parking lot with proposed recreational facilities in conjunction with the Magnolia Science Academy 1, an LAUSD Charter School. The school has been conducting student lunch, student recreation and drop-off/pick-up actives incidental to the school, in the surface parking lot through Zone Variances since 2006 (ZA-2005-3787-ZV, ZA-2008-0748-ZV, ZA-2014-0995-ZV). The Zone Change would allow for the installation of landscaping and a natural turf lawn and the construction of a permanent structure for active recreation and a lunch area. Therefore the Zone Change is consistent with Framework Policy 9.32.2 as it would allow an underutilized site to be developed into a permanent school facility.

- c. The **Mobility Element** of the General Plan (Mobility Plan 2035) is not likely to be affected by the recommended action herein. The site abuts an alley to the north. The alley is dedicated to a width of 20 feet and improved with asphalt roadway. As part of the (T) Tentative Classification conditions incorporated herein, the applicant must reconstruct the 20-foot wide alley and longitudinal concrete gutter as necessary to provide safe access for students and pedestrian traffic. The site does not abut any other public right-of-way.
- d. The **Sewerage Facilities Element** of the General Plan will not be affected by the recommended action. While the sewer system might be able to accommodate the total flows for the proposed project, further detailed gauging and evaluation may be needed as part of the permit process to identify a specific sewer connection point. If the public sewer has insufficient capacity then the developer will be required to build sewer lines to a point in the sewer system with sufficient capacity. A final approval for sewer capacity and connection permit will be made at that time. Ultimately, this sewage flow will be conveyed to the Hyperion Treatment Plant, which has sufficient capacity for the project.

#### Zone Change and "T"/"Q" Classification Findings

### 3. Pursuant to Section 12.32 of the Municipal Code, the zone change and classifications are necessary because:

- a. <u>Public Necessity</u>: On April 29, 2019, Mayor Eric Garcetti released LA's Green New Deal (Sustainable City pLAn), a roadmap to achieve short-term results while setting the path to strengthen the transformation of the City in the decades to come. As part of the plan, the Mayor set forth a goals to supply 100% renewable energy by 2045 and increase tree canopy in areas of greatest need by at least 50% by 2028 to grow a more equitable urban forest that provides cooling, public health, habitat, energy savings, and other benefits. The Zone Change would allow an asphalt surface parking lot site to be redeveloped into an outdoor recreation facility with a solar shade structure and many new trees. By providing solar panels and increasing the tree canopy coverage on site, the project is aligned with the city's sustainability goals and is consistent with public necessity.
- b. <u>Convenience</u>: The project site is located in the Reseda Central Business District, an area that is highly urbanized with commercial uses, residential uses, and transportation infrastructure. The proposed project would improve the outdoor recreational facilities of an existing school. Granting the Zone Change to the C2 Zone would allow residents local access to an LAUSD Charter School that offers high quality, on-site, outdoor recreational facilities for its students. Improving the school's facilities provides a public convenience allowing residents the option to send their children to an amenity rich school nearby.
- c. <u>General Welfare</u>: Granting the Zone Change to the C2 Zone would allow the redevelopment of an underutilized surface parking lot with improved outdoor recreational areas in conjunction with an existing school. Pursuant to three previously granted Zone Variances, the school already programs a portion of the P-zoned surface parking lot as a recreation area and lunch area for students. The Zone Change to the C2 Zone would allow the school to continue the existing use while also upgrading the outdoor school facilities with a solar shade structure, a natural turf grass field, landscaping, and trees. Students at the school would benefit from the improved facilities, additional greenery, and shade.
- d. <u>Good Zoning Practice</u>: The proposed C2 Zone is identified as a corresponding zone of the Community Commercial land use designation. The development would be consistent with the land use designation. The proposed Zone Change from [Q]P-1L-CDO to (T)[Q]C2-1L-CDO will allow for the development of an underutilized surface parking lot with recreational facilities for an LAUSD Charter School, which has been conducting student lunch, student recreation and drop-off/pick-up actives in the surface parking lot through Zone Variances since 2006 (ZA-2005-3787-ZV, ZA-2008-0748-ZV, ZA-2014-0995-ZV). As such, the C2-Zone would allow site's facilities to be upgraded while continuing a long standing authorized use of the site without the need for future Zone Variances.
- e. <u>"T" and "Q" Classification Findings</u>: Per Section 12.32-G,1 and 2 of the Municipal Code, the current action, as recommended, has been made contingent upon compliance with new "T" and "Q" conditions of approval. Such limitations are necessary to ensure the identified dedications, improvements, and actions are undertaken to meet the public's needs, convenience, and general welfare served by the required actions. The conditions that limit the scale, design and scope of future development on the site are also necessary to protect the best interests of and to assure a development more compatible with surrounding properties and the overall pattern of the existing development in the community, to secure an appropriate development in harmony with the General Plan as discussed in Findings Section 1. The existing Permeant Qualified Q condition is not

proposed for change as it has been in effect since 2005 and is related to the CDO. The Q prohibits certain uses, and regulates building height, signage, and parking.

#### Design Overlay Plan Approval Findings pursuant to LAMC Section 13.08

### 4. The project substantially complies with the adopted Community Design Overlay Guidelines and Standards.

The proposed project would replace a portion of an existing surface parking lot with improved outdoor school facilities including an open recreation area with a landscaped grass field, a modular restroom and storage facility, and an outdoor covered student activity area with a solar shade structure 25 feet in height. A portion of the parking lot will be re-striped and maintained as a surface parking lot with the addition of landscaping and trees. The project also includes installation of an eight-foot tall concrete masonry unit block wall along the southern property line and a six-foot tall security fence along the east and west property lines.

The proposed Project is located in a Community Commercial General Plan Land Use Designation within in the Reseda-West Van Nuys Community Plan and is further located within the Central Business District of Reseda Community Design Overlay District (CDO). The project, as proposed and conditioned herein, demonstrates compliance with the guidelines and standards (Section 5) of the adopted CDO as described below.

In compliance with Section 5.A of the CDO, proposed fences and freestanding walls are conditioned to comply with Guideline 6, Standard 6a, and Standard 6b of the CDO. Consistent with the CDO, a combination wrought iron and masonry wall is proposed. Masonry wall materials are conditioned to comply with the CDO's standard materials (brick, stone, split face concrete, or other decorate materials). A condition to prohibit the use of chain link fencing is also included.

In compliance with Section 5.B of the CDO, proposed landscaping includes softscape and hardscape, introduces a variety of species of trees and shrubs on a site where there were previously none, and includes a required minimum 15 foot wide landscaped buffer along the southern property line abutting single family zoned properties.

In compliance with Section 5.C of the CDO, building colors are conditioned to coordinate with the existing school building's blue gray color scheme. Highly reflective colors are prohibited in the conditions. The conditions require shielded lighting to be provided on-site. Consistent with the security devices guidelines and standards, a condition has been added prohibiting barbed wire, razor wire, and concertina wire.

Section 5.D of the CDO regarding signage does not apply to the proposed project as no signage is proposed for the site.

Therefore, the project complies will all subsections (A through D) of the Section 5, *Design Guidelines and Standards* of the CDO.

# 5. The structures, site plan and landscaping are harmonious in scale and design with existing development and any cultural, scenic or environmental resources adjacent to the site and in the vicinity.

The property to the north of the alley is developed with the Magnolia Science Academy 1 classroom and administrative buildings. The eastern and western adjoining properties are developed with surface parking. The southern adjoining properties are zoned R1 and are

developed with four single-family dwellings. The site is not adjacent to, or in the vicinity of, any recognized and/or known cultural, scenic, or environmental resources.

The project's proposed structures, site plan and landscape plan are harmonious in scale and design with existing development and will provide a visual improvement to the surrounding community. The site's asphalt surface parking lot will be replaced with an open recreation area with a landscaped grass field, a modular restroom and storage facility, and an outdoor covered student activity area with a solar shade structure. A portion of the parking lot will be re-striped and maintained as a surface parking lot with the addition of landscaping and trees. A 15-foot wide landscaped buffer with shrubs and trees will be located on the southern property line adjacent to the single family homes, providing screening, beatification, and privacy for the neighboring properties. The solar shade structure is proposed 25 feet in height and is compliant with the height requirements for structures in commercial zones that abut R1 single family zones (LAMC Section 12.21.1-A.10). Currently there is no landscaping in the parking lot and thus the proposed landscaped recreational areas and landscaped parking lot will add greenery to an otherwise asphalt parking lot. Thus, the structures, site plan and landscaping are harmonious in scale and design with existing development.

#### Environmental Findings

- 6. **Environmental Findings.** Pursuant to CEQA Guidelines Section 15074(b), after consideration of the whole of the administrative record, including the Mitigated Negative Declaration, No. ENV-2019-1065-MND, as circulated on August 22, 2019, ("Mitigated Negative Declaration"), and all comments received, with the imposition of mitigation measures, there is no substantial evidence that the project will have a significant effect on the environment. The Mitigated Negative Declaration reflects the independent judgment and analysis of the City. The mitigation measures have been made enforceable conditions on the project and a Mitigation Monitoring Program prepared for the Mitigated Negative Declaration
- 7. **Flood Insurance.** The National Flood Insurance Program rate maps, which are a part of the Flood Hazard Management Specific Plan adopted by the City Council by Ordinance No. 172,081, have been reviewed and it has been determined that this project is located in Zone C, areas of minimal flooding.



## MAGNOLIA SCIENCE ACADEMY 1 MASTERPLANNING

## **Received by Planning** September 12, 2019

EXHIBIT A APCSV-2019-1064-ZC-CDO Page 1 of 7

1. P. 1.

Powered by BoardOnTrack

SEPTEMBER 10, 2019

gkkworks CANNONDESIGN







#### **CURRENT & PROPOSED ZONING**

MAGNOLIA SCIENCE ACADEMY 1 - MASTERPLANNING

**EXHIBIT A** APCSV-2019-1064-ZC-CDO Page 2 of 7

Powered by BoardOnTrack

Magnolia Public Schools - Regular Board Meeting - Agenda - Thursday August 12, 2021 at 6:00 PM

SEPTEMBER 10, 2019 JKKWORKS CANNONDESIGN





#### CAMPUS ACCESS PATHWAYS

**MAGNOLIA SCIENCE ACADEMY 1 - MASTERPLANNING** 

### EXHIBIT A APCSV-2019-1064-ZC-CDO Page 3 of 7

SEPTEMBER 10, 2019 JKKWORKS CANNONDESIGN

Powered by BoardOnTrack



Magnolia Public Schools - Regular Board Meeting - Agenda - Thursday August 12, 2021 at 6:00 PM



**MAGNOLIA SCIENCE ACADEMY 1 - MASTERPLANNING** 

APCSV-2019-1064-ZC-CDO Page 5 of 7

Powered by BoardOnTrack



SOLAR SHADE STRUCTURE





MODULAR RESTROOM





NEW LIGHT POLES AT WEST PRPOERTY TO MATCH LIGHT POLES AT EAST PROPERTY (PICTURED ABOVE)



PROPOSED LIGHT FIXTURE AT SHADE STRUCTURE METEOR WHIZ SERIES OUTDOOR HIGH CEILING FIXTURE



NEW WROUGHT IRON FENCE AT SOUTH PROPERTIES TO MATCH NEW FENCE AT NEW MSA-1 HIGH SCHOOL (PICTURED ABOVE)









#### **MATERIALS AND FINISHES**

**MAGNOLIA SCIENCE ACADEMY 1 - MASTERPLANNING** 

1



WALKWAY PAVING ALTERNATING PATTERN: COLORED SANDBLASTED FINISH AND NATURAL CONCRETE IN BROOM FINISH



#### EXTERIOR PAINT FOR TEMPORARY RESTROOM STRUCTURE TO MATCH EXISTING PAINT COLOR AT MSA-1 HIGH SCHOOL BUILDING



DE5787 PACIFIC PALISADE

#### EXTERIOR PAINT FOR SHADE STRUCTURE TO MATCH EXISTING PAINT COLOR AT MSA-1 HIGH SCHOOL BUILDING



#### DE6348 DRAW YOUR SWORD





Powered by BoardOnTrack



DE6372 LACE VEIL



CUSTOM BUILT SHADE STRUCTURE





5



### Exhibit D

### Soils Report

### Follow Link:

https://www.dropbox.com/sh/r50ihghnhjzrwve/AAAiRKUhML-PFhnvSqGiiQeha?dl=0

### Exhibit E

# Form of Agreement -- AIA A141-2014 Standard Form of Agreement between Owner and Design-Builder



### Exhibit D

### Mobile Modular Restroom Design





### Exhibit E

Mobile Modular Proposal



5/26/2021

Dear Priti,

Thank you for the opportunity to provide what we believe is the optimum solution for the Magnolia Science Academy in Reseda.

Mobile Modular has the proven experience to deliver quality buildings on-time and on- budget. With decades of combined experience, our team has the resources and knowledge to meet the needs of discriminating clients like you who understand the value of high quality permanent modular construction. With our extensive network of design, engineering, manufacturing, and contracting partners, we work with you to determine the best solution that will address your functional, schedule, and budget requirements creating the perfect fit for your unique situation. Our strategy is to be more than a building solution; rather we want to be a trusted partner and an integral member of your project team.

Mobile Modular's dedicated Custom Modular Solutions Team, led by our Director of Construction, is comprised of Business Development Managers, Construction Managers, Project Managers and Site Superintendents.

Mobile Modular has developed a comprehensive proposal to specifically address your needs and requirements. If you have any questions and/or need any additional information please do not hesitate to call me.

Respectfully,

Carrie Gerard Business Development Manager <u>carrie.gerard@mobilemodular.com</u> 415.336.6687 cell



Modular Construction Proposal For:

### Magnolia Science Center



### May 26, 2021 Revised

Prepared For:

**Franco Architects Inc.** Priti Sawhney Arneja - Project Manager Ph: 310.621.1940 | E: priti@francoarchitects.com

### Submitted By:

Custom Modular Solutions Division Mobile Modular Carrie Gerard - Business Development Manager Ph: 415.336.6687 | E: Carrie.Gerard@mobilemodular.com



### **Table of Contents**

- Who we A.R.E.
- Environmental, Health, & Safety Statement (EHS)
- Your Project Team
- Floor Plan(s)
- Building Specifications
- Project Scope of Work
- Clarifications
- Financial Overview



#### Mobile Modular: Who We A.R.E.

Since 1979, Mobile Modular has been providing modular buildings to meet the space needs of a variety of markets, including education, commercial, government, healthcare, construction, development, industrial, manufacturing, petrochemical and retail. With a strong foundation and rich history, we are growing and exploring new frontiers and constantly pushing our limits. Headquartered in Livermore, CA, Mobile Modular Management Corporation is a division of McGrath RentCorp, which is a public company, traded on the NASDAQ Exchange (Symbol: MGRC).

Our commitment to serving our customers is summarized by three key elements of who we A.R.E. – Availability, Response and Execution. We pride ourselves on being available to help you through the design and project planning phase, responding quickly whenever you need us and creating, exceptional customer experiences by doing what we say we will do. Let our experts guide you through the entire process, ensuring that your project runs smoothly and efficiently. We continually strive to make your modular solution experience streamlined and easy. We have more than 500 years of modular experience within our leadership, sales and construction teams. We're ready to support your project.

From the start of your project, we will consult with you to develop a design for your building, a project scope, a schedule and a budget in a comprehensive and accurate proposal - to the end of your project where we will proudly hand you the keys – your project is our commitment.

Our commitment to our team members and our customers is best captured in our core values, which you can read more about on our website at <u>www.mgrc.com/#values</u>. We believe one of our most important values is "Exceptional customer experiences are our lifeblood". Our team members are committed to putting their personal best into your project, following through on their commitments and making sure you know you matter. We know relationships are built with open, honest, and direct communication and we value getting it right. We will be transparent and accountable every step of the way.

I hope that when you are ready to choose a modular building company for your project, you choose Mobile Modular. If you'd like to talk with me more about who we are or about your project please reach out to me at <a href="mailto:phil.hawkins@mobilemodular.com">phil.hawkins@mobilemodular.com</a>. Thank you for the opportunity to serve you.

*Phil Hawkins* VP, Mobile Modular



#### Environmental, Health, and Safety Statement

Mobile Modular, a division of McGrath RentCorp (MGRC), is fully committed to the principle of "safety first" and providing employees with a safe working environment. "Safety first" means that no job is more important, and no order is more urgent, that it takes precedence over the safety of our employees and workplaces.

Our leadership team is responsible for emphasizing safe working practices and behaviors, and helping to ensure a safe operation in their respective work areas, at all times. Our employees must always act in a safe manner, observe all rules of safety and health, and utilize proper safety equipment and technique in carrying out their work assignments.

All of us at McGrath RentCorp understand that nothing is more important than the safety of our employees, contractors, customers, neighbors, and the public. Incidents can result in serious personal injury and property damage, all of which can cause needless human suffering and financial loss. Therefore, MGRC implements programs that encourage employees to engage in dialogue with peers and management – about safe and at-risk behaviors.

McGrath RentCorp builds a proactive safety culture by:

- Driving continual advances and improvement of our Environmental Health and Safety Program through investigating and seeking opportunities to learn from incidents, analyzing leading and lagging indicators to identify areas of opportunity, and communication with employees, contractors and other stakeholders;
- Educating, training and equipping employees to ensure that they are empowered to avoid unsafe situations and to respond appropriately to unexpected events;
- Being a leader in the community to influence safety among our employees, contractors, suppliers, and industries.

We take responsibility for our actions and results, and we hold each other accountable to safely provide the most sustainable and efficient product possible.

Phil Couto Director, Environmental, Health, and Safety



YOUR PROJECT TEAM		
Phil Hawkins	VP Mobile Modular	Philip B. Hawkins was appointed Vice President and Division Manager of Mobile Modular in November 2011. In June of 2019, he assumed further responsibility overseeing Enviroplex, Inc. He previously served as Vice President and Division Manager of TRS-RenTelco from June 2007 to November 2011 and also held the role of Manager, Corporate Financial Planning and Analysis from June 2004 to June 2007. Mr. Hawkins was a Senior Business Analyst for Technology Rentals and Services (TRS), an electronics equipment rental division of CIT Technologies Corporation from December 2003 until TRS was acquired by the Company in June 2004. He previously served as Director of Portfolio Management and held other leadership roles with Dell Financial Services from April 1999 to December 2003. Mr. Hawkins received B.S. degrees in Accounting, Finance and Computer Information Systems from Arizona State University.
Carrie Gerard	Business Development Manager	Carrie has over twenty years of sales and account management experience. Her background includes the modular building industry as well as educational and Active Directory technology. She is focused on providing excellent customer service and being a trusted advisor to our customers and partners.
Joe Duarte	Director of Construction	Joe has worked in the industry since 1995 and is currently responsible for leading the Custom Modular Solutions Team across the US. Prior to joining Mobile Modular team, Joe held leadership positions in the Modular Construction Industry, responsible for leading turnkey high profile projects across the US and Caribbean. Joe has a Bachelor of Science degree in Civil Engineering from Temple University.
Richard Carrillo	Construction Manager	Richard has worked in the construction industry since 1987. During his many years in the field he has worked with commercial, residential, and manufactured homes. He has also been part of the installation of the infrastructure components to all the previously mentioned construction markets. Prior to joining our team he operated his own General Contracting firm. Richard obtained his "B" General Contracting License from California's CSLB.

FLOOR PLAN(S)


# Specifications

Size:	24x40 B- R/R		
COMPONENT	SUB-COMPONENT	DESCRIPTION	QUANTITY
FRAME			
	OUTRIGGER	PERIMETER CHASSIS 12X40	2
	AXLES	Quad Axle (4 Brake Axles)	2
FLOOR			
	DECKING	3/4" T&G Plywood	
	DECKING	Mechanical Fastener	
	INSULATION	R-19 Unfaced	
	COVERING	Sheet Vinyl .080 Commercial	
	Throughout (not Plumbing Chase)	Detters Deered Meltile Flow	
EXTERIOR WALL	MISC	Bottom Board Mobile-Flex	
EXTERIOR WALL	STUDS	2 X 6 Ext. Wall Framing	
	SHEATHING	1/2" CDX	
	SIDING	5/16"x 12" Hardie lap Siding	
	w/house wrap.(10-3/4" EXPOSURE).		
		Fiber Cement Vertical Siding - 5/16" -	
	SIDING	Stucco or 8" O/C -	
	W/Housewrap		
	Bottom 6' of Building around perimeter.		
	COVERING	1/2" Gypsum - Unfinished 9' To 10' Tall	
	COVERING	(Exterior Wall)	
	COVERING	Tape, Texture & Paint - 9' To 10' Tall	
	INSULATION	R-19 Kraft (Exterior Wall)	
	MISC	UPGRADE 5/4 -4" SMART SIDE TRIM	
	MISC	UPGRADE TO 5/4 -8" SMART SIDE TRIM	
INTERIOR WALL			
	STUDS	Upgrade to Plumbing Wall	
	STUDS	2 X 4 Wall Framing - 8' Tall 6X6 Ceramic Tiles in Restrooms 6' AFF	
	COVERING	6x6 Ceramic Thes in Restrooms 6 AFF	
INTERIOR WALL			
	COVERING	4' FRP Wainscot	
	Behind Janitor sink.		
	COVERING	1/2" Gypsum - Unfinished 10' Tall	
	Plumbing Chase Raw Stud.(NO WALL		
	COVERING)		
	COVERING	Tape, Texture & Paint - 9' To 10' Tall	
	MISC	Wall Extended To Rafters	
COVE BASE		6" Self-Cove	
ROOF	COVE BASE	6 Sell-Cove	
NUUF	JOIST	2 X 10	
	Double Framing at Over Hangs	2 / 10	
	JOIST	Joist Hanger	
	End Walls.		
	MATE BEAM	Modline Strap	
	MATE BEAM	20# Steel Clearspan - 12' Wide	
	MATE BEAM	Steel Clearspan Truss Ends Support	4

	INSULATION	R-38C Unfaced (W/Support Netting)	
	DECKING	1/4" Dens-Deck	
	DECKING	5/8" Wood Decking (Roof)	
	COVERING	EPDM .045 - White	
	CEILING	2 X 4 - T-Grid - Armstrong #769A	
	Nominal 9'.		
	CEILING	2X4 - Vinyl Face Tile - Upgrade in R/R's	
	Throughout.		
	MISC	Overhang - 5' Rear of Building	
ROOF			
	MISC	Overhang - 2'-6" Front on Building	
WINDOW			
	MISC	60x24 Vinyl Frame- Obscured glass -Low	2
	MISC	E	2
EXTERIOR DOOR			
	D000	3070 Tell 18ga Steel Door w/ 16 ga steel	0
	DOOR	frame.	8
	(Spartan) ALL EXTERIOR DOORS TO SWING		
	OUT.		
	LOCKSET	Schlage ND95	8
	Vandlgard Classroom Security Lock.		•
	HARDWARE	Door Closer	8
	7500 Door Closer.		0
	7500 0001 Close1.		
INTERIOR DOOR	D000	2070 16 as sheel Dear w/ 16 as France	1
	DOOR	2070 16 ga steel Door w/ 16 ga Frame	1
	LOCKSET	Passage - Tell Grade 2 - Lever	1
	LOCKSET	Deadbolt - Grade 2 - Single Cylinder	1
ELECTRICAL			
	PANEL	Nema 3R, External, 200 Amp Single Phase	1
			-
	RACEWAY	Flex W/ Ground	
	LIGHT	Dome Light 11" LED	2
	Single Use R/R's.		
	LIGHT	Troffer 2 X 4 - LED - 4000 To 6000 Lumen	8
		Troffer 2 X 4 - LED - 4000 To 6000 Luffien	õ
	LIGHT	Troffer 2X2 - LED - 3400 Lumen	1
	Plumbing Chase.		
	EXTERIOR LIGHTS	Exterior 12w LED With Photo Control	6
	SWITCH	CEILING OCCUPANCY SENSOR	2
	LMPC-100		_
	SWITCH	WALL OCCUPANCY SENSORS	8
	WATTSTOPPER PW-301-W OR EQUAL.	WALL OCCUTANCE SENSORS	0
	WATISTOTTERT W-301-W ON EQUAL.		
ELECTRICAL			
	SWITCH	ROOM CONTROLLER	8
			0
	1-PER ROOM.(LMRC-211)	204 M/D CECI Directorial M// M/hile to Lies	
	RECEPTACLE	20A - WR GFCI Protected W/ While In Use	1
		Cover	

(mounted on roof at hvac location) Installed by others on site.

	RECEPTACLE RECEPTACLE MISC	20A-Duplex-Tamper Resistant 20A-GFCI-Duplex-Tamper Resistant CAT 6 CABLE(FOR DLM WATTSTOPPER SYSTEM)	6 5
PLUMBING	PIPE (TYPE-L)	Copper Plumbing Drop	
	WATER CLOSET	Water Closet (Wall Hung w/Carrier) Manual Flush Valve.	7
	URINALS LAVATORIES MOP SINK Janitors Closet.	Urinal - Low Flow Lavatory - Wall Hung Mop Sink - Floor Mount	1 6 1
	FLOOR DRAIN WATER HEATERS Janitors Closet Exposed.	Floor Drain W/ Drip Primer 40-Gallon - Electric Water Heater	3 1
	ACCESSORIES	Toilet Paper Dispenser	3
	ACCESSORIES	Partition - Urinal Screen	1
	ACCESSORIES	CLASSIC SERIES RECESSED MULTI-ROLL TOILET	4
	B-3888 OR EQUAL. (ADA STALL'S/R/R'S.)		
	ACCESSORIES	Modesty Partition	5
	ACCESSORIES	Grab-Bar Set	4
PLUMBING			
	ACCESSORIES	CLASSIC TOILET SEAT COVER DISPENSER SURF	7
	B-221 OR EQUAL. ACCESSORIES B-165 or equal. MISC	18"X30" STAINLESS STEEL MIIRROR Water Hammer Arrestor - 3/4"	6
			-
	MISC	Hose Bibb - Frost Proof (WALL HYDRANT)	1
	ZORO -G1548504, Or equal. MISC MISC	In-Line Strainer Pipe Insulation	1

HVAC

MISC **Restroom Sign** DUCT Flex - Supply Air Duct Galvanized 26 ga wrapped with R-6. EXHAUST FANS 180 CFM - Ceiling Mount EXHAUST FANS 70 CFM - Ceiling Mount 1- per Restroom, (1) per Locker Rm. 4-Ton Daikin Model DBH 14 Seer Rooftop MISC Heat Pump 100% Fresh Air.(Include Merv 8 filter). MISC Barometric Relief Damper MISC MISC Close-Up MISC Shipping Wall

#### STATES STATE APP

4

2

6

1

8



### SCOPE OF WORK

SCOPE OF WORK				
Description	Details	Excluded	Mobile Modular	Client
BUILDING-24'x40' Custom				
New Modular Building-24'x40'	The new manufactured modular building(s) to be provided will consist of approximately 960 square feet and is to be designed/constructed in accordance with the Building Specifications.		x	
TRANSPORTATION				
Deliver New Modular Units	Transport (2) modular units measuring a nominal 12' Wide x 40' Long x 13'-6" High from the Manufacturing Facility located in Phoenix, AZ to the project site located at Reseda, CA. Includes all surveys, permits, fees and escorts as required for delivery to site staging area. If excess detours or mileage are required for delivery to site, additional cost could apply.		x	
BUILDING SETUP				
Building Setup (Crane)	Provide required labor, materials and equipment to crane set the modular units onto the building foundation. Crane placement is assumed to be within 30' of proposed building(s). Setup Contractor is to provide all necessary rigging straps/cables to ensure proper lift of units.		x	
Remove Axles/Tire	Axles and tires to be removed and stockpiled on site. Removal from project site is by Client		x	
Remove Hitches	Hitches to be removed and stockpiled on site. Removal from project site is by Client.		x	
DIVISION 1 - GENERAL REQUIREMENTS				
Architecture/Engineering & Testing				
Modular Building Drawings	Modular shop drawings, State 3rd party approved in accordance with applicable state and national model codes and project requirements Architecture of Record, Including but not limited to modular building and other		x	
Architectural Fees	building drawings. Modular building drawings to be furnished by building manufacturer.	х		
Civil Engineering Fees	Site plan including but not limited to existing/proposed grading plan, site development, landscaping, etc. in accordance with specific codes and project requirements	x		
Structural Engineering Fees - Foundations	Foundation structural drawings in accordance with applicable project and code requirements. This is for standard soil conditions. If non standard soil conditions are present fees will be adjust to account for these non standard conditions		x	
Structural Engineering Fees	Complete structural drawings with exception of building foundations. All drawings shall be in accordance with applicable codes and project requirements	x		
Geotechnical Fees	Complete geotechnical testing at site and develop a comprehensive report with recommendations in accordance with applicable codes and project requirements	х		
MEP Engineering Fees	Complete MEP drawings with exception to the building envelope plans and schematics provided by the modular building manufacturer. All drawings shall be in accordance with applicable codes and project requirements	x		
Surveying	Complete a survey with a topo map of the required project area including survey limits, range, contours, all surface features and underground utilities within the area to be surveyed shall be shown and identified on the maps. In addition, these features shall be located by sufficient distance ties and labeled on the sheets to permit accurate scaling and identification. Establish project benchmarks and corner locations of the building(s).	x		
SWPPP Planning/Design	Provide a complete SWPPP plan and details for work necessary prior, during and after proposed construction. All plans and details shall be in accordance with applicable codes	x		
Soil Testing	Provide soil testing services prior to and during construction as required	х		
Concrete Testing	Provide labor and materials to complete slump and compressive testing in accordance with code and project requirements All cubenterchard and accordance to the construction	х		
As-Built Drawings	All subcontractors and manufacturer shall record any changes to the construction drawings and submit to Mobile Modular as a record for As-Build Drawings. Mobile Modular will incorporate those changes into a complete As-Build set and submit to the client.		x	
Permits, Fees, Bonds, Labor				
NONE	Not Proposed or Included in this Scope of Work	х		
Prevailing Wage	Non Union / Prevailing Wage to Include Certified Payrolls Submitted Weekly.		X	
Equipment Crane Rental & Operator	Crane Contractor shall provide a crane of sufficient type and size based on the project site conditions, modular unit dimensions of 12' Wide x 40 ' Long x 13'-6" High, heaviest modular unit weight of 28,800 pounds, and a maximum working radius of 40' from hook to pin center. Crane shall contain up-to-date inspection reports and be in complete compliance with all OSHA requirements.		x	
Crane Mobilization & De-mobilization	Crane Contractor shall be responsible for all labor, equipment, permits, escorts, etc. as required to erect and break-down the crane at the project site. A laydown area shall be provide by the owner/client of sufficient size as directed by the crane contractor.		x	



### **SCOPE OF WORK**

SCOPE OF WORK				
Description	Details	Excluded	Mobile Modular	Client
	Crane Contractor shall provide all rigging to include, but not limited to; spreader			
	bars, wedge sockets, slings, wire rope, shackles, chain, etc. All equipment shall be			
Crane Rigging & Equipment	inspected prior to use and shall be in compliance with all OSHA requirements.		x	
	NOTE: When performing a "basket lift" (under the modular unit's steel frame)			
	Wire Rope of sufficient size and capacity shall be used. The use of slings for a			
	basket lift is NOT be permitted!!! Crane Contractor shall provide a Certified Rigger(s) to assemble and connect all			
Rigger(s)	rigging for each lift.		x	
Project Safety				
	Including but not limited to hardhats, eye protection, hi-vis shirt/vest/jacket,			
Safety PPE	work boots, long pants, shirts with sleeves (i.e. no tank tops or muscle shirts). All		x	
	workers shall be properly equipped by their company per OSHA			
First Aid Kits	Appropriate size first aid kits located in the construction trailer or contractor		x	
	vehicle(s) and at each building exit in the proposed building to be constructed		^	
Project Management				
	Mobile Modular Project Manager has overall responsibility of the project budget			
Project Manager	and schedule. The Project Manager will be the POC for the client with any		x	
	specific clarifications, change orders, schedule adjustments, etc			
	Mobile Modular's Project Superintendent will be on site daily when any work is being performed by Mobile Modular or their contractors. The Project			
Project Superintendent	being performed by Mobile Modular or their contractors. The Project		<b>v</b>	
Project Superintendent	Superintendent is responsible for managing day to day Mobile Modular project		x	
	activities, coordination and communication with Mobile Modular Project			
	Manager and Other Project Superintendents Working hours for this project are established as 7:00 AM to 4:00 PM Monday			
Norking Hours	through Friday. If subcontractors desire to work outside of these hours, prior		x	
	approval must be granted by Mobile Modular and approved by the Client		^	
ubmittals & Plans				
	Mobile Modular's Project Manager will turnover to the client all O&M manuals in			
Operation & Maintenance Submittals/Manuals	electronic format (i.e. PDF) on a flash drive. Each contractor shall provide a		x	
	complete electronic (PDF) O&M manual for their scope of work.			
	Mobile Modular's Project Manager will turnover to the client all necessary			
	closeout documents including but not limited to as-build drawings, engineering			
Closeout Submittals	letters, certifications, etc as required by the project documents and/or code.		x	
	Each contractor shall provide a complete electronic and hard copy of their			
	closeout for their scope of work.			
	Mobile Modular will have on site available a safety plan. All Mobile Modular			
Safety Plan	contractors shall be responsible for understanding the requirements outlined in		x	
	the safety manual in accordance by OSHA, EM385 and/or Mobile Modular Safety			
	Manual as required.			
Narranty	Mobile Modular's Project Manager will turnover to the client all necessary		x	
	warranty documents as required by the project documents.			
DIVISION 3 - CONCRETE				
Concrete Foundation-Initial Segment	24x40 at grade foundation w/ 18" crawl space and 12" footing embedment(initial foundation compared		x	
Concrete Foundation-Additional Segment	foundation segment 24x40 at grade foundation (additional segments)		x	
Concrete Foundation-Additional Segment	Supply and install (8) steel weld plates into concrete stem wall foundation		x	
Foundation Vents	6' Polyvent		x	
oundation Flashing	12" high, 24 ga metal flashing		x	
DIVISION 5 - METALS			~	
	Prefabricated Aluminum Canopies Shall be Designed by a Professional			
	Engineer/Architect. Provide labor, materials and equipment to design, ship,			
Aluminum Canopy	unload, layout on site, install and inspect canopies as required. All welding on or		x	
	off site shall be completed by a certified welder.			
Aluminum Canopy-Site Install	Labor and materials to install (2) aluminum canapoies on site		x	
DIVISION 15 - MECHANICAL				
Plumbing				
	Labor and materials to install sanitary waste manifold from building and connect			
Plumbing-Site Close Out	to site connection. To include the installation of domestic water line to building		x	
	connection from site connection.			
leating, Ventilation and Air Conditioning				
Aisc HVAC	Labor and materials to install primary condensate line for roof mounted HVAC		x	
DIVISION 16 - ELECTRICAL				
lectrical Service				
	Complete electrical service in the proposed building to a Main Distribution Panel			
Electrical Service - Proposed Building to MDP	(MDP) or Disconnect installed in or on the proposed building. Excludes	х		
	excavation, conduit, wire, transformers, disconnects, switchgear, inspections and			
	tie-ins, beyond the proposed MDP or Disconnect.			
Electrical Building	Mabile Medular electrical contractor shall foundate all the second sector (2)			
	Mobile Modular electrical contractor shall furnish all labor and materials to wire			
Wire Roof Mount HVAC Units (Power)	the HVAC units from the designated electrical panel to the RTU. Building	х		
	Manufacturer to furnish HVAC disconnect for installation and testing by Mobile			
	Modular electrical contractor.			

SCOPE OF WORK					
Description	Details	Excluded	Mobile Modular	Client	
	Furnish all labor and materials to wire and install HVAC Thermostats. Building manufacturer to furnish HVAC thermostats for installation and testing by Mobile Modular electrical contractor.	x			



#### **Contractual - Sale Transaction**

Customer is hereby advised of the following: Due to the extreme volatility in the materials market Mobile Modular retains the right to apply an equitable adjustment in the amount of any or all of the charges set forth in this Sale Agreement to account for any increases in material costs which may occur between the "Date Printed" shown on this Sale Agreement to the date that the Scope of Work is performed by Mobile Modular, to the extent that such increases are not the result of any fault or negligence on the part of Mobile Modular.

Mobile Modular's Proposal, in its entirety, shall be included and made a part of any contract resulting from this Proposal. In the event of conflict, the contents of the Proposal shall be considered binding upon the parties and supersede any and all other documents.

In the event that Mobile Modular's failure to perform according to its Schedule is shown to be the sole and direct cause of Contractor not meeting its contract completion date absent (i) any Contractor failure to properly schedule and complete its own work (ii) Contractor interrupting, delaying or causing rework by Mobile Modular, (iii) Contractor or Owner not approving or responding to any submittals or requests by Mobile Modular necessary to ensure the seamless and timely performance of its Work, (iv) any negligence, delay or omission of the Contractor or Owner, or (v) any Force Majeure event ((i) through (v), each a "Concurrent Delay"), then Mobile Modular will bear responsibility for the payment of liquidated damages imposed on Contractor by Owner. In the event that Mobile Modular's failure to complete its contracted scope of work according to the Schedule contained in its Proposal is shown to be caused or contributed in whole or in part by a Concurrent Delay, , then Contractor will bear responsibility for the payment of liquidated damages responsibility for the payment of liquidated to the Schedule contained in its Proposal is shown to be caused or contractor by Owner.

The prices quoted herein exclude any and all taxes, fees, etc. Such taxes or fees shall be delineated as a separate line on the Mobile Modular invoice at time of billing, and shall be the responsibility of the Customer.

The cost of a performance bond, if required, is \$15.00 per \$1,000.00 of the total contract amount, shall be the responsibility of the Customer, and is not included herein.

The transaction is subject to Mobile Modular Corporate Credit and Risk Management approval.

Mobile Modular will not be responsible for nor accept any and all claims and demands for loss of profits or other incidental, consequential and/or punitive damages arising out of or in connection with this agreement.

Mobile Modular will furnish its standard insurance certificate evidencing statutory Workers' Compensation, General Liability coverage of \$1,000,000 per occurrence, \$2,000,000 aggregate per policy, and Auto Liability with combined single limits of \$2,000,000, to be effective while Mobile Modular is working on the project site. The Customer and the property owner only shall be named on the Certificate as additional insureds with respect to liability coverage. Mobile Modular's insurance is primary with respect to its scope of work only. Waivers of subrogation and copies of policies will not be provided. Mobile Modular subcontractors will provide insurance coverage in accordance with Modular Building Industry or Mobile Modular Standards. Mobile Modular will indemnify the Customer against loss, cost, expense or liability to the extent that the cause or causes of such loss, cost, expense or liability results from the proven negligence or willful misconduct of Mobile Modular.

The Customer shall maintain insurance and agrees to be responsible for all losses, costs or damages incurred or paid by Mobile Modular on account of any claim under Worker's Compensation Acts or other employee benefit acts, any claim for damages because of bodily injury, including death, to Customer's employees and all others, and any claims for damages to property caused by, resulting from, or arising out of Customer's activities while Mobile Modular is performing work at the site.

With regard to new/custom equipment, if the Customer terminates the contract prior to the fabrication of the modular building, then the Customer shall reimburse Mobile Modular for all costs plus 15%. Once fabrication of the modular facility commences, the right of termination shall diminish and the full contract sum shall become due and payable.

Mobile Modular's delivery of the equipment described in this proposal is subject to delays in manufacture, delivery or installation due to fire, flood, windstorm, riot, civil disobedience, strike, failure to secure materials from the usual source of supply, Act of God, or any other circumstances beyond Mobile Modular's control.



For the purpose of acceptance, the Customer shall inspect the equipment within 48 hours following the substantial completion of the Mobile Modular scope of work, and shall provide Mobile Modular notice of defects in, or other proper objections to, the equipment. Such inspection and acceptance must take place prior to any beneficial occupancy of the equipment. Mobile Modular reserves the right to correct such deficiencies within a reasonable time period. The Customer shall not proceed with corrections without the prior written approval of Mobile Modular. Should the Customer fail to notify Mobile Modular of defects within the required 48 hours period, it shall be presumed that the Customer has inspected the equipment and it is in good condition and is deemed acceptable to the Customer.

Customer will provide free and clear access for delivery and installation of the equipment by standard mobile transport vehicles. Customer will be solely responsible, at its cost, for preparation of the site on which the equipment is to be used, including any required structural or grade alterations and identification of utility lines. Customer will provide firm and level ground on no more than a six-inch (6") slope from one end to the other for safe and unobstructed installation of the equipment. Site selection is the sole responsibility of the Customer.

The Customer warrants that he owns, or has the right to construct buildings on, the property upon which the equipment as described herein is to be delivered, constructed, or other work performed, and will designate to Mobile Modular the location of the corner stakes of the property and will furnish the plot plan showing the boundary dimensions and angles of the property, and the proposed location of the site of the building or other work to be performed, together with all necessary information concerning contours, grades, soil conditions, tree locations, utility service lines, rights of way, easements and restrictions, dimensions and other relevant data pertaining to existing structures on the premises. Mobile Modular shall not be responsible for encroachments of any type. The Customer warrants that the said construction will not violate zoning restrictions or other laws, and the Customer agrees to indemnify and hold Mobile Modular harmless from all loss or damage or liability which may result by reason of the construction of the said building or other work done, or from any lack or defect of title in the Customer, or by reason of said construction violating any zoning restrictions or other laws.

The Customer agrees not to interfere with the progress of the work, and not to occupy any portion of the building until all terms and conditions herein are fulfilled by both parties. Customer further agrees not to permit any workmen other than those of Mobile Modular to work at or in the immediate vicinity of the building without Mobile Modular's written consent until Mobile Modular's work on the building is completed. The Customer agrees to pay Mobile Modular for any damage that may be caused by anyone other than workmen or subcontractors of Mobile Modular, by reason of disturbing or damaging concrete forms, grade finishing or any construction work in process whatsoever.

Mobile Modular's price and schedule are based on the assumption that the site is environmentally clean and has no subsurface conditions. If any are found, the Customer will be responsible for the remediation of pre-existing environmental conditions and the removal/displacement of subsurface conditions found at the project site.

Risk of loss shall transfer to the Customer upon delivery of the Equipment to the Site. Substantial Completion shall be achieved when the work is completed in accordance with Mobile Modular's proposal and is ready for its intended use and operation, except for punch list items of work by Mobile Modular. Title to the equipment will be furnished to the Customer within thirty (30) days of Mobile Modular's receipt of payment in full from the Customer.

This Agreement is strictly by and between Mobile Modular and Customer and Mobile Modular is not responsible for the terms and conditions of any prime contract between Customer and Owner.

This transaction is subject to Mobile Modular Corporate Credit and Risk Management approval.

This Agreement shall be construed as to its construction, interpretation and effect by the laws of the State of California without regard to principles of choice of laws.

#### General Construction:

This proposal is based on Mobile Modular providing a building, which meets or exceeds the requirements for the State of California. Any additional requirements or directives by local inspectors and/or other agencies shall be the responsibility of the Customer, and the Customer shall be responsible for providing such information to Mobile Modular.

Mobile Modular's price and schedule are based on the assumption that the site is environmentally clean and has no subsurface conditions. If any are found, the Customer would be responsible for the remediation of pre-existing environmental conditions and the removal/displacement of subsurface conditions found at the project site.

Mobile Modular standard building materials and installation methods have been provided for this building unless otherwise noted.



This proposal is subject to finalized engineering and architectural details.

Mobile Modular will provide all permits necessary for the manufacture and delivery of the modules to the site.

All dimensions provided are nominal.

Mobile Modular is providing manufacturers shop drawings only. Any other drawings, certifications and/or tests required are to be supplied by others

Mobile Modular will provide all permits necessary for the manufacture and delivery of the module(s) to the site. The Customer shall obtain and be responsible for all permits pertaining to the building, installation, site work and occupancy. Customer, prior to delivery, shall have obtained all necessary approvals and permits required for the installation of the equipment. Customer will indemnify and hold Mobile Modular harmless from and against any fines, penalties and liabilities that may arise from the failure of Customer to obtain any necessary permits.

In the event that any act or omission by Customer, including the failure of Customer to complete any work or obtain any permits for which it is responsible, or Customer's failure to make the site available and ready causes a delay in substantial completion, Customer will be liable for any additional costs incurred by Mobile Modular to the extent caused by such delay.

The definitive project completion schedule shall be mutually negotiated and agreed to by the parties upon execution of the agreement.

The pricing provided herein is based on Non-Union Labor Open Shop Labor

The Customer shall be responsible for site security.

NOTE: "others" denotes not by Mobile Modular, or its agents, employees and/or subcontractors.

#### Schedule:

The Mobile Modular proposal is based on a standard 40-Hour work week. Overtime, weekend or holiday labor, if required, may result in additional costs.

**Building:** 

The Owner shall approve the manufacturer's shop drawings prior to fabrication of the modular facilities.

Customer will be responsible for normal maintenance such as changing HVAC filters, light bulbs, janitorial services, and other minor repairs.

Site:

Mobile Modular assumes the minimum required distance from any and all assumed and/or common property lines.

Mobile Modular shall not be responsible for any and all environmental and/or subsurface conditions including but not limited to rock, unsuitable soil conditions, hazardous materials, etc.

All pricing is based on normal level site conditions or standard modular set-up procedures.

Mobile Modular assumes clear access in all directions with off-street staging area. Customer shall provide parking control during staging and set-up as required.

Site work removal or relocation of obstructions (above or below ground) is not in the contract. Standard "ROCK CLAUSE" applies. Assume level site within 8" over total area of complex.

Mobile Modular for reasons of safety and schedule considerations fully expects that the portable site be clear of all other trade's and any excavations or ditches caused by those trades until such a time that the "Modular Buildings" are in place and anchored, and that access to the portable site will not be constrained by other subcontractors of trades that are not contained in our scope of work.

Customer is responsible to insure the site grading allows water to run off away from the proposed buildings. The customer is also responsible to insure the proper grading is maintained while the buildings are on site to assure that water is not present under the structure.

A staging area located adjacent to the work site.

Suitable and acceptable access to the site for the module size(s) to be provided.

No dewatering of subsurface water.

All underground obstructions, if any, within the proposed building envelope/work area to be located and marked above grade, by others.

Storm water management/erosion and sedimentation control by others.

Mobile Modular assumes that all electrical, water supply and sewer are sized to provide adequate supply to the modular building. Should any system not be sufficiently sized, Owner would be responsible for the cost to adjust any system to provide sufficient supply/volume to the new modular building.

Foundation:

Assume minimum 2,500 PSF soil compaction at grade and/or frost line.

A soils test may be required by the local building department to support the foundation design. The cost of the soil test and any design changes resulting from this report will be the responsibility of the Customer.

Foundation to be a concrete stem wall foundation at grade.

Building Installation/Setup:

Placement of module(s) to be accomplished by the use of crane



Wheels, axels, and hitches to be removed from building and stored on site for removal by customer.

Decks, Steps, Ramps, and/or Canopies:

No ramps, steps, stoops, and/or walks.

Alunimum canopies to be installed on site

Mechanical:

H.V.A.C. balancing by others.



### **PROPOSED PRICING - PURCHASE**

### **Magnolia Science Center**

•	
BUILDING-24'x40' Custom	\$ 319,850.00
TRANSPORTATION	\$ 7,470.00
BUILDING SETUP	\$ 10,980.00
Total - Building, Delivery and Install	\$ 338,300.00
DIVISION 1 - GENERAL REQUIREMENTS	\$ 31,070.00
DIVISION 3 - CONCRETE	\$ 46,470.00
DIVISION 5 - METALS	\$ 23,100.00
DIVISION 15 - MECHANICAL	\$ 7,730.00
TOTAL DIVISIONS 1 THRU 16	\$ 108,370.00
GRAND TOTAL	\$ 446,670.00
ALTERNATES	
Ceramic Tile Flooring Upgrade	\$ 13,820.00
Epoxy Flooring Upgrade	\$ 44,800.00
Total Alternate Items	\$ 58,620.00

# **Cover Sheet**

### Approval of MSA-1 Shade Structure Design Contract

Section:	IV. Action Items
Item:	C. Approval of MSA-1 Shade Structure Design Contract
Purpose:	Vote
Submitted by:	
Related Material:	Contract for Shade Structure (Board).pdf



Agenda Item #:	IV C: Action Item
Date:	August 12, 2021
To:	Magnolia Educational & Research Foundation dba Magnolia Public Schools (" <u>MPS</u> ") Board of Directors
From:	Alfredo Rubalcava, CEO & Superintendent
Staff Lead:	MPS Audit/Facilities Committee
RE:	Approval of MSA-1 Contract for Shade Structure Project

### I. Proposed Board Motion

Staff recommends and moves that the Board approve the selection of M Bar C Construction for the design and construction of the shade structure that is a part of the Magnolia Science Academy—1 ("<u>MSA-1</u>") middle school parking lot site improvement project (the "<u>Project</u>") for up to \$400,000 (with design costs not to exceed \$25,000) and the execution of the agreement attached as <u>Exhibit A</u>, with such changes as MPS Staff may deem advisable or in the best interests of MPS.

### II. Background

MSA-1 filed for and received a zone change for its parking lot parcels from the City of Los Angeles. As a condition to the zone change, MSA-1 agreed to make certain improvements to the parking lot parcel that serves the MSA-1 middle school building. See <u>Exhibit B</u>. They include the following: (1) the fixing of drainage on the middle school parking lot; (2) the creation of an open green space and a concrete pad with a solar shade structure; (3) fencing surrounding the entire MSA-1 parking lot parcels; (4) the creation of a track; and (5) the installation of modular outdoor restrooms. MSA-1 wishes to undertake these improvements because it will assist the school during the current pandemic to allow for activities to take place outside. Looking to the future, having a good recreation and outdoor space will benefit the students and may in fact attract more students to the campus.

### III. Procurement

Franco Architects worked with Staff to solicit bids for the solar shade structure. Staff prepared a Request for Proposals and published it on the MPS website. See <u>Exhibit C</u>. Franco Architects targeted specific vendors to solicit proposals. After some back and forth conversations with a couple vendors, M Bar C was the only vendor that provided a proposal for the shade structure. Concurrently, Staff priced out the cost of a custom designed and built shade structure. A custom design would require a change order to

210812 - MPS - Shade Structure .docx

August 9, 2021 Page 2



the Franco Architects contract as Franco was hired to help MPS secure a modular restroom. Note the language from the March 11, 2021 board report – "[f]or all proposals, all assume that the solar shade structure and the restroom structure will be "design build" or "modular", meaning that a vendor will be identified and they will be responsible for designing and permitting the structure. Each design firm will assist in identifying vendors to provide these services." We further sought input from a structural engineer and general contractor on the cost of building a custom designed solar shade structure. The cost of a custom designed solar shade structure is difficult to determine with the cost escalations in the construction industry; however, pricing was estimated at \$473,000. Staff believes that the M Bar C design is both attractive and functional and will allow MPS to lock in prices early. (See Exhibit D proposal). Staff expects that the shade structure will be constructed without solar panels and then when additional funding is available they will be installed.

### IV. Conclusion & Budget Impact

Staff recommends that MPS sign a contract for up to \$400,000 with M Bar C for the solar shade structure that is part of the Project. The proposal is for \$340,000 however we expect some price increases due to some recent changes to the design, specifically, increasing the span of the shade

Staff intends to seek financing for the Project. Until such financing is secured, MSA-1 may need to pay for any expenses, for example, progress payments, out of its operating budget or its reserves. Such expenditure will not have a material impact on either MSA-1's operating budget or its reserves.

Exhibit A	Prime Contract	Pg. 3
Exhibit B	Site Improvement Site Plan	Pg. 15
Exhibit C	RFP	Pg. 18
Exhibit D	M Bar C Design and Proposal	Pg. 87



# Exhibit A

### Prime Contract



THE STRENGTH BETWEEN SUN AND SHADE™

### PRIME CONTRACT BETWEEN CUSTOMER & CONTRACTOR

This Agreement ("Contract") is made this \_ day of ("Date"), between M Bar C Construction, Inc. ("Contractor"), located at 1770 La Costa Meadows Drive, San Marcos, CA 92078 and ("Customer"), located at ("address")

The work described in Section 1 below shall be performed exclusively in accordance with this Contract, all plans, specifications and other mutually agreed to contract documents for the project known as:

Project Name: Project Address:

**SECTION 1.** <u>SCOPE.</u> The Contractor agrees to furnish all labor and other necessary services required to complete the work as defined in the attached Contractor's Proposal ("Project Proposal") in its entirety, attached hereto and made a part hereof.

**SECTION 2.** <u>PRICE AND PAYMENT.</u> The Customer agrees to pay the Contractor for the strict performance of the work, the sum of: ("\$ Amount") subject to adjustments for changes in the work as may be agreed to in writing by the Customer and the Contractor, as may be required under this Contract.

The Customer agrees to pay the Contractor in monthly progress payments for all work acceptably completed. Payment shall be made on acceptable completion of the Contract where the work is completed within a period of 30 days. Payment shall be made to the Contractor by the 10th of the month following the date of any billing.

Irrespective of any other provision in this contract, it is the intent of the parties hereto that any and all payments be made to subcontractor within 90 days of last day subcontractor provides onsite labor in order to preserve lien rights in accordance with California statutory authority.

The Contractor shall not be required to proceed with the installation of the work if the payments applying on same have not been made as specified in the Contract. In the event suit is filed by the Contractor to collect any monies due hereunder or to enforce any other provisions of this Contract, to the extent the Contract prevails in the suit the Customer agrees to pay all costs, and the attorney's fees incurred to the extent authorized by applicable law or court rule.



THE STRENGTH BETWEEN SUN AND SHADE™

In the event of a dispute, except as otherwise provided herein, the Contractor shall continue to perform its obligations under this Contract.

Contractor shall provide a Schedule of Values ("SOV") outlining Contractor's expected progress payment schedule in a format that includes the Project's Phase, Percent Complete and Value of such percent complete.

Item No.	Description of Work	Scheduled Value
А	В	С
1	Notice to Proceed with Design	
2	Permits having been obtained	
3	PO issued for Materials	
4	Materials Delivered to Site	
5	Structural Mechanical Completion	
6	Electrical Mechanical Completion	
7	Substantial Completion	
8	Final Completion (Final AHJ Inspection)	
	TOTAL	

**SECTION 3.** <u>ENTIRE AGREEMENT.</u> This agreement represents the entire agreement between the Contractor and the Customer regarding the work described in Section 1 and supersedes any prior written or oral agreements or representations as to said work.

**SECTION 4.** <u>TIME.</u> Time is of the essence of this agreement. The Contractor shall provide the Customer with a progress and completion schedule, including any changes to that schedule agreed to between the Customer and the Contractor. Customer shall not unreasonably delay agreement to said schedule.

**SECTION 5.** <u>CONTRACT DOCUMENTS.</u> This Contract and any attachments referenced herein are in accordance with the requirements of the project and from verbal information which Contractor has received from the Customer (all such verbal communication must be confirmed in writing to the Customer and confirmed by Customer before it is considered authorized), its



THE STRENGTH BETWEEN SUN AND SHADE™

representative or agent, and written plans and specifications furnished by Customer or Contractor and agreed to by the parties. The Contractor assumes no responsibility as to the accuracy of such plans, specifications or other document provided by the Customer. Contractor shall not be bound by any documents which have not been furnished to Contractor prior to signing this Contract unless otherwise agreed to in writing.

It is further understood and agreed that the Contract does not include any labor or materials not specifically mentioned herein. Unless otherwise provided in the plans and specifications, the Contractor shall have the right to select all materials. When specified materials are unavailable, the Contractor shall, with Customer's prior written consent, have the right to substitute for such specified materials other materials of equal or better quality.

**SECTION 6.** <u>OWNERSHIP OF DOCUMENTS.</u> Contractor's Proposal, including any plans, specifications, drawings, or engineering data are furnished by the Contractor to the Customer in trust for determining the scope of the work to be performed and shall remain the property of the Contractor. They shall be immediately returned to the Contractor in the event the Contractor is not awarded the Contract to perform such work. Such plans, specifications, or other data cannot be used for the purpose of obtaining other bids or in connection with the installation. Contractor vests in Customer a limited irrevocable license to use the documents and work product in connection with Customer's operation and repair of the work and Contractor shall provide with a copy of the "as-built" plans related to the work.

**SECTION 7.** <u>EXISTING SITE CONDITIONS.</u> Working surface is assumed to be structurally sufficient to support all work activities such that damage from the weight of typical equipment expected for this scope of work, such as: reach forks, concrete trucks, drill rigs, etc. will not occur. Subcontractor will verify existing conditions on site prior to construction activities and will advise the Contractor of any concerns. Contractor acknowledges that normal tire marks caused by equipment during construction operations will occur and will not be required to be removed by Subcontractor. Any landscape or hardscape damaged by Subcontractor will be repaired by Subcontractor resulting from unavoidable and/or negligent action

It is a condition of this Contract that the Customer will assume any and all responsibility for any material or equipment not furnished by the Contractor, and that such will be of a type that is approved for the purpose unless otherwise specifically set forth in this Contract.

**SECTION 8.** <u>DIFFERING SITE CONDITIONS.</u> Contractor shall promptly notify the Customer in writing of any discovered conditions that Contractor may identify that were not readily observable by visual inspection of the site or that differ from conditions shown in any drawing, document or other information, if any, that Customer is required to deliver to Contractor,



THE STRENGTH BETWEEN SUN AND SHADE™

however, nothing imposes a requirement of inspection on the part of Contractor to search for any defects, non-conformities, unknown or inconsistent conditions.

If the Contractor encounters subsurface or latent physical conditions or physical conditions at the site of any unusual nature differing materially from those ordinarily encountered in the trade upon standard industry inspection, or discovers any unforeseen or unknown physical conditions, differing materially from those indicated in the Contract documents, the Contractor will promptly notify the Customer. If such conditions cause an increase in the cost of, or the time required for, performance of the work, the Contract Price and the time for completion will be equitably adjusted for costs incurred including 15% profit and 10% overhead, and time lost to the extent supported by reasonable documentation.

The Customer shall promptly, so as not to unreasonably delay the Contractor's schedule of work, investigate any and all conditions that Contractor identifies and notifies Customer of such existence.

If the Customer finds that the worksite conditions do materially differ, or involve hazardous waste or toxic pollutants, the Customer shall cause a decrease or increase in the Contractor's cost of, or the time required for, performance of the affected part of the work by issuing a change order under the procedures described in the Contract.

**SECTION 9.** <u>ALTERATIONS & CHANGES IN WORK.</u> The work shall be subject to changes or additions, deletions or revisions by the Customer. The Contractor will be notified by receipt of written additions and/or revised drawings, specifications, exhibits or written orders.

Alterations or additional work ordered by Customer or their agent shall constitute an addition to this Contract. The Contractor shall not be obligated to perform changes in the work or additional work until the Customer has approved, in writing, the changes to the Contract price and the Contract time. All additions above and beyond the Contract Price and outside the general Scope of Work herein shall be adjusted including 15% profit and 10% overhead by the Contractor.

If any additions outside the scope to the Contract are ordered and/or necessary, the Contractor shall provide the Customer with a Change Order and/or billing the amount to be paid for the same including overhead and profit. If a change requested by the Customer shall reduce the amount of labor or material, or both, that the Contractor shall furnish to the project, the Contractor shall provide a deductive Change Order and/or billing to the Customer. The Contractor shall receive written orders for all additional work or changes, signed by the Customer representative or their authorized agent before proceeding with such additions or changes.



THE STRENGTH BETWEEN SUN AND SHADE™

Payments for additions or changes shall be made under the same terms and conditions provided under this Contract. Approvals by Customer shall not be unreasonably withheld or delayed.

Whenever an adjustment in the Contract Price or Contract time is required due to a material change in site conditions, errors in the plans and specifications, or other circumstances beyond the control of Contractor (including lack of worksite access, weather, fires, floods, strikes, acts of God, natural disasters, or acts of third parties), the Contractor shall submit to the Customer within a reasonable time a detailed estimate, with supporting calculations, pricing and adjustments in the schedule of the change to the Contract Price and the Contract time. Pricing of the adjustment shall be in accordance with the pricing structure of this Contract. However, to the extent that such pricing is inapplicable, cost of the change or the amount of the adjustment shall be determined on the basis of the cost to the Contractor for labor and material plus Contractor's standard 15% profit and 10%

**SECTION 10.** <u>DELAYS AND TIME EXTENSION.</u> Contractor will not be held liable for damages or delay which arises out of causes beyond its reasonable control. In the event the work is delayed by such causes, the time of completion will be extended in proportion thereto and any premiums for overtime work requested in writing by the Customer shall be reimbursed to the Contractor.

Notwithstanding anything to the contrary contained herein, neither party shall be liable for any delays or failures in performance resulting from acts beyond its reasonable control including, without limitation, acts of God, acts of war or terrorism, shortage of supply, breakdowns or malfunctions, interruptions or malfunction of computer facilities, or loss of data due to power failures or mechanical difficulties with information storage or retrieval systems, labor difficulties or civil unrest. Notwithstanding the foregoing, in the event of such an occurrence, each party agrees to make a good faith effort to perform its obligations hereunder.

Notwithstanding the foregoing, Owner/Contractor agrees to a 10 day review and approval period for all submissions that affect schedule and/or payment.

In the event of extraordinary demands placed on the Contractor due to delay, disruption, hindrance and other causes beyond Contractor's reasonable control, Contractor reserves the right to file a claim for extension of time.

The Contract Price has been determined on the basis of straight time and normal work week. No overtime will be worked unless ordered by Customer or their representative. In the event overtime is worked, the overtime premium rates prescribed by applicable law will be paid as an extra.



THE STRENGTH BETWEEN SUN AND SHADE™

Liquidated Damages shall not be assessed to Contractor or Customer under any circumstances.

**SECTION 11.** <u>SUSPENSION OF WORK.</u> If any payment is not made to Contractor as required under this Contract for an amount not in dispute and Customer does not provide a plan to cure, Contractor may suspend work until such payment is made. Contractor may also suspend work under this Contract if a dispute over payment for extra work, differing site conditions, changes by Customer or other circumstances beyond Contractor's control will cause the Contractor to suffer financial hardship if Contractor is required to continue the work. Contractor may request that Customer provide written proof of Customer's ability to pay Contractor for the work remaining to be performed by Contractor at any time prior to or during performance of this Contract if the Contractor has a reasonable grounds for insecurity with respect to Customer's performance. Failure of Customer to provide such proof shall be justification for Contractor's suspension of work under this Contract.

Any suspension of work under this Contract will also suspend the progress and completion dates set forth herein.

**SECTION 12. INSPECTION OF THE WORK**. The Contractor shall make the work accessible during normal business hours and provide reasonable time for inspection by the Customer.

**SECTION 13.** <u>SITE ACCESS AND RIGHTS OF WAY.</u> The Customer shall provide, no later than the date when needed by the Contractor, all necessary access to the areas upon which the Work is to be performed, including convenient access to property and any other property designated in the Contract Documents for use or access by the Contractor. Customer shall continue to provide such access until completion of the Contract. Any failure to provide such access shall entitle the Contractor to an equitable adjustment in the Contract price and the Contract time.

**SECTION 14.** <u>**REPORTS AND SURVEYS.</u>** Contractor shall not be responsible for conditions not reasonably discoverable prior to the commencement of Contractor's work hereunder. If conditions are discovered and such conditions cause an increase in the cost of, or the time required for, performance of the work, the Contract Price and the time for completion will be equitably adjusted.</u>

**SECTION 15.** <u>PERMITS, LICENSES AND REGULATIONS.</u> Permits and licenses of a temporary nature necessary for the prosecution of the Work shall be obtained and paid for by the Contractor. The Customer shall assist the Contractor in obtaining such permits and licenses.



THE STRENGTH BETWEEN SUN AND SHADE™

The Customer shall assume all monetary and legal risk if the Customer desires to procure material before permits have been received from the appropriate governing authority and written direction will be required.

The Contractor shall give all notices and comply with all laws, ordinances, rules and regulations bearing on the performance of the Work. If the Contractor observes that drawings, specifications or other Contract documents are at variance with such laws, ordinances, rules and regulations, the Customer shall promptly be notified and, if reasonably necessary, an adjustment made to the Contract time or Contract Price as mutually agreed to by the parties in writing.

**SECTION 16.** <u>TAXES.</u> Unless otherwise provided herein, the amount of any present or future sales or other tax, Federal, State or City, which Contractor now, or hereafter shall be required to pay, on behalf of the Customer with respect to any material covered by this Contract shall be added to the Contract Price herein to the extent supported by documentation and paid by the Customer in the same manner and with the same effect as if originally added hereto.

**SECTION 17.** <u>BONDS.</u> Any bonds required by Customer shall be furnished by the Contractor, and the cost thereof added to and become a part of this Contract and Contract Price, unless otherwise provided in Contractor's Proposal. All such additional costs require prior notice to the Customer and Customer's prior written approval.

**SECTION 18.** <u>INDEMNIFICATION.</u> To the extent permitted by law, each Party (the "Indemnitor") hereby indemnifies and agrees to defend and hold harmless the other Party (the "Indemnitee") and its directors, officers, managers, employees and agents (collectively "Representatives") from and against any and all indemnity claims, whether or not involving a third-party claim, caused by, resulting from, relating to or arising out of any breach of any of the representations, warranties or covenants made in this Agreement by the Indemnitor or any of its directors, officers, employees or agents or any negligence or intentional misconduct on the part of the Indemnitor or any of its directors, officers, employees or agents, provided that the Indemnitor will not have any obligation to indemnify the Indemnitee or its Representatives from or against any indemnity claims to the extent caused by, resulting from, relating to or arising out of a breach of this Agreement or the negligence or willful misconduct of the Indemnitee or any of its Representatives.

**SECTION 19.** <u>INSURANCE.</u> Commercial General Liability, Automobile Liability and Workers' Compensation Insurance are carried by the Contractor, and certificates will be delivered upon request. The Contractor agrees to procure and maintain, at its own cost, general liability Insurance. Limits of liability will be maintained at \$1,000,000 per occurrence/\$2,000,000 annual aggregate. Coverage will include Products Completed Operations and medical expense of



THE STRENGTH BETWEEN SUN AND SHADE™

\$1,000,000. Contractor shall procure and maintain Workers' Compensation insurance in accordance with applicable law. Contractor shall furnish to Customer with a completed certificate of insurance coverage, and which specifically requires thirty (30) calendar days prior notice to Customer of cancellation, termination or any material change of any such insurance policy. Wrap Insurance Policies, including but not limited to Owner Controlled Insurance Programs, are expressly excluded from, and shall not be included under this Contract.

**SECTION 20.** <u>LIABILITY AND CONSEQUENTIAL DAMAGES.</u> Contractor bears the risk of accidental damage or destruction of the project prior to final completion and acceptance. It is hereby stipulated that the Contractor shall not be held responsible or liable for any loss, damage, or delay caused by fire, strikes, accident, or any other cause beyond the Contractor's control after final completion and acceptance. In no event shall either party be liable for any lost profits or consequential, indirect, incidental or punitive damages.

**SECTION 21.** <u>ASSIGNMENT AND TRANSFER.</u> Neither party shall assign or transfer this Contract without prior written consent of the other party. Assignment of Contractor's agreement shall be limited to a party that assumes all obligations of the Customer and is no less able from a financial or operational standpoint to fulfill the Customer's obligations in a timely manner.

Title to any of the material sold or installed hereunder by the Contractor shall remain in the Contractor until paid for by Customer, and in the event such materials are affixed in realty, it is expressly understood and agreed that they shall remain personal property subject to removal by the Contractor. The Contractor is responsible for claims for damage to said realty or buildings caused by the removal of said materials or any part thereof that are reasonably avoidable.

**SECTION 22.** <u>**TERMINATION.</u>** The Customer reserves the right to terminate the work for its convenience upon notice in writing to the Contractor. In such an event, the Contractor shall be paid its actual costs for the portion of the work performed through the date of termination, and for all of Contractor's incurred costs of termination, including demobilizations and any termination charges by vendors and subcontractors, plus 20% of all of Contractor's actual and incurred costs for overhead and profit. Contractor shall make every reasonable effort to contain costs associated with any termination.</u>

**SECTION 23.** <u>DISPUTE RESOLUTION.</u> In the event of any dispute, controversy or claim arising out of or relating to this Contract or its alleged breach, which cannot be resolved by mutual agreement through direct discussion, shall be settled by mediation with a neutral third-party mediator acceptable to both parties before recourse to any form of binding resolution. The cost of any legal proceedings shall be shared equally by the parties participating.



THE STRENGTH BETWEEN SUN AND SHADE™

**SECTION 24.** <u>WARRANTY.</u> Contractor shall provide a workmanship warranty for one (1) year from the date of Contractor's substantial completion of the Work or as otherwise provided by applicable law, whichever is greater. Contractor's warranty period shall not be extended by corrective work performed by Contractor. Contractor will pass through all individual supplier warranties with specific details on the coverage. Manufacturer warranties shall be provided upon completion of the work to the Customer.

**SECTION 25.** <u>ACCEPTANCE AND AUTORIZATION.</u> The Contractor and Customer, by accepting this contract, agree to all of the conditions and terms specified herein.

### SECTION 26. SPECIAL PROVISIONS.

Addendum to construction contract, M Bar C's Proposal, Rate sheet, and Schedule of Values are attached hereto and made a part hereof.

### Notice:

Contractors are required by law to be licensed and regulated by the Contractors State License Board which has jurisdiction to investigate complaints against contractors if a complaint regarding a patent act or omission is filed within four (4) years of the date of the alleged violation. A complaint regarding a latent act or omission pertaining to structural defects must be filed within 10 years of the date of the alleged violation. Any questions concerning a contractor may be referred to the Registrar, Contractors State License Board, P.O. Box 26000, Sacramento, California, 95826.

STATE LAW REQUIRES ANYONE WHO CONTRACTS TO DO CONSTRUCTION WORK TO BE LICENSED BY THE CONTRACTORS' STATE LICENSE BOARD IN THE LICENSE CATEGORY IN WHICH THE CONTRACTOR IS GOING TO BE WORKING – IF THE TOTAL PRICE OF THE JOB IS \$500 OR MORE (INCLUDING LABOR AND MATERIALS).

LICENSED CONTRACTORS ARE REGULATED BY LAWS DESIGNED TO PROTECT THE PUBLIC. IF YOU CONTRACT WITH SOMEONE WHO DOES NOT HAVE A LICENSE, THE CONTRACTORS' STATE LICENSE BOPARD MAY BE UNABLE TO ASSIST YOU WITH A COMPLAINT. YOUR ONLY REMEDY AGAINST AN UNLICENSED CONTRACTOR MAY BE IN CIVIL COURT, AND YOU MAY BE LIABLE FOR DAMAGES ARISING OUT OF ANY INJURIES TO THE CONTRACTOR OR HIS OR HER EMPLOYEES.

YOU MAY CONTACT THE CONTRACTORS' STATE LICENSE BOARD TO FIND OUT IF THIS CONTRACTOR HAS A VALID LICENSE. THE BOARD HAS COMPLETE INFORMATION ON THE HISTORY OF LICENSED CONTRACTORS, INCLUDING ANY POSSIBLE SUSPENSIONS, REVOCATIONS, JUDGMENTS, AND CITATIONS. THE BOARD HAS OFFICES THROUGHOUT CALIFORNIA. PLEASE





THE STRENGTH BETWEEN SUN AND SHADE"

CHECK THE GOVERNMENT PAGES OF THE WHITE PAGES FOR THE OFFICE NEAREST YOU OR CALL 1-800-321 CSLB FOR MORE INFORMATION.



THE STRENGTH BETWEEN SUN AND SHADE"

CONTRACTOR:

("Company Name")

M Bar C Construction, Inc.

Date: \_\_\_\_\_

Date: \_\_\_\_\_

Signature

Print Name

Print Name

Signature

Title

Title



# Exhibit B

Site Improvement Site Plan



136 of 258









Exhibit C

RFP



# MAGNOLIA PUBLIC SCHOOLS

Request for Qualifications/Proposals to Provide a Solar Shade Structure for Magnolia Science Academy 1 | Reseda

> Due Date: May 7, 2021

1

### 1.0 INTRODUCTION

Magnolia Education & Research Foundation doing-business as Magnolia Public Schools ("**MPS**"), a charter school management organization, operates Magnolia Science Academy 1 ("**MSA-1**") located at 18220 Sherman Way and 18238 Sherman Way in the Reseda neighborhood of Los Angeles. The former houses MSA-1's high school and the latter houses MSA-1's middle school. The purpose of this RFP is to solicit bids to design and build a solar shade structure (the "**Project**") in the existing surface parking lot belonging to the middle school (the "**Project Site**"). The Project is part of a larger site improvement plan for the middle school parking lot wherein recreation and open space will be developed (the "**Site Improvement Plan**"). The location of the Project Site is shown on <u>Exhibit A</u>. The overall Site Improvement Plan is shown on <u>Exhibit B</u>.

Franco Architects is the architect of record for the Site Improvement Plan. The successful respondent will be required to coordinate their efforts with them and the rest of the design team.

#### Site Tour

A site tour will be facilitated if requested.

Proposals Due

Responses to the RFP are due no later than 5:00 PM (PST), Friday, May 7, 2021, to the following individual:

Patrick Ontiveros Director of Facilities Magnolia Public Schools 250 East 1<sup>st</sup> Street Suite 1500 Los Angeles, CA 90012 pontiveros@magnoliapublicschools.org (213) 628-3634 x 103

Questions regarding this RFP may be directed to the individual identified above via email.

Proposal Format:

One (1) electronic PDF copy (by email) of your proposal must be delivered to the person indicated by the deadline stated above and one (1) hard copy must be delivered to the above address no later than two (2) business days thereafter. Please endeavor to keep any emailed material to a single manageable file size (at or about 10 MBs) so that it may be easily distributed to the Selection Committee.

Respondents are encouraged to only include information pertinent to the Project and the Selection Committee's ability to select the vendor best suited to successfully complete this job.

#### Interviews:

Interviews will be held at the discretion of MPS and MSA-1. Interviews, if any, are expected to be held during the week of May 10th. Location and time are to be determined.

If interviews are held, respondents will meet with the Selection Committee for approximately ½ hour to 1 hour. The interview will be an open, unscripted format. Respondents will have 20 minutes to present any information they feel is pertinent followed by questions and answers and general discussion.

#### Magnolia Public Schools

#### Selection Committee:

The Selection Committee will be composed of representatives from MPS and MSA-1, and the Architect.

### 1.1 Timeline

RFP Distributed:	April 22, 2021
Proposals Due:	May 7, 2021
Interviews, if any (exact date and time TBD):	May 10, 11 or 12, 2021
Selection Announced:	May 13, 2021
Contract Execution:	Week of May 17, 2021

### 2.0 PROJECT DESCRIPTION

The Project is the design and construction of a solar shade structure that will cover a basketball court and outdoor eating facilities which will also be constructed. The Site Improvement Plan, including the Project, was part of a zone change application that was approved the City of Los Angeles. Conditions of the zone change approval are contained in Exhibit C attached hereto. Respondent is responsible for reviewing the conditions of approval to insure their Project proposal meets the therein stated requirements. The Project will be 25 feet high and will be 80 feet wide by 125 feet long and will have solar panels on the top. Exhibit B is a depiction of the Site Improvement Plan, including the Project.

A soils report for the immediately adjacent surface parking lot parcel is attached as Exhibit D.

#### 3.0 PROPOSAL FORMAT

Respondent shall format its response as set forth below to facilitate timely review and selection. Please be specific to the RFP, and do not include materials not explicitly requested, such as generic marketing materials.

#### Your response should include the following:

- Letter of interest
- Name of your company and the individual responsible for the account
- Restate all the requirements of Section 4.0 and provide responses to each

See Section 1.0 for additional proposal format clarifications.

#### 4.0 PROPOSAL REQUIREMENTS

### 4.1 Vendor Qualifications and Experience

### 4.1.1 Vendor Description.

Provide a description of your company and why it is qualified to undertake the Project.

### Provide the following:

A minimum of three (3) references, including

- (a) name and scope of the project
- (b) client name and contact information
- (c) contract amount

### 4.1.2 Qualifications and Experience of Key Personnel.

Identify the person(s) that will be principally responsible for working with the MPS and leading this engagement and their qualifications and experience

### 4.1.3 Insurance.

Provide a description of vendor's insurance coverage.

### 4.2 Cost

Respondent's proposal should include an overall not to exceed cost and should be broken down in detail. The proposal should also provide a break-down of any and all other costs and fees including, but not limited to, labor, delivery fees, installation fees, applicable taxes, etc.

### 4.3 **Product Information**

Respondent will confirm its ability to provide the Project as described above in Section 2.2. Where a respondent believes there are multiple options for the Project, the respondent may provide such alternatives explaining the difference in each alternative and describing the cost differences.

### 4.4 Warranty Information

Provide a description of the warranty provided for the Project.

#### 4.5 Schedule

MPS and MSA-1 desire to complete construction of the Project and the Site Improvement Plan in time for the 2022-23 school year. Respondents shall provide an affirmative statement of its ability to collaborate with the Architect to submit plans to the Los Angeles Department of Building and Safety ("LADBS") in order to meet the desired delivery date. Respondent shall also provide an affirmative statement describing the construction duration for the Project upon receipt of permits.

Owner and architect believe that the plans for the Project should be ready for submission to LADBS by July 9, 2021 in order to meet the timeline of completing construction by the start of the 2022-23 school year on or about August deadline.

### 4.6 Contract

The successful respondent will be required to sign an agreement with Owner substantially similar to <u>Exhibit</u> <u>E</u> attach hereto. Please provide an affirmative statement of respondent's concurrence or else any changes that respondent desires to make to the form agreement.

### 5.0 CONTACT

Questions to Owner will be accepted via email by the Director of Facilities identified above. Answers to questions will be provided to all participants as available.

#### 6.0 RFP/Q EXHIBITS

- Exhibit A: Project Site Location
- Exhibit B: Project
- Exhibit C: Letter of Determination
- **Exhibit D:** Soils Report
- Exhibit E: Form of Contract -- AIA A141-2014 Standard Form of Agreement between Owner and Design-Builder

### 7.0 BID ACCEPTANCE/REJECTION

The Owner reserves the right to reject any or all proposals, or cancel the solicitation process at its sole discretion.

### 8.0 PROPOSAL VALIDITY

RFP responses shall be valid until execution of a contract, which is expected to occur on or about May 2021. No changes to information received within the Respondent's proposal shall be changed or altered without approval by the Owner.

# Exhibit A


## Exhibit B

## Site Improvement Plan



## Exhibit C

## Letter of Determination / Conditions of Approval



SOUTH VALLEY AREA PLANNING COMMISSION

200 North Spring Street, Room 272, Los Angeles, California, 90012-4801, (213) 978-1300 www.planning.lacity.org

## **\*CORRECTED LETTER OF DETERMINATION**

MAILING DATE: APR 28 2020

Case No. APCSV-2019-1064-ZC-CDO CEQA: ENV-2019-1065-MND Plan Area: Reseda – West Van Nuys Council District: 3 - Blumenfield

Project Site: 18216 – 18238 West Sherman Way

Applicant: Magnolia Science Academy 1 Representative: Veronica Becerra, Rabuild Commercial Services LLC

At its meeting of **November 14, 2019**, the South Valley Area Planning Commission took the actions below in conjunction with the approval of the following Project:

An outdoor facilities improvement to the Magnolia Science Academy 1 campus. The Project involves the grading of an existing surface parking lot and the construction, use, and maintenance of an open recreation area with a landscaped grass field, a modular restroom and storage facility, and an outdoor covered student activity area with a solar shade structure 25 feet in height. A portion of the parking lot will be re-striped and maintained as a surface parking lot with the addition of landscaping and trees. The Project also includes installation of an eight foot tall concrete masonry unit block wall along the southern property line and a six foot tall security fence along the east and west property lines. The proposed project will not increase the student enrollment capacity of the school.

- 1. **Found**, pursuant to CEQA Guidelines Section 15074(b), after consideration of the whole of the administrative record, including the Mitigated Negative Declaration, No. ENV-2019-1065-MND ("Mitigated Negative Declaration"), and all comments received, with the imposition of the mitigation measures, there is no substantial evidence that the project will have a significant effect on the environment; **Found** the Mitigated Negative Declaration reflects the independent judgment and analysis of the City; **Found** the mitigation measures have been made enforceable conditions on the project; and **Adopt** the Mitigated Negative Declaration Declaration and the Mitigation Monitoring Program prepared for the Mitigated Negative Declaration:
- 2. **Approved** and **recommended** that the City Council adopt, pursuant to Section 12.32 F of the Los Angeles Municipal Code (LAMC), a Zone change from [Q]P-1L-CDO to (T)[Q]C2-1L-CDO to permit the construction and installation of outdoor recreational facilities, a modular restroom a solar shade structure, and landscaping;
- 3. **Approved**, \*<sup>1</sup>pursuant to LAMC Section 13.08 E.3 and the Reseda Central Business District Community Design Overlay, a Design Overlay Plan Approval for the construction,

<sup>&</sup>lt;sup>1</sup> The above-referenced technical correction is not discretionary, therefore, issuance of this corrected letter of determination does not extend or require a new appeal period.

installation, use, and maintenance of a proposed shade structure, modular bathroom, landscaping, and surface parking areas.

- 4. Adopted the Conditions of Approval; and
- 5. **Adopted** the Findings.

This action was taken by the following vote:

Moved:	Bishop
Seconded:	Beatty
Ayes:	Mather, Menedjian
Absent:	Dierking

Vote: 4 – 0

Rafael Vega, Commission Executive Assistant

Fiscal Impact Statement: There is no General Fund impact as administrative costs are recovered through fees.

If you seek judicial review of any decision of the City pursuant to California Code of Civil Procedure Section 1094.5, the petition for writ of mandate pursuant to that section must be filed no later than the 90th day following the date on which the City's decision became final pursuant to California Code of Civil Procedure Section 1094.6. There may be other time limits which also affect your ability to seek judicial review.

Attachments: Zone Change Ordinance, Map, Findings

c: JoJo Pewsawang, City Planner Renata Ooms, City Planning Associate ORDINANCE NO. \_\_\_\_\_

An ordinance amending Section 12.04 of the Los Angeles Municipal Code by amending the Zoning map.

THE PEOPLE OF THE CITY OF LOS ANGELES DO ORDAIN AS FOLLOWS:

Section 1. Section 12.04 of the Los Angeles Municipal Code is hereby amended by changing the zone and zone boundaries shown upon a portion of the zone map attached thereto and made a part of Article 2, Chapter 1 of the Los Angeles Municipal Code, so that such portion of the zoning map shall be as follows:



150 of 258

## [Q] QUALIFIED CLASSIFICATIONS

The subject property shall continue to be subject to the Reseda Central Business District Community Design Overly Permanent [Q] Conditions imposed by Ordinance No. 176558 and adopted on March 1, 2005.

## CONDITIONS FOR EFFECTUATING (T) TENTATIVE CLASSIFICATION REMOVAL

Pursuant to Section 12.32-G of the Municipal Code, the (T) Tentative Classification shall be removed by posting of guarantees through the B-permit process of the City Engineer to secure the following without expense to the City of Los Angeles, with copies of any approval or guarantees provided to the Department of City Planning for attachment to the subject planning case file.

Dedication(s) and Improvement(s). Prior to the issuance of any building permits, the following public improvements and dedications for streets and other rights of way adjoining the subject property shall be guaranteed to the satisfaction of the Bureau of Engineering, Department of Transportation, Fire Department (and other responsible City, regional and federal government agencies, as may be necessary):

## Responsibilities/Guarantees.

- 1. As part of early consultation, plan review, and/or project permit review, the applicant/developer shall contact the responsible agencies to ensure that any necessary dedications and improvements are specifically acknowledged by the applicant/developer.
- 2. Bureau of Engineering. Prior to issuance of sign offs for final site plan approval and/or project permits by the Department of City Planning, the applicant/developer shall provide written verification to the Department of City Planning from the responsible agency acknowledging the agency's consultation with the applicant/developer. The required dedications and improvements may necessitate redesign of the project. Any changes to project design required by a public agency shall be documented in writing and submitted for review by the Department of City Planning.
  - a. Dedication Required: None
  - b. Improvements Required:
    - i. **Sherman Way** Construct a new full-width concrete sidewalk. Remove and replace bad order concrete curb and gutter along the property frontage.
    - ii. **Alley** Reconstruct the 20-foot wide alley and longitudinal concrete gutter as necessary to provide safe access for students and pedestrian traffic. (Note: The site plan does not show the existing public right-of-way for the Alley turning area. This public right-of-way must be shown on all plans.)

## Notes:

Broken curb and/or gutter includes segments within existing score lines that are depressed or upraised by more than <sup>1</sup>/<sub>4</sub> inch from the surrounding concrete work or are separated from the main body of the concrete piece by a crack through the entire vertical segment and greater than 1/8 inch at the surface of the section.

Non-ADA compliant sidewalk shall include any sidewalk that has a cross slope that exceeds 2% and/or is depressed or upraised by more than ¼ inch from the surrounding concrete work or has full concrete depth cracks that have separations greater than 1/8

inch at the surface. The sidewalk also includes that portion of the pedestrian path of travel across a driveway.

All new sidewalk curb and gutter shall conform to the Bureau of Engineering Standard Plans S410-2, S410-4, S442-5 and S444-0.

Install tree wells with root barriers and plant street trees satisfactory to the City Engineer and the Urban Forestry Division of the Bureau of Street Services. The applicant should contact the Urban Forestry Division for further information (213) 847-3077.

Street lighting may be required satisfactory to the Bureau of Street Lighting (213) 847-1551.

Department of Transportation may have additional requirements for dedication and improvements.

Regarding any conflicts with power pole matters, contact the Department of Water and Power at (213) 367-2715.

- c. Provide proper site and street drainages for all streets being improved.
- d. Mainline sewer exists in Sherman Way and in the Alley. All Sewerage Facilities Charges and Bonded Sewer Fees are to be paid prior to obtaining a building permit.
- e. Submit a parking area and driveway plan to the Valley District Office of the Bureau of Engineering and the Department of Transportation for review and approval.

## 3. Department of Transportation.

- a. Access to Sherman Way shall be prohibited. Access to the project site is proposed via the alley.
- b. A driveway apron width of W=30 feet is required for any non-alley access.
- c. A minimum of 20-foot reservoir space is required between any security gate or parking stall and the property line for any non-alley access, to the satisfaction of the Department of Transportation.
- d. A parking area and driveway plan should be submitted to the Citywide Planning Coordination Section of the Department of Transportation for approval prior to submittal of building permit plans for plan check by the Department of Building and Safety. Transportation approvals are conducted at 6262 Van Nuys Blvd., Room 320, Van Nuys, CA 91401.
- e. That permitting fees be paid to the Department of Transportation as required per Ordinance No. 183270 and LAMC Section 19.15 prior to any building permit approval. Note: the applicant may be required to comply with any other applicable fees per this new ordinance.

### 4. Bureau of Street Lighting

a. IMPROVEMENT CONDITION: No street lighting improvements if no street widening per BOE improvement conditions. Otherwise relocate and upgrade street lights; two (2) on Sherman Way. Notes:

The quantity of street lights identified may be modified slightly during the plan check process based on illumination calculations and equipment selection.

Conditions set: 1) in compliance with a Specific Plan, 2) by LADOT, or 3) by other legal instrument excluding the Bureau of Engineering conditions, requiring an improvement that will change the geometrics of the public roadway or driveway apron may require additional or the reconstruction of street lighting improvements as part of that condition.

## 5. Urban Forestry Division.

a. Plant street trees and remove any existing trees within dedicated streets or proposed dedicated streets as required by the Urban Forestry Division of the Bureau of Street Services. Parkway tree removals shall be replanted at a 2:1 ratio. All street tree plantings shall be brought up to current standards. When the City has previously been paid for tree plantings, the sub divider or contractor shall notify the Urban Forestry Division (213-847-3077) upon completion of construction to expedite tree planting.

Note: Removal or planting of any tree in the public right-of-way requires approval of the Board of Public Works. Contact Urban Forestry Division at: (213) 847-3077 for permit information. CEQA document must address parkway tree removal.

## 6. Fire Department.

- a. Access for Fire Department apparatus and personnel to and into all structures shall be required.
- b. One or more Knox Boxes will be required to be installed for LAFD access to project. Location and number to be determined by LAFD Field Inspector. (Refer to FPB Requirement No. 75).
- c. 505.1 Address identification. New and existing buildings shall have approved building identification placed in a position that is plainly legible and visible from the street or road fronting the property.
- d. No building or portion of a building shall be constructed more than 150 feet from the edge of a roadway of an improved street, access road, or designated fire lane.
- e. Fire lane width shall not be less than 20 feet. When a fire lane must accommodate the operation of Fire Department aerial ladder apparatus or where fire hydrants are installed, those portions shall not be less than 28 feet in width.
- f. The width of private roadways for general access use and fire lanes shall not be less than 20 feet, and the fire lane must be clear to the sky.
- g. Fire lanes, where required and dead ending streets shall terminate in a cul-de-sac or other approved turning area. No dead ending street or fire lane shall be greater than 700 feet in length or secondary access shall be required.
- h. Submit plot plans indicating access road and turning area for Fire Department approval.

- i. Adequate off-site public and on-site private fire hydrants may be required. Their number and location to be determined after the Fire Department's review of the plot plan.
- j. All parking restrictions for fire lanes shall be posted and/or painted prior to any Temporary Certificate of Occupancy being issued.
- k. Plans showing areas to be posted and/or painted, "FIRE LANE NO PARKING" shall be submitted and approved by the Fire Department prior to building permit application signoff.
- I. Electric Gates approved by the Fire Department shall be tested by the Fire Department prior to Building and Safety granting a Certificate of Occupancy.
- m. That in order to provide assurance that the proposed common fire lane and fire protection facilities, for the project, not maintained by the City, are properly and adequately maintained, the sub-divider shall record with the County Recorder, prior to the recordation of the final map, a covenant and agreement (Planning Department General Form CP-6770) to assure the following:
  - i. The establishment of a property owners association, which shall cause a yearly inspection to be, made by a registered civil engineer of all common fire lanes and fire protection facilities. The association will undertake any necessary maintenance and corrective measures. Each future property owner shall automatically become a member of the association or organization required above and is automatically subject to a proportionate share of the cost.
  - ii. The future owners of affected lots with common fire lanes and fire protection facilities shall be informed or their responsibility for the maintenance of the devices on their lots. The future owner and all successors will be presented with a copy of the maintenance program for their lot. Any amendment or modification that would defeat the obligation of said association as the Advisory Agency must approve required hereinabove in writing after consultation with the Fire Department.
  - iii. In the event that the property owners association fails to maintain the common property and easements as required by the CC and R's, the individual property owners shall be responsible for their proportional share of the maintenance.
  - iv. Prior to any building permits being issued, the applicant shall improve, to the satisfaction of the Fire Department, all common fire lanes and install all private fire hydrants to be required.
  - v. That the Common Fire Lanes and Fire Protection facilities be shown on the Final Map.
  - vi. The plot plans shall be approved by the Fire Department showing fire hydrants and access for each phase of the project prior to the recording of the final map for that phase. Each phase shall comply independently with code requirements.

Note: The applicant is further advised that all subsequent contact regarding these conditions must be with the Hydrant and Access Unit. This would include clarification, verification of condition compliance and plans or building permit applications, etc., and shall be accomplished <u>by appointment only</u>, in order to assure that you receive service with a minimum amount of waiting please call **(818) 374-4351.** You should advise any consultant representing you of this requirement as well.

## CONDITIONS OF APPROVAL

Pursuant to Section 13.08 of the Los Angeles Municipal Code, the following conditions are hereby imposed upon the use of the subject property:

- 1. **Development.** The use and development of the property shall be in substantial conformance with the plot plan submitted with the application and marked Exhibit "A" except as may be revised as a result of this action.
- 2. Height. The height of the solar shade structure shall not exceed 25 feet.
- 3. **Setbacks.** A 15 foot setback along the southern property line shall be maintained with landscaping and trees.
- 4. Landscape Plan. Landscaping shall be provided in substantial compliance with the submitted Landscape Plan (as seen in Exhibit A). Landscaped areas shall be planted with a variety of plant materials which include shrubs, trees and ground cover, and all plants and trees shall be drought-resistant. Artificial plants are prohibited. All landscaped areas shall be equipped with an automatic sprinkler or drip irrigation system designed to conserve water.
- 5. **Mechanical Equipment**. All mechanical equipment on the roof shall be screened from view. The transformer, if visible from the ally or adjoining properties, shall be screened with landscaping.
- 6. **Lighting.** Lighting shall be installed in the parking lot and within the covered and uncovered recreation areas. Outdoor lighting shall be designed and installed with shielding, such that the light source cannot be seen from adjacent residential properties or the public right-of-way, nor from above.
- 7. **Building Materials.** A gray and blue color scheme as proposed in Exhibit "A" shall be used. No florescent colors shall be used.

## 8. Fence and Wall Materials.

- a. The fence along the east and west property lines shall be combination wrought iron and masonry wall. Chain link fencing is prohibited.
- b. Any masonry walls along any property line should be constructed from decorative brick, stone, split face concrete block, or other decorative material. Masonry block walls should be finished with a masonry cap.
- c. The use of barbed wire, razor wire, and concertina wire is prohibited.
- 9. **Maintenance**. The subject property (including all trash storage areas, parking facilities, sidewalks, yard areas, parkways, and exterior walls and fences along the property lines) shall be maintained in an attractive condition and shall be kept free of trash and debris.
- 10. **Solar Panels.** Solar panels shall be installed on the rooftop of the shade structure to be connected to the school building's electrical system. The entirety of the roof area shall be reserved for the installation of a solar photovoltaic system, to be installed prior to the issuance of a certificate of occupancy, in substantial conformance with the plans stamped "Exhibit A".

## Administrative Conditions of Approval

- 11. **Approval, Verification and Submittals.** Copies of any approvals, guarantees or verification of consultations, review or approval, plans, etc., as may be required by the subject conditions, shall be provided to the Department of City Planning for placement in the subject file.
- 12. **Code Compliance.** Area, height and use regulations of the (T)(Q)RD3-1 Zone classification of the subject property shall be complied with, except where herein conditions are more restrictive.
- 13. **Covenant.** Prior to the issuance of any permits relative to this matter, an agreement concerning all the information contained in these conditions shall be recorded in the County Recorder's Office. The agreement shall run with the land and shall be binding on any subsequent property owners, heirs or assign. The agreement must be submitted to the Department of City Planning for approval before being recorded. After recordation, a copy bearing the Recorder's number and date shall be provided to the Department of City Planning for attachment to the file.
- 14. **Definition.** Any agencies, public officials or legislation referenced in these conditions shall mean those agencies, public officials, legislation or their successors, designees or amendment to any legislation.
- 15. **Enforcement.** Compliance with these conditions and the intent of these conditions shall be to the satisfaction of the Department of City Planning and any designated agency, or the agency's successor and in accordance with any stated laws or regulations, or any amendments thereto.
- 16. **Building Plans.** Page 1 of the grants and all the conditions of approval shall be printed on the building plans submitted to the Department of City Planning and the Department of Building and Safety.
- 17. **Corrective Conditions.** The authorized use shall be conducted at all time with due regards to the character of the surrounding district, and the right is reserved to the North Valley Area Planning Commission, or the Director pursuant to Section 12.27.1 of the Municipal Code to impose additional corrective conditions, if in the Commission's or Director's opinion such conditions are proven necessary for the protection of persons in the neighborhood or occupants of adjacent property.
- 18. **Expediting Processing Section.** Prior to the clearance of any conditions, the applicant shall show that all fees have been paid to the Department of City Planning Expedited Processing Section.

## 19. Indemnification and Reimbursement of Litigation Costs.

Applicant shall do all of the following:

a. Defend, indemnify and hold harmless the City from any and all actions against the City relating to or arising out of, in whole or in part, the City's processing and approval of this entitlement, including <u>but not limited to</u>, an action to attack, challenge, set aside, void or otherwise modify or annul the approval of the entitlement, the environmental review of the entitlement, or the approval of subsequent permit decisions or to claim personal property damage, including from inverse condemnation or any other constitutional claim.

- b. Reimburse the City for any and all costs incurred in defense of an action related to or arising out of, in whole or in part, the City's processing and approval of the entitlement, including but not limited to payment of all court costs and attorney's fees, costs of any judgments or awards against the City (including an award of attorney's fees), damages and/or settlement costs.
- c. Submit an initial deposit for the City's litigation costs to the City within 10 days' notice of the City tendering defense to the Applicant and requesting a deposit. The initial deposit shall be in an amount set by the City Attorney's Office, in its sole discretion, based on the nature and scope of action, but in no event shall the initial deposit be less than \$50,000. The City's failure to notice or collect the deposit does not relieve the Applicant from responsibility to reimburse the City pursuant to the requirement in paragraph (b).
- d. Submit supplemental deposits upon notice by the City. Supplemental deposits may be required in an increased amount from the initial deposit if found necessary by the City to protect the City's interests. The City's failure to notice or collect the deposit does not relieve the Applicant from responsibility to reimburse the City pursuant to the requirement (b).
- e. If the City determines it necessary to protect the City's interests, execute an indemnity and reimbursement agreement with the City under terms consistent with the requirements of this condition.

The City shall notify the applicant within a reasonable period of time of its receipt of any action and the City shall cooperate in the defense. If the City fails to notify the applicant of any claim, action or proceeding in a reasonable time, or if the City fails to reasonably cooperate in the defense, the applicant shall not thereafter be responsible to defend, indemnify or hold harmless the City.

The City shall have the sole right to choose its counsel, including the City Attorney's office or outside counsel. At its sole discretion, the City may participate at its own expense in the defense of any action, but such participation shall not relieve the applicant of any obligation imposed by this condition. In the event the Applicant fails to comply with this condition, in whole or in part, the City may withdraw its defense of the action, void its approval of the entitlement, or take any other action. The City retains the right to make all decisions with respect to its representations in any legal proceeding, including its inherent right to abandon or settle litigation.

For purposes of this condition, the following definitions apply:

"City" shall be defined to include the City, its agents, officers, boards, commission, committees, employees and volunteers.

"Action" shall be defined to include suits, proceedings (including those held under alternative dispute resolution procedures), claims or lawsuits. Actions includes actions, as defined herein, alleging failure to comply with any federal, state or local law.

Nothing in the definitions included in this paragraph are intended to limit the rights of the City or the obligations of the Applicant otherwise created by this condition.

## FINDINGS

## General Plan/Charter Findings

## 1. General Plan Land Use Designation.

General Plan Land Use Designation. The subject property is located within the Reseda – West Van Nuys Community Plan area which was updated by the City Council on November 17, 1999 and designates the subject property for Community Commercial land uses corresponding to the CR, C2, C4, RAS3, RAS4, P and PB Zones. The site is zoned with the [Q]P-1L-CDO Zone. The proposed C2 zone is a corresponding zone for the Community Commercial land use designation. The proposed outdoor recreation facilities, bathroom with storage area, and solar shade structure is consistent with development permitted in the proposed (T)[Q]C2-1L-CDO Zone and is consistent with the regulations of the qualified Q condition. Therefore, the project is in substantial conformance with the purposes, intent and provisions of the General Plan as reflected in the adopted Framework Element and Community Plan.

## 2. General Plan Text

a. The **Framework Element** of the General Plan (Framework Element) was adopted by the City of Los Angeles in December 1996 and re-adopted in August 2001. The Framework Element provides guidance regarding policy issues for the entire City of Los Angeles, including the project site. The Framework Element also sets forth a Citywide comprehensive long-range growth strategy and defines Citywide polices regarding such issues as land use, housing, urban form, neighborhood design, open space, economic development, transportation, infrastructure, and public services.

The Framework Element Long Range Land Use Diagram for the San Fernando Valley identifies Sherman Way in Reseda as a "Community Center." Community Centers are a focal point for surrounding residential neighborhoods and contain a diversity of uses such as small offices and overnight accommodations, cultural and entertainment facilities, and schools and libraries, in addition to neighborhood oriented services. The proposed Zone Change to a C2 zone is consistent with a "Community Center" and the proposed school recreation facilities are consist with the long range vision for the area.

The Framework Element includes the following Goals, Objectives and Policies relevant to the request:

GOAL 9N: Public schools that provide a quality education for all of the City's children, including those with special needs, and adequate school facilities to serve every neighborhood in the City so that students have an opportunity to attend school in their neighborhoods.

Objective 9.32 Work constructively with LAUSD to promote the siting and construction of adequate school facilities phased with growth.

Policy 9.32.2: Explore creative alternatives for providing new school sites in the City, where appropriate.

The proposed Zone Change from [Q]P-1L-CDO to (T)[Q]C2-1L-CDO will allow for the development of a recreational facility where an underutilized surface parking is currently located. The project is in conjunction with Magnolia Science Academy 1, an LAUSD

Charter School, which has been conducting student lunch, student recreation and dropoff/pick-up actives incidental to the school, in the surface parking lot through Zone Variances since 2006 (ZA-2005-3787-ZV, ZA-2008-0748-ZV, ZA-2014-0995-ZV). The Zone Change is consistent with Framework Policy 9.32.2 as it would allow an underutilized site to be developed into a permanent school facility.

Therefore, the Zone Change is consistent with the Distribution of Land Use goals, objectives and policies of the General Plan Framework Element.

b. **Reseda – West Van Nuys Community Plan.** The Community Plan text includes the following relevant land use Objectives and Policies:

GOAL 4: PUBLIC SCHOOLS THAT PROVIDE A QUALITY EDUCATION FOR ALL OF THE CITY'S CHILDREN, INCLUDING THOSE WITH SPECIAL NEEDS, AND ADEQUATE SCHOOL FACILITIES TO SERVE EVERY NEIGHBORHOOD IN THE CITY.

Objective 4-1: Work constructively with LAUSD to promote the siting and construction of adequate school facilities phased with growth.

Policy 4-1.1: Explore creative alternatives for providing new school sites in the City, where appropriate.

The proposed Zone Change from [Q]P-1L-CDO to (T)[Q]C2-1L-CDO will allow for the redevelopment of an underutilized surface parking lot with proposed recreational facilities in conjunction with the Magnolia Science Academy 1, an LAUSD Charter School. The school has been conducting student lunch, student recreation and drop-off/pick-up actives incidental to the school, in the surface parking lot through Zone Variances since 2006 (ZA-2005-3787-ZV, ZA-2008-0748-ZV, ZA-2014-0995-ZV). The Zone Change would allow for the installation of landscaping and a natural turf lawn and the construction of a permanent structure for active recreation and a lunch area. Therefore the Zone Change is consistent with Framework Policy 9.32.2 as it would allow an underutilized site to be developed into a permanent school facility.

- c. The **Mobility Element** of the General Plan (Mobility Plan 2035) is not likely to be affected by the recommended action herein. The site abuts an alley to the north. The alley is dedicated to a width of 20 feet and improved with asphalt roadway. As part of the (T) Tentative Classification conditions incorporated herein, the applicant must reconstruct the 20-foot wide alley and longitudinal concrete gutter as necessary to provide safe access for students and pedestrian traffic. The site does not abut any other public right-of-way.
- d. The **Sewerage Facilities Element** of the General Plan will not be affected by the recommended action. While the sewer system might be able to accommodate the total flows for the proposed project, further detailed gauging and evaluation may be needed as part of the permit process to identify a specific sewer connection point. If the public sewer has insufficient capacity then the developer will be required to build sewer lines to a point in the sewer system with sufficient capacity. A final approval for sewer capacity and connection permit will be made at that time. Ultimately, this sewage flow will be conveyed to the Hyperion Treatment Plant, which has sufficient capacity for the project.

## Zone Change and "T"/"Q" Classification Findings

## 3. Pursuant to Section 12.32 of the Municipal Code, the zone change and classifications are necessary because:

- a. <u>Public Necessity</u>: On April 29, 2019, Mayor Eric Garcetti released LA's Green New Deal (Sustainable City pLAn), a roadmap to achieve short-term results while setting the path to strengthen the transformation of the City in the decades to come. As part of the plan, the Mayor set forth a goals to supply 100% renewable energy by 2045 and increase tree canopy in areas of greatest need by at least 50% by 2028 to grow a more equitable urban forest that provides cooling, public health, habitat, energy savings, and other benefits. The Zone Change would allow an asphalt surface parking lot site to be redeveloped into an outdoor recreation facility with a solar shade structure and many new trees. By providing solar panels and increasing the tree canopy coverage on site, the project is aligned with the city's sustainability goals and is consistent with public necessity.
- b. <u>Convenience</u>: The project site is located in the Reseda Central Business District, an area that is highly urbanized with commercial uses, residential uses, and transportation infrastructure. The proposed project would improve the outdoor recreational facilities of an existing school. Granting the Zone Change to the C2 Zone would allow residents local access to an LAUSD Charter School that offers high quality, on-site, outdoor recreational facilities for its students. Improving the school's facilities provides a public convenience allowing residents the option to send their children to an amenity rich school nearby.
- c. <u>General Welfare</u>: Granting the Zone Change to the C2 Zone would allow the redevelopment of an underutilized surface parking lot with improved outdoor recreational areas in conjunction with an existing school. Pursuant to three previously granted Zone Variances, the school already programs a portion of the P-zoned surface parking lot as a recreation area and lunch area for students. The Zone Change to the C2 Zone would allow the school to continue the existing use while also upgrading the outdoor school facilities with a solar shade structure, a natural turf grass field, landscaping, and trees. Students at the school would benefit from the improved facilities, additional greenery, and shade.
- d. <u>Good Zoning Practice</u>: The proposed C2 Zone is identified as a corresponding zone of the Community Commercial land use designation. The development would be consistent with the land use designation. The proposed Zone Change from [Q]P-1L-CDO to (T)[Q]C2-1L-CDO will allow for the development of an underutilized surface parking lot with recreational facilities for an LAUSD Charter School, which has been conducting student lunch, student recreation and drop-off/pick-up actives in the surface parking lot through Zone Variances since 2006 (ZA-2005-3787-ZV, ZA-2008-0748-ZV, ZA-2014-0995-ZV). As such, the C2-Zone would allow site's facilities to be upgraded while continuing a long standing authorized use of the site without the need for future Zone Variances.
- e. <u>"T" and "Q" Classification Findings</u>: Per Section 12.32-G,1 and 2 of the Municipal Code, the current action, as recommended, has been made contingent upon compliance with new "T" and "Q" conditions of approval. Such limitations are necessary to ensure the identified dedications, improvements, and actions are undertaken to meet the public's needs, convenience, and general welfare served by the required actions. The conditions that limit the scale, design and scope of future development on the site are also necessary to protect the best interests of and to assure a development more compatible with surrounding properties and the overall pattern of the existing development in the community, to secure an appropriate development in harmony with the General Plan as discussed in Findings Section 1. The existing Permeant Qualified Q condition is not

proposed for change as it has been in effect since 2005 and is related to the CDO. The Q prohibits certain uses, and regulates building height, signage, and parking.

## Design Overlay Plan Approval Findings pursuant to LAMC Section 13.08

## 4. The project substantially complies with the adopted Community Design Overlay Guidelines and Standards.

The proposed project would replace a portion of an existing surface parking lot with improved outdoor school facilities including an open recreation area with a landscaped grass field, a modular restroom and storage facility, and an outdoor covered student activity area with a solar shade structure 25 feet in height. A portion of the parking lot will be re-striped and maintained as a surface parking lot with the addition of landscaping and trees. The project also includes installation of an eight-foot tall concrete masonry unit block wall along the southern property line and a six-foot tall security fence along the east and west property lines.

The proposed Project is located in a Community Commercial General Plan Land Use Designation within in the Reseda-West Van Nuys Community Plan and is further located within the Central Business District of Reseda Community Design Overlay District (CDO). The project, as proposed and conditioned herein, demonstrates compliance with the guidelines and standards (Section 5) of the adopted CDO as described below.

In compliance with Section 5.A of the CDO, proposed fences and freestanding walls are conditioned to comply with Guideline 6, Standard 6a, and Standard 6b of the CDO. Consistent with the CDO, a combination wrought iron and masonry wall is proposed. Masonry wall materials are conditioned to comply with the CDO's standard materials (brick, stone, split face concrete, or other decorate materials). A condition to prohibit the use of chain link fencing is also included.

In compliance with Section 5.B of the CDO, proposed landscaping includes softscape and hardscape, introduces a variety of species of trees and shrubs on a site where there were previously none, and includes a required minimum 15 foot wide landscaped buffer along the southern property line abutting single family zoned properties.

In compliance with Section 5.C of the CDO, building colors are conditioned to coordinate with the existing school building's blue gray color scheme. Highly reflective colors are prohibited in the conditions. The conditions require shielded lighting to be provided on-site. Consistent with the security devices guidelines and standards, a condition has been added prohibiting barbed wire, razor wire, and concertina wire.

Section 5.D of the CDO regarding signage does not apply to the proposed project as no signage is proposed for the site.

Therefore, the project complies will all subsections (A through D) of the Section 5, *Design Guidelines and Standards* of the CDO.

# 5. The structures, site plan and landscaping are harmonious in scale and design with existing development and any cultural, scenic or environmental resources adjacent to the site and in the vicinity.

The property to the north of the alley is developed with the Magnolia Science Academy 1 classroom and administrative buildings. The eastern and western adjoining properties are developed with surface parking. The southern adjoining properties are zoned R1 and are

developed with four single-family dwellings. The site is not adjacent to, or in the vicinity of, any recognized and/or known cultural, scenic, or environmental resources.

The project's proposed structures, site plan and landscape plan are harmonious in scale and design with existing development and will provide a visual improvement to the surrounding community. The site's asphalt surface parking lot will be replaced with an open recreation area with a landscaped grass field, a modular restroom and storage facility, and an outdoor covered student activity area with a solar shade structure. A portion of the parking lot will be re-striped and maintained as a surface parking lot with the addition of landscaping and trees. A 15-foot wide landscaped buffer with shrubs and trees will be located on the southern property line adjacent to the single family homes, providing screening, beatification, and privacy for the neighboring properties. The solar shade structure is proposed 25 feet in height and is compliant with the height requirements for structures in commercial zones that abut R1 single family zones (LAMC Section 12.21.1-A.10). Currently there is no landscaping in the parking lot and thus the proposed landscaped recreational areas and landscaped parking lot will add greenery to an otherwise asphalt parking lot. Thus, the structures, site plan and landscaping are harmonious in scale and design with existing development.

## **Environmental Findings**

- 6. **Environmental Findings.** Pursuant to CEQA Guidelines Section 15074(b), after consideration of the whole of the administrative record, including the Mitigated Negative Declaration, No. ENV-2019-1065-MND, as circulated on August 22, 2019, ("Mitigated Negative Declaration"), and all comments received, with the imposition of mitigation measures, there is no substantial evidence that the project will have a significant effect on the environment. The Mitigated Negative Declaration reflects the independent judgment and analysis of the City. The mitigation measures have been made enforceable conditions on the project and a Mitigation Monitoring Program prepared for the Mitigated Negative Declaration
- 7. **Flood Insurance.** The National Flood Insurance Program rate maps, which are a part of the Flood Hazard Management Specific Plan adopted by the City Council by Ordinance No. 172,081, have been reviewed and it has been determined that this project is located in Zone C, areas of minimal flooding.



## MAGNOLIA SCIENCE ACADEMY 1 MASTERPLANNING

## **Received by Planning** September 12, 2019

EXHIBIT A APCSV-2019-1064-ZC-CDO Page 1 of 7

1. P. 1.

Powered by BoardOnTrack

SEPTEMBER 10, 2019

gkkworks CANNONDESIGN







## **CURRENT & PROPOSED ZONING**

MAGNOLIA SCIENCE ACADEMY 1 - MASTERPLANNING

**EXHIBIT A** APCSV-2019-1064-ZC-CDO Page 2 of 7

Powered by BoardOnTrack

Magnolia Public Schools - Regular Board Meeting - Agenda - Thursday August 12, 2021 at 6:00 PM

SEPTEMBER 10, 2019 JKKWORKS CANNONDESIGN





## CAMPUS ACCESS PATHWAYS

**MAGNOLIA SCIENCE ACADEMY 1 - MASTERPLANNING** 

## EXHIBIT A APCSV-2019-1064-ZC-CDO Page 3 of 7

SEPTEMBER 10, 2019 JKKWORKS CANNONDESIGN

Powered by BoardOnTrack



Magnolia Public Schools - Regular Board Meeting - Agenda - Thursday August 12, 2021 at 6:00 PM



**MAGNOLIA SCIENCE ACADEMY 1 - MASTERPLANNING** 

APCSV-2019-1064-ZC-CDO Page 5 of 7

Powered by BoardOnTrack



SOLAR SHADE STRUCTURE





MODULAR RESTROOM





NEW LIGHT POLES AT WEST PRPOERTY TO MATCH LIGHT POLES AT EAST PROPERTY (PICTURED ABOVE)



PROPOSED LIGHT FIXTURE AT SHADE STRUCTURE METEOR WHIZ SERIES OUTDOOR HIGH CEILING FIXTURE



NEW WROUGHT IRON FENCE AT SOUTH PROPERTIES TO MATCH NEW FENCE AT NEW MSA-1 HIGH SCHOOL (PICTURED ABOVE)









## **MATERIALS AND FINISHES**

**MAGNOLIA SCIENCE ACADEMY 1 - MASTERPLANNING** 

1



WALKWAY PAVING ALTERNATING PATTERN:



## EXTERIOR PAINT FOR TEMPORARY RESTROOM STRUCTURE TO MATCH EXISTING PAINT COLOR AT MSA-1 HIGH SCHOOL BUILDING



DE5787 PACIFIC PALISADE

## EXTERIOR PAINT FOR SHADE STRUCTURE TO MATCH EXISTING PAINT COLOR AT MSA-1 HIGH SCHOOL BUILDING



## DE6348 DRAW YOUR SWORD





Powered by BoardOnTrack



DE6372 LACE VEIL



CUSTOM BUILT SHADE STRUCTURE







170 of 258

## Exhibit D

## Soils Report

## Follow Link:

https://www.dropbox.com/sh/r50ihghnhjzrwve/AAAiRKUhML-PFhnvSqGiiQeha?dl=0

## Exhibit E

# Form of Agreement -- AIA A141-2014 Standard Form of Agreement between Owner and Design-Builder



## Standard Form of Agreement Between Owner and Design-Builder

**AGREEMENT** made as of the « » day of « » in the year « » (*In words, indicate day, month and year.*)

### **BETWEEN** the Owner:

(Name, legal status, address and other information)

~	≫≪	))
«	»	
«	»	
~	»	

and the Design-Builder: (Name, legal status, address and other information)

for the following Project: (Name, location and detailed description)

« »

« » « »

The Owner and Design-Builder agree as follows.



author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

Consultation with an attorney is also encouraged with respect to professional licensing requirements in the jurisdiction where the Project is located.





**ELECTRONIC COPYING** of any portion of this AIA<sup>®</sup> Document to another electronic file is prohibited and constitutes a violation of copyright laws as set forth in the footer of this document.

AIA Document Al41" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No. 4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. (2034726772)

#### TABLE OF ARTICLES

- 1 **GENERAL PROVISIONS**
- 2 COMPENSATION AND PROGRESS PAYMENTS
- 3 GENERAL REQUIREMENTS OF THE WORK OF THE DESIGN-BUILD CONTRACT
- 4 WORK PRIOR TO EXECUTION OF THE DESIGN-BUILD AMENDMENT
- WORK FOLLOWING EXECUTION OF THE DESIGN-BUILD AMENDMENT 5
- 6 CHANGES IN THE WORK
- 7 **OWNER'S RESPONSIBILITIES**
- 8 TIME
- 9 PAYMENT APPLICATIONS AND PROJECT COMPLETION
- 10 PROTECTION OF PERSONS AND PROPERTY
- 11 UNCOVERING AND CORRECTION OF WORK
- 12 **COPYRIGHTS AND LICENSES**
- 13 **TERMINATION OR SUSPENSION**
- CLAIMS AND DISPUTE RESOLUTION 14
- 15 **MISCELLANEOUS PROVISIONS**
- SCOPE OF THE AGREEMENT 16
- TABLE OF EXHIBITS
- Α **DESIGN-BUILD AMENDMENT**
- В INSURANCE AND BONDS
- С SUSTAINABLE PROJECTS

## ARTICLE 1 GENERAL PROVISIONS

## § 1.1 Owner's Criteria

This Agreement is based on the Owner's Criteria set forth in this Section 1.1. (Note the disposition for the following items by inserting the requested information or a statement such as "not

applicable" or "unknown at time of execution." If the Owner intends to provide a set of design documents, and the requested information is contained in the design documents, identify the design documents and insert "see Owner's design documents" where appropriate.)

### **§ 1.1.1** The Owner's program for the Project:

(Set forth the program, identify documentation in which the program is set forth, or state the manner in which the program will be developed.)

« »

#### § 1.1.2 The Owner's design requirements for the Project and related documentation:

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, 2 e-mail copyright@aia.org. User Notes:



(Identify below, or in an attached exhibit, the documentation that contains the Owner's design requirements, including any performance specifications for the Project.)

« »

### § 1.1.3 The Project's physical characteristics:

(Identify or describe, if appropriate, size, location, dimensions, or other pertinent information, such as geotechnical reports; site, boundary and topographic surveys; traffic and utility studies; availability of public and private utilities and services; legal description of the site; etc.)

« »

§ 1.1.4 The Owner's anticipated Sustainable Objective for the Project, if any:

(Identify the Owner's Sustainable Objective for the Project such as Sustainability Certification, benefit to the environment, enhancement to the health and well-being of building occupants, or improvement of energy efficiency. If the Owner identifies a Sustainable Objective, incorporate AIA Document A141<sup>™</sup>-2014, Exhibit C, Sustainable Projects, into this Agreement to define the terms, conditions and Work related to the Owner's Sustainable Objective.)

« »

§ 1.1.5 Incentive programs the Owner intends to pursue for the Project, including those related to the Sustainable Objective, and any deadlines for receiving the incentives that are dependent on, or related to, the Design-Builder's services, are as follows:

(Identify incentive programs the Owner intends to pursue for the Project and deadlines for submitting or applying for the incentive programs.)

« »

§ 1.1.6 The Owner's budget for the Work to be provided by the Design-Builder is set forth below: (Provide total for Owner's budget, and if known, a line item breakdown of costs.)

#### « »

§ 1.1.7 The Owner's design and construction milestone dates:

.1 Design phase milestone dates:

« »

.2 Submission of Design-Builder Proposal:

« »

.3 Phased completion dates:

« »

.4 Substantial Completion date:

« »

.5 Other milestone dates:

« »



§ 1.1.8 The Owner requires the Design-Builder to retain the following Architect, Consultants and Contractors at the Design-Builder's cost:

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 3 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes (2034726772)

(List name, legal status, address and other information.)



« »

§ 1.2.3 The Owner will retain the following consultants and separate contractors: (List discipline, scope of work, and, if known, identify by name and address.)

« »

§ 1.2.4 The Design-Builder identifies the following representative in accordance with Section 3.1.2:

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "ATA," the ATA Logo, and "ATA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by ATA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 4 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes: (2034726772)

(List name, address and other information.)

« » « »

« »

« »

« »

« »

§ 1.2.5 Neither the Owner's nor the Design-Builder's representative shall be changed without ten days' written notice to the other party.

### § 1.3 Binding Dispute Resolution

For any Claim subject to, but not resolved by, mediation pursuant to Section 14.3, the method of binding dispute resolution shall be the following:

(Check the appropriate box. If the Owner and Design-Builder do not select a method of binding dispute resolution below, or do not subsequently agree in writing to a binding dispute resolution other than litigation, Claims will be resolved by litigation in a court of competent jurisdiction.)

[ « » ] Arbitration pursuant to Section 14.4

Litigation in a court of competent jurisdiction

[« »] Other: (Specify)

« »

### § 1.4 Definitions

§ 1.4.1 Design-Build Documents. The Design-Build Documents consist of this Agreement between Owner and Design-Builder and its attached Exhibits (hereinafter, the "Agreement"); other documents listed in this Agreement; and Modifications issued after execution of this Agreement. A Modification is (1) a written amendment to the Contract signed by both parties, including the Design-Build Amendment, (2) a Change Order, or (3) a Change Directive.

§ 1.4.2 The Contract. The Design-Build Documents form the Contract. The Contract represents the entire and integrated agreement between the parties and supersedes prior negotiations, representations or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Design-Build Documents shall not be construed to create a contractual relationship of any kind between any persons or entities other than the Owner and the Design-Builder.

§ 1.4.3 The Work. The term "Work" means the design, construction and related services required to fulfill the Design-Builder's obligations under the Design-Build Documents, whether completed or partially completed, and includes all labor, materials, equipment and services provided or to be provided by the Design-Builder. The Work may constitute the whole or a part of the Project.

§ 1.4.4 The Project. The Project is the total design and construction of which the Work performed under the Design-Build Documents may be the whole or a part, and may include design and construction by the Owner and by separate contractors.

§ 1.4.5 Instruments of Service. Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Design-Builder, Contractor(s), Architect, and Consultant(s) under their respective agreements. Instruments of Service may include, without limitation, studies, surveys, models, sketches, drawings, specifications, digital models and other similar materials.

§ 1.4.6 Submittal. A Submittal is any submission to the Owner for review and approval demonstrating how the Design-Builder proposes to conform to the Design-Build Documents for those portions of the Work for which the Design-Build Documents require Submittals. Submittals include, but are not limited to, shop drawings, product data, and samples. Submittals are not Design-Build Documents unless incorporated into a Modification.

§ 1.4.7 Owner. The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Design-Build Documents as if singular in number. The term "Owner" means the Owner or the Owner's authorized representative.

**§ 1.4.8 Design-Builder.** The Design-Builder is the person or entity identified as such in the Agreement and is referred to throughout the Design-Build Documents as if singular in number. The term "Design-Builder" means the Design-Builder or the Design-Builder's authorized representative.

**§ 1.4.9 Consultant.** A Consultant is a person or entity providing professional services for the Design-Builder for all or a portion of the Work, and is referred to throughout the Design-Build Documents as if singular in number. To the extent required by the relevant jurisdiction, the Consultant shall be lawfully licensed to provide the required professional services.

**§ 1.4.10 Architect.** The Architect is a person or entity providing design services for the Design-Builder for all or a portion of the Work, and is lawfully licensed to practice architecture in the applicable jurisdiction. The Architect is referred to throughout the Design-Build Documents as if singular in number.

**§ 1.4.11 Contractor**. A Contractor is a person or entity performing all or a portion of the construction, required in connection with the Work, for the Design-Builder. The Contractor shall be lawfully licensed, if required in the jurisdiction where the Project is located. The Contractor is referred to throughout the Design-Build Documents as if singular in number and means a Contractor or an authorized representative of the Contractor.

**§ 1.4.12 Confidential Information.** Confidential Information is information containing confidential or business proprietary information that is clearly marked as "confidential."

**§ 1.4.13 Contract Time.** Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, as set forth in the Design-Build Amendment for Substantial Completion of the Work.

**§ 1.4.14 Day.** The term "day" as used in the Design-Build Documents shall mean calendar day unless otherwise specifically defined.

**§ 1.4.15 Contract Sum.** The Contract Sum is the amount to be paid to the Design-Builder for performance of the Work after execution of the Design-Build Amendment, as identified in Article A.1 of the Design-Build Amendment.

### ARTICLE 2 COMPENSATION AND PROGRESS PAYMENTS

§ 2.1 Compensation for Work Performed Prior To Execution of Design-Build Amendment

**§ 2.1.1** Unless otherwise agreed, payments for Work performed prior to Execution of the Design-Build Amendment shall be made monthly. For the Design-Builder's performance of Work prior to the execution of the Design-Build Amendment, the Owner shall compensate the Design-Builder as follows:

(Insert amount of, or basis for, compensation, including compensation for any Sustainability Services, or indicate the exhibit in which the information is provided. If there will be a limit on the total amount of compensation for Work performed prior to the execution of the Design-Build Amendment, state the amount of the limit.)

« »

**§ 2.1.2** The hourly billing rates for services of the Design-Builder and the Design-Builder's Architect, Consultants and Contractors, if any, are set forth below.

(If applicable, attach an exhibit of hourly billing rates or insert them below.)

« »

Individual or Position

Rate

AIA Document Al41" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No. 4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. (2034726772)

### § 2.1.3 Compensation for Reimbursable Expenses Prior To Execution of Design-Build Amendment

§ 2.1.3.1 Reimbursable Expenses are in addition to compensation set forth in Section 2.1.1 and 2.1.2 and include expenses, directly related to the Project, incurred by the Design-Builder and the Design-Builder's Architect, Consultants, and Contractors, as follows:

- Transportation and authorized out-of-town travel and subsistence; .1
- .2 Dedicated data and communication services, teleconferences, Project web sites, and extranets;
- .3 Fees paid for securing approval of authorities having jurisdiction over the Project;
- .4 Printing, reproductions, plots, standard form documents;
- .5 Postage, handling and delivery;
- .6 Expense of overtime work requiring higher than regular rates, if authorized in advance by the Owner;
- Renderings, physical models, mock-ups, professional photography, and presentation materials .7 requested by the Owner;
- .8 All taxes levied on professional services and on reimbursable expenses; and
- .9 Other Project-related expenditures, if authorized in advance by the Owner.

§ 2.1.3.2 For Reimbursable Expenses, the compensation shall be the expenses the Design-Builder and the Design-Builder's Architect, Consultants and Contractors incurred, plus an administrative fee of « » percent ( « » %) of the expenses incurred.

### § 2.1.4 Payments to the Design-Builder Prior To Execution of Design-Build Amendment

§ 2.1.4.1 Payments are due and payable upon presentation of the Design-Builder's invoice. Amounts unpaid « » ( « » ) days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Design-Builder. (Insert rate of monthly or annual interest agreed upon.)

#### « »% « »

§ 2.1.4.2 Records of Reimbursable Expenses and services performed on the basis of hourly rates shall be available to the Owner at mutually convenient times for a period of two years following execution of the Design-Build Amendment or termination of this Agreement, whichever occurs first.

#### § 2.2 Contract Sum and Payment for Work Performed After Execution of Design-Build Amendment

For the Design-Builder's performance of the Work after execution of the Design-Build Amendment, the Owner shall pay to the Design-Builder the Contract Sum in current funds as agreed in the Design-Build Amendment.

#### ARTICLE 3 GENERAL REQUIREMENTS OF THE WORK OF THE DESIGN-BUILD CONTRACT § 3.1 General

§ 3.1.1 The Design-Builder shall comply with any applicable licensing requirements in the jurisdiction where the Project is located.

§ 3.1.2 The Design-Builder shall designate in writing a representative who is authorized to act on the Design-Builder's behalf with respect to the Project.

§ 3.1.3 The Design-Builder shall perform the Work in accordance with the Design-Build Documents. The Design-Builder shall not be relieved of the obligation to perform the Work in accordance with the Design-Build Documents by the activities, tests, inspections or approvals of the Owner.

§ 3.1.3.1 The Design-Builder shall perform the Work in compliance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities. If the Design-Builder performs Work contrary to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities, the Design-Builder shall assume responsibility for such Work and shall bear the costs attributable to correction.

§ 3.1.3.2 Neither the Design-Builder nor any Contractor, Consultant, or Architect shall be obligated to perform any act which they believe will violate any applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities. If the Design-Builder determines that implementation of any instruction received from the Owner, including those in the Owner's Criteria, would cause a violation of any applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, the Design-Builder shall notify the Owner in

writing. Upon verification by the Owner that a change to the Owner's Criteria is required to remedy the violation, the Owner and the Design-Builder shall execute a Modification in accordance with Article 6.

§ 3.1.4 The Design-Builder shall be responsible to the Owner for acts and omissions of the Design-Builder's employees, Architect, Consultants, Contractors, and their agents and employees, and other persons or entities performing portions of the Work.

§ 3.1.5 General Consultation. The Design-Builder shall schedule and conduct periodic meetings with the Owner to review matters such as procedures, progress, coordination, and scheduling of the Work.

§ 3.1.6 When applicable law requires that services be performed by licensed professionals, the Design-Builder shall provide those services through qualified, licensed professionals. The Owner understands and agrees that the services of the Design-Builder's Architect and the Design-Builder's other Consultants are performed in the sole interest of, and for the exclusive benefit of, the Design-Builder.

§ 3.1.7 The Design-Builder, with the assistance of the Owner, shall prepare and file documents required to obtain necessary approvals of governmental authorities having jurisdiction over the Project.

### § 3.1.8 Progress Reports

§ 3.1.8.1 The Design-Builder shall keep the Owner informed of the progress and quality of the Work. On a monthly basis, or otherwise as agreed to by the Owner and Design-Builder, the Design-Builder shall submit written progress reports to the Owner, showing estimated percentages of completion and other information identified below:

- Work completed for the period; .1
- .2 Project schedule status;
- .3 Submittal schedule and status report, including a summary of outstanding Submittals;
- .4 Responses to requests for information to be provided by the Owner;
- .5 Approved Change Orders and Change Directives;
- .6 Pending Change Order and Change Directive status reports;
- .7 Tests and inspection reports;
- .8 Status report of Work rejected by the Owner;
- .9 Status of Claims previously submitted in accordance with Article 14;
- .10 Cumulative total of the Cost of the Work to date including the Design-Builder's compensation and Reimbursable Expenses, if any;
- .11 Current Project cash-flow and forecast reports; and
- .12 Additional information as agreed to by the Owner and Design-Builder.

§ 3.1.8.2 In addition, where the Contract Sum is the Cost of the Work with or without a Guaranteed Maximum Price, the Design-Builder shall include the following additional information in its progress reports:

- .1 Design-Builder's work force report;
- .2 Equipment utilization report; and
- .3 Cost summary, comparing actual costs to updated cost estimates.

## § 3.1.9 Design-Builder's Schedules

§ 3.1.9.1 The Design-Builder, promptly after execution of this Agreement, shall prepare and submit for the Owner's information a schedule for the Work. The schedule, including the time required for design and construction, shall not exceed time limits current under the Design-Build Documents, shall be revised at appropriate intervals as required by the conditions of the Work and Project, shall be related to the entire Project to the extent required by the Design-Build Documents, shall provide for expeditious and practicable execution of the Work, and shall include allowances for periods of time required for the Owner's review and for approval of submissions by authorities having jurisdiction over the Project.

§ 3.1.9.2 The Design-Builder shall perform the Work in general accordance with the most recent schedules submitted to the Owner.

§ 3.1.10 Certifications. Upon the Owner's written request, the Design-Builder shall obtain from the Architect, Consultants, and Contractors, and furnish to the Owner, certifications with respect to the documents and services provided by the Architect, Consultants, and Contractors (a) that, to the best of their knowledge, information and belief,

180 of 258
the documents or services to which the certifications relate (i) are consistent with the Design-Build Documents, except to the extent specifically identified in the certificate, and (ii) comply with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities governing the design of the Project; and (b) that the Owner and its consultants shall be entitled to rely upon the accuracy of the representations and statements contained in the certifications. The Design-Builder's Architect, Consultants, and Contractors shall not be required to execute certificates or consents that would require knowledge, services or responsibilities beyond the scope of their services.

#### § 3.1.11 Design-Builder's Submittals

§ 3.1.11.1 Prior to submission of any Submittals, the Design-Builder shall prepare a Submittal schedule, and shall submit the schedule for the Owner's approval. The Owner's approval shall not unreasonably be delayed or withheld. The Submittal schedule shall (1) be coordinated with the Design-Builder's schedule provided in Section 3.1.9.1, (2) allow the Owner reasonable time to review Submittals, and (3) be periodically updated to reflect the progress of the Work. If the Design-Builder fails to submit a Submittal schedule, the Design-Builder shall not be entitled to any increase in Contract Sum or extension of Contract Time based on the time required for review of Submittals.

§ 3.1.11.2 By providing Submittals the Design-Builder represents to the Owner that it has (1) reviewed and approved them, (2) determined and verified materials, field measurements and field construction criteria related thereto, or will do so and (3) checked and coordinated the information contained within such Submittals with the requirements of the Work and of the Design-Build Documents.

§ 3.1.11.3 The Design-Builder shall perform no portion of the Work for which the Design-Build Documents require Submittals until the Owner has approved the respective Submittal.

§ 3.1.11.4 The Work shall be in accordance with approved Submittals except that the Design-Builder shall not be relieved of its responsibility to perform the Work consistent with the requirements of the Design-Build Documents. The Work may deviate from the Design-Build Documents only if the Design-Builder has notified the Owner in writing of a deviation from the Design-Build Documents at the time of the Submittal and a Modification is executed authorizing the identified deviation. The Design-Builder shall not be relieved of responsibility for errors or omissions in Submittals by the Owner's approval of the Submittals.

§ 3.1.11.5 All professional design services or certifications to be provided by the Design-Builder, including all drawings, calculations, specifications, certifications, shop drawings and other Submittals, shall contain the signature and seal of the licensed design professional preparing them. Submittals related to the Work designed or certified by the licensed design professionals, if prepared by others, shall bear the licensed design professional's written approval. The Owner and its consultants shall be entitled to rely upon the adequacy, accuracy and completeness of the services, certifications or approvals performed by such design professionals.

§ 3.1.12 Warranty. The Design-Builder warrants to the Owner that materials and equipment furnished under the Contract will be of good quality and new unless the Design-Build Documents require or permit otherwise. The Design-Builder further warrants that the Work will conform to the requirements of the Design-Build Documents and will be free from defects, except for those inherent in the quality of the Work or otherwise expressly permitted by the Design-Build Documents. Work, materials, or equipment not conforming to these requirements may be considered defective. The Design-Builder's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Design-Builder, improper or insufficient maintenance, improper operation, or normal wear and tear and normal usage. If required by the Owner, the Design-Builder shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

# § 3.1.13 Royalties, Patents and Copyrights

§ 3.1.13.1 The Design-Builder shall pay all royalties and license fees.

§ 3.1.13.2 The Design-Builder shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner and its separate contractors and consultants harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process or product of a particular manufacturer or manufacturers is required by the Owner, or where the copyright violations are required in the Owner's Criteria. However, if the Design-Builder has reason to believe that the design, process or product required in the Owner's Criteria is an infringement of a copyright or a patent, the Design-Builder shall be responsible for such loss unless such information is promptly furnished to the Owner. If the Owner receives notice from a patent or copyright owner of an

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 9 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes (2034726772)

alleged violation of a patent or copyright, attributable to the Design-Builder, the Owner shall give prompt written notice to the Design-Builder.

#### § 3.1.14 Indemnification

§ 3.1.14.1 To the fullest extent permitted by law, the Design-Builder shall indemnify and hold harmless the Owner, including the Owner's agents and employees, from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work, but only to the extent caused by the negligent acts or omissions of the Design-Builder, Architect, a Consultant, a Contractor, or anyone directly or indirectly employed by them or anyone for whose acts they may be liable. Such obligation shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity that would otherwise exist as to a party or person described in this Section 3.1.14.

§ 3.1.14.2 The indemnification obligation under this Section 3.1.14 shall not be limited by a limitation on amount or type of damages, compensation, or benefits payable by or for Design-Builder, Architect, a Consultant, a Contractor, or anyone directly or indirectly employed by them, under workers' compensation acts, disability benefit acts or other employee benefit acts.

#### § 3.1.15 Contingent Assignment of Agreements

§ 3.1.15.1 Each agreement for a portion of the Work is assigned by the Design-Builder to the Owner, provided that

- assignment is effective only after termination of the Contract by the Owner for cause, pursuant to .1 Sections 13.1.4 or 13.2.2, and only for those agreements that the Owner accepts by written notification to the Design-Builder and the Architect, Consultants, and Contractors whose agreements are accepted for assignment; and
- .2 assignment is subject to the prior rights of the surety, if any, obligated under bond relating to the Contract.

When the Owner accepts the assignment of an agreement, the Owner assumes the Design-Builder's rights and obligations under the agreement.

§ 3.1.15.2 Upon such assignment, if the Work has been suspended for more than 30 days, the compensation under the assigned agreement shall be equitably adjusted for increases in cost resulting from the suspension.

§ 3.1.15.3 Upon such assignment to the Owner under this Section 3.1.15, the Owner may further assign the agreement to a successor design-builder or other entity. If the Owner assigns the agreement to a successor design-builder or other entity, the Owner shall nevertheless remain legally responsible for all of the successor design builder's or other entity's obligations under the agreement.

§ 3.1.16 Design-Builder's Insurance and Bonds. The Design-Builder shall purchase and maintain insurance and provide bonds as set forth in Exhibit B.

#### ARTICLE 4 WORK PRIOR TO EXECUTION OF THE DESIGN-BUILD AMENDMENT § 4.1 General

§ 4.1.1 Any information submitted by the Design-Builder, and any interim decisions made by the Owner, shall be for the purpose of facilitating the design process and shall not modify the Owner's Criteria unless the Owner and Design-Builder execute a Modification.

§ 4.1.2 The Design-Builder shall advise the Owner on proposed site use and improvements, selection of materials, and building systems and equipment. The Design-Builder shall also provide the Owner with recommendations, consistent with the Owner's Criteria, on constructability; availability of materials and labor; time requirements for procurement, installation and construction; and factors related to construction cost including, but not limited to, costs of alternative designs or materials, preliminary budgets, life-cycle data, and possible cost reductions.

#### § 4.2 Evaluation of the Owner's Criteria

§ 4.2.1 The Design-Builder shall schedule and conduct meetings with the Owner and any other necessary individuals or entities to discuss and review the Owner's Criteria as set forth in Section 1.1. The Design-Builder shall thereafter again meet with the Owner to discuss a preliminary evaluation of the Owner's Criteria. The preliminary evaluation shall address possible alternative approaches to design and construction of the Project and include the

182 of 258

Design-Builder's recommendations, if any, with regard to accelerated or fast-track scheduling, procurement, or phased construction. The preliminary evaluation shall consider cost information, constructability, and procurement and construction scheduling issues.

**§ 4.2.2** After the Design-Builder meets with the Owner and presents the preliminary evaluation, the Design-Builder shall provide a written report to the Owner, summarizing the Design-Builder's evaluation of the Owner's Criteria. The report shall also include

- .1 allocations of program functions, detailing each function and their square foot areas;
- .2 a preliminary estimate of the Cost of the Work, and, if necessary, recommendations to adjust the Owner's Criteria to conform to the Owner's budget;
- .3 a preliminary schedule, which shall include proposed design milestones; dates for receiving additional information from, or for work to be completed by, the Owner; anticipated date for the Design-Builder's Proposal; and dates of periodic design review sessions with the Owner; and
  .4 the following:
- the following: (List additional information, if any, to be included in the Design-Builder's written report.)

« »

**§ 4.2.3** The Owner shall review the Design-Builder's written report and, if acceptable, provide the Design-Builder with written consent to proceed to the development of the Preliminary Design as described in Section 4.3. The consent to proceed shall not be understood to modify the Owner's Criteria unless the Owner and Design-Builder execute a Modification.

#### § 4.3 Preliminary Design

**§ 4.3.1** Upon the Owner's issuance of a written consent to proceed under Section 4.2.3, the Design-Builder shall prepare and submit a Preliminary Design to the Owner. The Preliminary Design shall include a report identifying any deviations from the Owner's Criteria, and shall include the following:

- .1 Confirmation of the allocations of program functions;
- .2 Site plan;
- .3 Building plans, sections and elevations;
- .4 Structural system;
- .5 Selections of major building systems, including but not limited to mechanical, electrical and plumbing systems; and
- .6 Outline specifications or sufficient drawing notes describing construction materials.

The Preliminary Design may include some combination of physical study models, perspective sketches, or digital modeling.

**§ 4.3.2** The Owner shall review the Preliminary Design and, if acceptable, provide the Design-Builder with written consent to proceed to development of the Design-Builder's Proposal. The Preliminary Design shall not modify the Owner's Criteria unless the Owner and Design-Builder execute a Modification.

# § 4.4 Design-Builder's Proposal

§ 4.4.1 Upon the Owner's issuance of a written consent to proceed under Section 4.3.2, the Design-Builder shall prepare and submit the Design-Builder's Proposal to the Owner. The Design-Builder's Proposal shall include the following:

- .1 A list of the Preliminary Design documents and other information, including the Design-Builder's clarifications, assumptions and deviations from the Owner's Criteria, upon which the Design-Builder's Proposal is based;
- .2 The proposed Contract Sum, including the compensation method and, if based upon the Cost of the Work plus a fee, a written statement of estimated cost organized by trade categories, allowances, contingencies, Design-Builder's Fee, and other items that comprise the Contract Sum;
- .3 The proposed date the Design-Builder shall achieve Substantial Completion;
- .4 An enumeration of any qualifications and exclusions, if applicable;
- .5 A list of the Design-Builder's key personnel, Contractors and suppliers; and
- .6 The date on which the Design-Builder's Proposal expires.

§ 4.4.2 Submission of the Design-Builder's Proposal shall constitute a representation by the Design-Builder that it has visited the site and become familiar with local conditions under which the Work is to be completed.

§ 4.4.3 If the Owner and Design-Builder agree on a proposal, the Owner and Design-Builder shall execute the Design-Build Amendment setting forth the terms of their agreement.

#### ARTICLE 5 WORK FOLLOWING EXECUTION OF THE DESIGN-BUILD AMENDMENT

#### § 5.1 Construction Documents

§ 5.1.1 Upon the execution of the Design-Build Amendment, the Design-Builder shall prepare Construction Documents. The Construction Documents shall establish the quality levels of materials and systems required. The Construction Documents shall be consistent with the Design-Build Documents.

§ 5.1.2 The Design-Builder shall provide the Construction Documents to the Owner for the Owner's information. If the Owner discovers any deviations between the Construction Documents and the Design-Build Documents, the Owner shall promptly notify the Design-Builder of such deviations in writing. The Construction Documents shall not modify the Design-Build Documents unless the Owner and Design-Builder execute a Modification. The failure of the Owner to discover any such deviations shall not relieve the Design-Builder of the obligation to perform the Work in accordance with the Design-Build Documents.

#### § 5.2 Construction

§ 5.2.1 Commencement. Except as permitted in Section 5.2.2, construction shall not commence prior to execution of the Design-Build Amendment.

§ 5.2.2 If the Owner and Design-Builder agree in writing, construction may proceed prior to the execution of the Design-Build Amendment. However, such authorization shall not waive the Owner's right to reject the Design-Builder's Proposal.

§ 5.2.3 The Design-Builder shall supervise and direct the Work, using the Design-Builder's best skill and attention. The Design-Builder shall be solely responsible for, and have control over, construction means, methods, techniques, sequences and procedures, and for coordinating all portions of the Work under the Contract, unless the Design-Build Documents give other specific instructions concerning these matters.

§ 5.2.4 The Design-Builder shall be responsible for inspection of portions of Work already performed to determine that such portions are in proper condition to receive subsequent Work.

#### § 5.3 Labor and Materials

§ 5.3.1 Unless otherwise provided in the Design-Build Documents, the Design-Builder shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services, necessary for proper execution and completion of the Work, whether temporary or permanent, and whether or not incorporated or to be incorporated in the Work.

§ 5.3.2 When a material or system is specified in the Design-Build Documents, the Design-Builder may make substitutions only in accordance with Article 6.

§ 5.3.3 The Design-Builder shall enforce strict discipline and good order among the Design-Builder's employees and other persons carrying out the Work. The Design-Builder shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them.

# § 5.4 Taxes

The Design-Builder shall pay sales, consumer, use and similar taxes, for the Work provided by the Design-Builder, that are legally enacted when the Design-Build Amendment is executed, whether or not yet effective or merely scheduled to go into effect.

#### § 5.5 Permits, Fees, Notices and Compliance with Laws

§ 5.5.1 Unless otherwise provided in the Design-Build Documents, the Design-Builder shall secure and pay for the building permit as well as any other permits, fees, licenses, and inspections by government agencies, necessary for proper execution of the Work and Substantial Completion of the Project.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 12 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes: (2034726772)

§ 5.5.2 The Design-Builder shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities, applicable to performance of the Work.

§ 5.5.3 Concealed or Unknown Conditions. If the Design-Builder encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Design-Build Documents or (2) unknown physical conditions of an unusual nature that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Design-Build Documents, the Design-Builder shall promptly provide notice to the Owner before conditions are disturbed and in no event later than 21 days after first observance of the conditions. The Owner shall promptly investigate such conditions and, if the Owner determines that they differ materially and cause an increase or decrease in the Design-Builder's cost of, or time required for, performance of any part of the Work, shall recommend an equitable adjustment in the Contract Sum or Contract Time, or both. If the Owner determines that the conditions at the site are not materially different from those indicated in the Design-Build Documents and that no change in the terms of the Contract is justified, the Owner shall promptly notify the Design-Builder in writing, stating the reasons. If the Design-Builder disputes the Owner's determination or recommendation, the Design-Builder may proceed as provided in Article 14.

§ 5.5.4 If, in the course of the Work, the Design-Builder encounters human remains, or recognizes the existence of burial markers, archaeological sites, or wetlands, not indicated in the Design-Build Documents, the Design-Builder shall immediately suspend any operations that would affect them and shall notify the Owner. Upon receipt of such notice, the Owner shall promptly take any action necessary to obtain governmental authorization required to resume the operations. The Design-Builder shall continue to suspend such operations until otherwise instructed by the Owner but shall continue with all other operations that do not affect those remains or features. Requests for adjustments in the Contract Sum and Contract Time arising from the existence of such remains or features may be made as provided in Article 14.

#### § 5.6 Allowances

§ 5.6.1 The Design-Builder shall include in the Contract Sum all allowances stated in the Design-Build Documents. Items covered by allowances shall be supplied for such amounts, and by such persons or entities as the Owner may direct, but the Design-Builder shall not be required to employ persons or entities to whom the Design-Builder has reasonable objection.

§ 5.6.2 Unless otherwise provided in the Design-Build Documents,

- allowances shall cover the cost to the Design-Builder of materials and equipment delivered at the site .1 and all required taxes, less applicable trade discounts;
- .2 the Design-Builder's costs for unloading and handling at the site, labor, installation costs, overhead, profit, and other expenses contemplated for stated allowance amounts, shall be included in the Contract Sum but not in the allowances; and
- .3 whenever costs are more than or less than allowances, the Contract Sum shall be adjusted accordingly by Change Order. The amount of the Change Order shall reflect (1) the difference between actual costs and the allowances under Section 5.6.2.1 and (2) changes in Design-Builder's costs under Section 5.6.2.2.

§ 5.6.3 The Owner shall make selections of materials and equipment with reasonable promptness for allowances requiring Owner selection.

# § 5.7 Key Personnel, Contractors and Suppliers

§ 5.7.1 The Design-Builder shall not employ personnel, or contract with Contractors or suppliers to whom the Owner has made reasonable and timely objection. The Design-Builder shall not be required to contract with anyone to whom the Design-Builder has made reasonable and timely objection.

§ 5.7.2 If the Design-Builder changes any of the personnel, Contractors or suppliers identified in the Design-Build Amendment, the Design-Builder shall notify the Owner and provide the name and qualifications of the new personnel, Contractor or supplier. The Owner may reply within 14 days to the Design-Builder in writing, stating (1) whether the Owner has reasonable objection to the proposed personnel, Contractor or supplier or (2) that the Owner requires

additional time to review. Failure of the Owner to reply within the 14-day period shall constitute notice of no reasonable objection.

§ 5.7.3 Except for those persons or entities already identified or required in the Design-Build Amendment, the Design-Builder, as soon as practicable after execution of the Design-Build Amendment, shall furnish in writing to the Owner the names of persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for each principal portion of the Work. The Owner may reply within 14 days to the Design-Builder in writing stating (1) whether the Owner has reasonable objection to any such proposed person or entity or (2) that the Owner requires additional time for review. Failure of the Owner to reply within the 14-day period shall constitute notice of no reasonable objection.

§ 5.7.3.1 If the Owner has reasonable objection to a person or entity proposed by the Design-Builder, the Design-Builder shall propose another to whom the Owner has no reasonable objection. If the rejected person or entity was reasonably capable of performing the Work, the Contract Sum and Contract Time shall be increased or decreased by the difference, if any, occasioned by such change, and an appropriate Change Order shall be issued before commencement of the substitute person or entity's Work. However, no increase in the Contract Sum or Contract Time shall be allowed for such change unless the Design-Builder has acted promptly and responsively in submitting names as required.

#### § 5.8 Documents and Submittals at the Site

The Design-Builder shall maintain at the site for the Owner one copy of the Design-Build Documents and a current set of the Construction Documents, in good order and marked currently to indicate field changes and selections made during construction, and one copy of approved Submittals. The Design-Builder shall deliver these items to the Owner in accordance with Section 9.10.2 as a record of the Work as constructed.

#### § 5.9 Use of Site

The Design-Builder shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, lawful orders of public authorities, and the Design-Build Documents, and shall not unreasonably encumber the site with materials or equipment.

#### § 5.10 Cutting and Patching

The Design-Builder shall not cut, patch or otherwise alter fully or partially completed construction by the Owner or a separate contractor except with written consent of the Owner and of such separate contractor; such consent shall not be unreasonably withheld. The Design-Builder shall not unreasonably withhold from the Owner or a separate contractor the Design-Builder's consent to cutting or otherwise altering the Work.

# § 5.11 Cleaning Up

§ 5.11.1 The Design-Builder shall keep the premises and surrounding area free from accumulation of waste materials or rubbish caused by operations under the Contract. At completion of the Work, the Design-Builder shall remove waste materials, rubbish, the Design-Builder's tools, construction equipment, machinery and surplus materials from and about the Project.

§ 5.11.2 If the Design-Builder fails to clean up as provided in the Design-Build Documents, the Owner may do so and Owner shall be entitled to reimbursement from the Design-Builder.

#### § 5.12 Access to Work

The Design-Builder shall provide the Owner and its separate contractors and consultants access to the Work in preparation and progress wherever located. The Design-Builder shall notify the Owner regarding Project safety criteria and programs, which the Owner, and its contractors and consultants, shall comply with while at the site.

#### § 5.13 Construction by Owner or by Separate Contractors

#### § 5.13.1 Owner's Right to Perform Construction and to Award Separate Contracts

§ 5.13.1.1 The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces; and to award separate contracts in connection with other portions of the Project, or other construction or operations on the site, under terms and conditions identical or substantially similar to this Contract, including those terms and conditions related to insurance and waiver of subrogation. The Owner shall notify the Design-Builder

promptly after execution of any separate contract. If the Design-Builder claims that delay or additional cost is involved because of such action by the Owner, the Design-Builder shall make a Claim as provided in Article 14.

§ 5.13.1.2 When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "Design-Builder" in the Design-Build Documents in each case shall mean the individual or entity that executes each separate agreement with the Owner.

§ 5.13.1.3 The Owner shall provide for coordination of the activities of the Owner's own forces, and of each separate contractor, with the Work of the Design-Builder, who shall cooperate with them. The Design-Builder shall participate with other separate contractors and the Owner in reviewing their construction schedules. The Design-Builder shall make any revisions to the construction schedule deemed necessary after a joint review and mutual agreement. The construction schedules shall then constitute the schedules to be used by the Design-Builder, separate contractors and the Owner until subsequently revised.

§ 5.13.1.4 Unless otherwise provided in the Design-Build Documents, when the Owner performs construction or operations related to the Project with the Owner's own forces or separate contractors, the Owner shall be deemed to be subject to the same obligations, and to have the same rights, that apply to the Design-Builder under the Contract.

#### § 5.14 Mutual Responsibility

§ 5.14.1 The Design-Builder shall afford the Owner and separate contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Design-Builder's construction and operations with theirs as required by the Design-Build Documents.

§ 5.14.2 If part of the Design-Builder's Work depends upon construction or operations by the Owner or a separate contractor, the Design-Builder shall, prior to proceeding with that portion of the Work, prepare a written report to the Owner, identifying apparent discrepancies or defects in the construction or operations by the Owner or separate contractor that would render it unsuitable for proper execution and results of the Design-Builder's Work. Failure of the Design-Builder to report shall constitute an acknowledgment that the Owner's or separate contractor's completed or partially completed construction is fit and proper to receive the Design-Builder's Work, except as to defects not then reasonably discoverable.

§ 5.14.3 The Design-Builder shall reimburse the Owner for costs the Owner incurs that are payable to a separate contractor because of the Design-Builder's delays, improperly timed activities or defective construction. The Owner shall be responsible to the Design-Builder for costs the Design-Builder incurs because of a separate contractor's delays, improperly timed activities, damage to the Work or defective construction.

§ 5.14.4 The Design-Builder shall promptly remedy damage the Design-Builder wrongfully causes to completed or partially completed construction or to property of the Owner or separate contractors as provided in Section 10.2.5.

§ 5.14.5 The Owner and each separate contractor shall have the same responsibilities for cutting and patching the Work as the Design-Builder has with respect to the construction of the Owner or separate contractors in Section 5.10.

# § 5.15 Owner's Right to Clean Up

If a dispute arises among the Design-Builder, separate contractors and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up and will allocate the cost among those responsible.

# **ARTICLE 6 CHANGES IN THE WORK**

#### § 6.1 General

§ 6.1.1 Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order or Change Directive, subject to the limitations stated in this Article 6 and elsewhere in the Design-Build Documents.

§ 6.1.2 A Change Order shall be based upon agreement between the Owner and Design-Builder. The Owner may issue a Change Directive without agreement by the Design-Builder.

187 of 258

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," %ATA," the ATA Logo, and "ATA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by ATA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 15 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes

§ 6.1.3 Changes in the Work shall be performed under applicable provisions of the Design-Build Documents, and the Design-Builder shall proceed promptly, unless otherwise provided in the Change Order or Change Directive.

#### § 6.2 Change Orders

A Change Order is a written instrument signed by the Owner and Design-Builder stating their agreement upon all of the following:

- .1 The change in the Work;
- .2 The amount of the adjustment, if any, in the Contract Sum or, if prior to execution of the Design-Build Amendment, the adjustment in the Design-Builder's compensation; and
- .3 The extent of the adjustment, if any, in the Contract Time.

#### § 6.3 Change Directives

§ 6.3.1 A Change Directive is a written order signed by the Owner directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or, if prior to execution of the Design-Build Amendment, the adjustment in the Design-Builder's compensation, or Contract Time. The Owner may by Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, the Contract Sum or, if prior to execution of the Design-Build Amendment, the adjustment in the Design-Builder's compensation, and Contract Time being adjusted accordingly.

§ 6.3.2 A Change Directive shall be used in the absence of total agreement on the terms of a Change Order.

§ 6.3.3 If the Change Directive provides for an adjustment to the Contract Sum or, if prior to execution of the Design-Build Amendment, an adjustment in the Design-Builder's compensation, the adjustment shall be based on one of the following methods:

- .1 Mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation;
- .2 Unit prices stated in the Design-Build Documents or subsequently agreed upon;
- .3 Cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee; or
- .4 As provided in Section 6.3.7.

§ 6.3.4 If unit prices are stated in the Design-Build Documents or subsequently agreed upon, and if quantities originally contemplated are materially changed in a proposed Change Order or Change Directive so that application of such unit prices to quantities of Work proposed will cause substantial inequity to the Owner or Design-Builder, the applicable unit prices shall be equitably adjusted.

§ 6.3.5 Upon receipt of a Change Directive, the Design-Builder shall promptly proceed with the change in the Work involved and advise the Owner of the Design-Builder's agreement or disagreement with the method, if any, provided in the Change Directive for determining the proposed adjustment in the Contract Sum or, if prior to execution of the Design-Build Amendment, the adjustment in the Design-Builder's compensation, or Contract Time.

§ 6.3.6 A Change Directive signed by the Design-Builder indicates the Design-Builder's agreement therewith, including adjustment in Contract Sum or, if prior to execution of the Design-Build Amendment, the adjustment in the Design-Builder's compensation, and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.

§ 6.3.7 If the Design-Builder does not respond promptly or disagrees with the method for adjustment in the Contract Sum or, if prior to execution of the Design-Build Amendment, the method for adjustment in the Design-Builder's compensation, the Owner shall determine the method and the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase, an amount for overhead and profit as set forth in the Agreement, or if no such amount is set forth in the Agreement, a reasonable amount. In such case, and also under Section 6.3.3.3, the Design-Builder shall keep and present, in such form as the Owner may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Design-Build Documents, costs for the purposes of this Section 6.3.7 shall be limited to the following:

- Additional costs of professional services; .1
- .2 Costs of labor, including social security, unemployment insurance, fringe benefits required by agreement or custom, and workers' compensation insurance;

- .3 Costs of materials, supplies and equipment, including cost of transportation, whether incorporated or consumed;
- .4 Rental costs of machinery and equipment, exclusive of hand tools, whether rented from the Design-Builder or others;
- .5 Costs of premiums for all bonds and insurance, permit fees, and sales, use or similar taxes related to the Work: and
- .6 Additional costs of supervision and field office personnel directly attributable to the change.

§ 6.3.8 The amount of credit to be allowed by the Design-Builder to the Owner for a deletion or change that results in a net decrease in the Contract Sum or, if prior to execution of the Design-Build Amendment, in the Design-Builder's compensation, shall be actual net cost. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.

§ 6.3.9 Pending final determination of the total cost of a Change Directive to the Owner, the Design-Builder may request payment for Work completed under the Change Directive in Applications for Payment. The Owner will make an interim determination for purposes of certification for payment for those costs deemed to be reasonably justified. The Owner's interim determination of cost shall adjust the Contract Sum or, if prior to execution of the Design-Build Amendment, the Design-Builder's compensation, on the same basis as a Change Order, subject to the right of Design-Builder to disagree and assert a Claim in accordance with Article 14.

§ 6.3.10 When the Owner and Design-Builder agree with a determination concerning the adjustments in the Contract Sum or, if prior to execution of the Design-Build Amendment, the adjustment in the Design-Builder's compensation and Contract Time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and the Owner and Design-Builder shall execute a Change Order. Change Orders may be issued for all or any part of a Change Directive.

# **ARTICLE 7 OWNER'S RESPONSIBILITIES**

#### § 7.1 General

§ 7.1.1 The Owner shall designate in writing a representative who shall have express authority to bind the Owner with respect to all Project matters requiring the Owner's approval or authorization.

§ 7.1.2 The Owner shall render decisions in a timely manner and in accordance with the Design-Builder's schedule agreed to by the Owner. The Owner shall furnish to the Design-Builder, within 15 days after receipt of a written request, information necessary and relevant for the Design-Builder to evaluate, give notice of or enforce mechanic's lien rights. Such information shall include a correct statement of the record legal title to the property on which the Project is located, usually referred to as the site, and the Owner's interest therein.

#### § 7.2 Information and Services Required of the Owner

§ 7.2.1 The Owner shall furnish information or services required of the Owner by the Design-Build Documents with reasonable promptness.

§ 7.2.2 The Owner shall provide, to the extent under the Owner's control and if not required by the Design-Build Documents to be provided by the Design-Builder, the results and reports of prior tests, inspections or investigations conducted for the Project involving structural or mechanical systems; chemical, air and water pollution; hazardous materials; or environmental and subsurface conditions and information regarding the presence of pollutants at the Project site. Upon receipt of a written request from the Design-Builder, the Owner shall also provide surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site under the Owner's control.

§ 7.2.3 The Owner shall promptly obtain easements, zoning variances, and legal authorizations or entitlements regarding site utilization where essential to the execution of the Project.

§ 7.2.4 The Owner shall cooperate with the Design-Builder in securing building and other permits, licenses and inspections.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, 17 e-mail copyright@aia.org. User Notes

§ 7.2.5 The services, information, surveys and reports required to be provided by the Owner under this Agreement, shall be furnished at the Owner's expense, and except as otherwise specifically provided in this Agreement or elsewhere in the Design-Build Documents or to the extent the Owner advises the Design-Builder to the contrary in writing, the Design-Builder shall be entitled to rely upon the accuracy and completeness thereof. In no event shall the Design-Builder be relieved of its responsibility to exercise proper precautions relating to the safe performance of the Work.

§ 7.2.6 If the Owner observes or otherwise becomes aware of a fault or defect in the Work or non-conformity with the Design-Build Documents, the Owner shall give prompt written notice thereof to the Design-Builder.

§ 7.2.7 Prior to the execution of the Design-Build Amendment, the Design-Builder may request in writing that the Owner provide reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Design-Build Documents and the Design-Builder's Proposal. Thereafter, the Design-Builder may only request such evidence if (1) the Owner fails to make payments to the Design-Builder as the Design-Build Documents require; (2) a change in the Work materially changes the Contract Sum; or (3) the Design-Builder identifies in writing a reasonable concern regarding the Owner's ability to make payment when due. The Owner shall furnish such evidence as a condition precedent to commencement or continuation of the Work or the portion of the Work affected by a material change. After the Owner furnishes the evidence, the Owner shall not materially vary such financial arrangements without prior notice to the Design-Builder.

§ 7.2.8 Except as otherwise provided in the Design-Build Documents or when direct communications have been specially authorized, the Owner shall communicate through the Design-Builder with persons or entities employed or retained by the Design-Builder.

§ 7.2.9 Unless required by the Design-Build Documents to be provided by the Design-Builder, the Owner shall, upon request from the Design-Builder, furnish the services of geotechnical engineers or other consultants for investigation of subsurface, air and water conditions when such services are reasonably necessary to properly carry out the design services furnished by the Design-Builder. In such event, the Design-Builder shall specify the services required. Such services may include, but are not limited to, test borings, test pits, determinations of soil bearing values, percolation tests, evaluations of hazardous materials, ground corrosion and resistivity tests, and necessary operations for anticipating subsoil conditions. The services of geotechnical engineer(s) or other consultants shall include preparation and submission of all appropriate reports and professional recommendations.

§ 7.2.10 The Owner shall purchase and maintain insurance as set forth in Exhibit B.

# § 7.3 Submittals

§ 7.3.1 The Owner shall review and approve or take other appropriate action on Submittals. Review of Submittals is not conducted for the purpose of determining the accuracy and completeness of other details, such as dimensions and quantities; or for substantiating instructions for installation or performance of equipment or systems; or for determining that the Submittals are in conformance with the Design-Build Documents, all of which remain the responsibility of the Design-Builder as required by the Design-Build Documents. The Owner's action will be taken in accordance with the submittal schedule approved by the Owner or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time in the Owner's judgment to permit adequate review. The Owner's review of Submittals shall not relieve the Design-Builder of the obligations under Sections 3.1.11, 3.1.12, and 5.2.3. The Owner's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Owner, of any construction means, methods, techniques, sequences or procedures. The Owner's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

§ 7.3.2 Upon review of the Submittals required by the Design-Build Documents, the Owner shall notify the Design-Builder of any non-conformance with the Design-Build Documents the Owner discovers.

§ 7.4 Visits to the site by the Owner shall not be construed to create an obligation on the part of the Owner to make on-site inspections to check the quality or quantity of the Work. The Owner shall neither have control over or charge of, nor be responsible for, the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, because these are solely the Design-Builder's rights and responsibilities under the Design-Build Documents.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 18 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes

§ 7.5 The Owner shall not be responsible for the Design-Builder's failure to perform the Work in accordance with the requirements of the Design-Build Documents. The Owner shall not have control over or charge of, and will not be responsible for acts or omissions of the Design-Builder, Architect, Consultants, Contractors, or their agents or employees, or any other persons or entities performing portions of the Work for the Design-Builder.

§ 7.6 The Owner has the authority to reject Work that does not conform to the Design-Build Documents. The Owner shall have authority to require inspection or testing of the Work in accordance with Section 15.5.2, whether or not such Work is fabricated, installed or completed. However, neither this authority of the Owner nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Owner to the Design-Builder, the Architect, Consultants, Contractors, material and equipment suppliers, their agents or employees, or other persons or entities performing portions of the Work.

§ 7.7 The Owner shall determine the date or dates of Substantial Completion in accordance with Section 9.8 and the date of final completion in accordance with Section 9.10.

#### § 7.8 Owner's Right to Stop Work

If the Design-Builder fails to correct Work which is not in accordance with the requirements of the Design-Build Documents as required by Section 11.2 or persistently fails to carry out Work in accordance with the Design-Build Documents, the Owner may issue a written order to the Design-Builder to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Design-Builder or any other person or entity, except to the extent required by Section 5.13.1.3.

#### § 7.9 Owner's Right to Carry Out the Work

If the Design-Builder defaults or neglects to carry out the Work in accordance with the Design-Build Documents and fails within a ten-day period after receipt of written notice from the Owner to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to other remedies the Owner may have, correct such deficiencies. In such case, an appropriate Change Order shall be issued deducting from payments then or thereafter due the Design-Builder the reasonable cost of correcting such deficiencies. If payments then or thereafter due the Design-Builder are not sufficient to cover such amounts, the Design-Builder shall pay the difference to the Owner.

#### ARTICLE 8 TIME

#### § 8.1 Progress and Completion

§ 8.1.1 Time limits stated in the Design-Build Documents are of the essence of the Contract. By executing the Design-Build Amendment the Design-Builder confirms that the Contract Time is a reasonable period for performing the Work.

§ 8.1.2 The Design-Builder shall not, except by agreement of the Owner in writing, commence the Work prior to the effective date of insurance, other than property insurance, required by this Contract. The Contract Time shall not be adjusted as a result of the Design-Builder's failure to obtain insurance required under this Contract

§ 8.1.3 The Design-Builder shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time.

# § 8.2 Delays and Extensions of Time

§ 8.2.1 If the Design-Builder is delayed at any time in the commencement or progress of the Work by an act or neglect of the Owner or of a consultant or separate contractor employed by the Owner; or by changes ordered in the Work by the Owner; or by labor disputes, fire, unusual delay in deliveries, unavoidable casualties or other causes beyond the Design-Builder's control; or by delay authorized by the Owner pending mediation and binding dispute resolution or by other causes that the Owner determines may justify delay, then the Contract Time shall be extended by Change Order for such reasonable time as the Owner may determine.

§ 8.2.2 Claims relating to time shall be made in accordance with applicable provisions of Article 14.

§ 8.2.3 This Section 8.2 does not preclude recovery of damages for delay by either party under other provisions of the Design-Build Documents.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," %ATA," the ATA Logo, and "ATA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by ATA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the ATA Contract Documents® Terms of Service. To report copyright violations, 19 e-mail copyright@aia.org. User Notes: (2034726772)

#### ARTICLE 9 PAYMENT APPLICATIONS AND PROJECT COMPLETION

#### § 9.1 Contract Sum

The Contract Sum is stated in the Design-Build Amendment.

#### § 9.2 Schedule of Values

Where the Contract Sum is based on a stipulated sum or Guaranteed Maximum Price, the Design-Builder, prior to the first Application for Payment after execution of the Design-Build Amendment shall submit to the Owner a schedule of values allocating the entire Contract Sum to the various portions of the Work and prepared in such form and supported by such data to substantiate its accuracy as the Owner may require. This schedule, unless objected to by the Owner, shall be used as a basis for reviewing the Design-Builder's Applications for Payment.

#### § 9.3 Applications for Payment

§ 9.3.1 At least ten days before the date established for each progress payment, the Design-Builder shall submit to the Owner an itemized Application for Payment for completed portions of the Work. The application shall be notarized, if required, and supported by data substantiating the Design-Builder's right to payment as the Owner may require, such as copies of requisitions from the Architect, Consultants, Contractors, and material suppliers, and shall reflect retainage if provided for in the Design-Build Documents.

§ 9.3.1.1 As provided in Section 6.3.9, Applications for Payment may include requests for payment on account of changes in the Work that have been properly authorized by Change Directives, or by interim determinations of the Owner, but not yet included in Change Orders.

§ 9.3.1.2 Applications for Payment shall not include requests for payment for portions of the Work for which the Design-Builder does not intend to pay the Architect, Consultant, Contractor, material supplier, or other persons or entities providing services or work for the Design-Builder, unless such Work has been performed by others whom the Design-Builder intends to pay.

§ 9.3.2 Unless otherwise provided in the Design-Build Documents, payments shall be made for services provided as well as materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Design-Builder with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include the costs of applicable insurance, storage and transportation to the site for such materials and equipment stored off the site.

§ 9.3.3 The Design-Builder warrants that title to all Work, other than Instruments of Service, covered by an Application for Payment will pass to the Owner no later than the time of payment. The Design-Builder further warrants that, upon submittal of an Application for Payment, all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Design-Builder's knowledge, information and belief, be free and clear of liens, claims, security interests or encumbrances in favor of the Design-Builder, Architect, Consultants, Contractors, material suppliers, or other persons or entities entitled to make a claim by reason of having provided labor, materials and equipment relating to the Work.

#### § 9.4 Certificates for Payment

The Owner shall, within seven days after receipt of the Design-Builder's Application for Payment, issue to the Design-Builder a Certificate for Payment indicating the amount the Owner determines is properly due, and notify the Design-Builder in writing of the Owner's reasons for withholding certification in whole or in part as provided in Section 9.5.1.

#### § 9.5 Decisions to Withhold Certification

§ 9.5.1 The Owner may withhold a Certificate for Payment in whole or in part to the extent reasonably necessary to protect the Owner due to the Owner's determination that the Work has not progressed to the point indicated in the Design-Builder's Application for Payment, or the quality of the Work is not in accordance with the Design-Build Documents. If the Owner is unable to certify payment in the amount of the Application, the Owner will notify the Design-Builder as provided in Section 9.4. If the Design-Builder and Owner cannot agree on a revised amount, the Owner will promptly issue a Certificate for Payment for the amount that the Owner deems to be due and owing. The Owner may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued to such extent as may be necessary to protect the Owner from loss for which the Design-Builder is responsible because of

- defective Work, including design and construction, not remedied; .1
- .2 third party claims filed or reasonable evidence indicating probable filing of such claims unless security acceptable to the Owner is provided by the Design-Builder;
- .3 failure of the Design-Builder to make payments properly to the Architect, Consultants, Contractors or others, for services, labor, materials or equipment;
- .4 reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum;
- .5 damage to the Owner or a separate contractor;
- reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid .6 balance would not be adequate to cover actual or liquidated damages for the anticipated delay; or
- .7 repeated failure to carry out the Work in accordance with the Design-Build Documents.

§ 9.5.2 When the above reasons for withholding certification are removed, certification will be made for amounts previously withheld.

§ 9.5.3 If the Owner withholds certification for payment under Section 9.5.1.3, the Owner may, at its sole option, issue joint checks to the Design-Builder and to the Architect or any Consultants, Contractor, material or equipment suppliers, or other persons or entities providing services or work for the Design-Builder to whom the Design-Builder failed to make payment for Work properly performed or material or equipment suitably delivered.

#### § 9.6 Progress Payments

§ 9.6.1 After the Owner has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Design-Build Documents.

§ 9.6.2 The Design-Builder shall pay each Architect, Consultant, Contractor, and other person or entity providing services or work for the Design-Builder no later than the time period required by applicable law, but in no event more than seven days after receipt of payment from the Owner the amount to which the Architect, Consultant, Contractor, and other person or entity providing services or work for the Design-Builder is entitled, reflecting percentages actually retained from payments to the Design-Builder on account of the portion of the Work performed by the Architect, Consultant, Contractor, or other person or entity. The Design-Builder shall, by appropriate agreement with each Architect, Consultant, Contractor, and other person or entity providing services or work for the Design-Builder, require each Architect, Consultant, Contractor, and other person or entity providing services or work for the Design-Builder to make payments to subconsultants and subcontractors in a similar manner.

§ 9.6.3 The Owner will, on request and if practicable, furnish to the Architect, a Consultant, Contractor, or other person or entity providing services or work for the Design-Builder, information regarding percentages of completion or amounts applied for by the Design-Builder and action taken thereon by the Owner on account of portions of the Work done by such Architect, Consultant, Contractor or other person or entity providing services or work for the Design-Builder.

§ 9.6.4 The Owner has the right to request written evidence from the Design-Builder that the Design-Builder has properly paid the Architect, Consultants, Contractors, or other person or entity providing services or work for the Design-Builder, amounts paid by the Owner to the Design-Builder for the Work. If the Design-Builder fails to furnish such evidence within seven days, the Owner shall have the right to contact the Architect, Consultants, and Contractors to ascertain whether they have been properly paid. The Owner shall have no obligation to pay or to see to the payment of money to a Consultant or Contractor, except as may otherwise be required by law.

§ 9.6.5 Design-Builder payments to material and equipment suppliers shall be treated in a manner similar to that provided in Sections 9.6.2, 9.6.3 and 9.6.4.

§ 9.6.6 A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work not in accordance with the Design-Build Documents.

§ 9.6.7 Unless the Design-Builder provides the Owner with a payment bond in the full penal sum of the Contract Sum, payments received by the Design-Builder for Work properly performed by the Architect, Consultants, Contractors and other person or entity providing services or work for the Design-Builder, shall be held by the Design-Builder for the Architect and those Consultants, Contractors, or other person or entity providing services or work for the Design-Builder, for which payment was made by the Owner. Nothing contained herein shall require money to be placed in a separate account and not commingled with money of the Design-Builder, shall create any fiduciary liability or tort liability on the part of the Design-Builder for breach of trust or shall entitle any person or entity to an award of punitive damages against the Design-Builder for breach of the requirements of this provision.

#### § 9.7 Failure of Payment

If the Owner does not issue a Certificate for Payment, through no fault of the Design-Builder, within the time required by the Design-Build Documents, then the Design-Builder may, upon seven additional days' written notice to the Owner, stop the Work until payment of the amount owing has been received. The Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Design-Builder's reasonable costs of shut-down, delay and start-up, plus interest as provided for in the Design-Build Documents.

#### § 9.8 Substantial Completion

§ 9.8.1 Substantial Completion is the stage in the progress of the Work when the Work or designated portion thereof is sufficiently complete in accordance with the Design-Build Documents so that the Owner can occupy or utilize the Work for its intended use. The date of Substantial Completion is the date certified by the Owner in accordance with this Section 9.8.

**§ 9.8.2** When the Design-Builder considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Design-Builder shall prepare and submit to the Owner a comprehensive list of items to be completed or corrected prior to final payment. Failure to include an item on such list does not alter the responsibility of the Design-Builder to complete all Work in accordance with the Design-Build Documents.

**§ 9.8.3** Upon receipt of the Design-Builder's list, the Owner shall make an inspection to determine whether the Work or designated portion thereof is substantially complete. If the Owner's inspection discloses any item, whether or not included on the Design-Builder's list, which is not sufficiently complete in accordance with the Design-Build Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Design-Builder shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Owner. In such case, the Design-Builder shall then submit a request for another inspection by the Owner to determine Substantial Completion.

**§ 9.8.4** Prior to issuance of the Certificate of Substantial Completion under Section 9.8.5, the Owner and Design-Builder shall discuss and then determine the parties' obligations to obtain and maintain property insurance following issuance of the Certificate of Substantial Completion.

**§ 9.8.5** When the Work or designated portion thereof is substantially complete, the Design-Builder will prepare for the Owner's signature a Certificate of Substantial Completion that shall, upon the Owner's signature, establish the date of Substantial Completion; establish responsibilities of the Owner and Design-Builder for security, maintenance, heat, utilities, damage to the Work and insurance; and fix the time within which the Design-Builder shall finish all items on the list accompanying the Certificate. Warranties required by the Design-Build Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion.

**§ 9.8.6** The Certificate of Substantial Completion shall be submitted by the Design-Builder to the Owner for written acceptance of responsibilities assigned to it in the Certificate. Upon the Owner's acceptance, and consent of surety, if any, the Owner shall make payment of retainage applying to the Work or designated portion thereof. Payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Design-Build Documents.

# § 9.9 Partial Occupancy or Use

**§ 9.9.1** The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Design-Builder, provided such occupancy or use is consented to, by endorsement or otherwise, by the insurer providing property insurance and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Design-Builder have accepted in writing the responsibilities assigned to each of them for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work and

insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Design-Build Documents. When the Design-Builder considers a portion substantially complete, the Design-Builder shall prepare and submit a list to the Owner as provided under Section 9.8.2. Consent of the Design-Builder to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Design-Builder.

§ 9.9.2 Immediately prior to such partial occupancy or use, the Owner and Design-Builder shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.

§ 9.9.3 Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Design-Build Documents.

#### § 9.10 Final Completion and Final Payment

§ 9.10.1 Upon receipt of the Design-Builder's written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Owner will promptly make such inspection. When the Owner finds the Work acceptable under the Design-Build Documents and the Contract fully performed, the Owner will, subject to Section 9.10.2, promptly issue a final Certificate for Payment.

§ 9.10.2 Neither final payment nor any remaining retained percentage shall become due until the Design-Builder submits to the Owner (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work, for which the Owner or the Owner's property might be responsible or encumbered, (less amounts withheld by Owner) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Design-Build Documents to remain in force after final payment is currently in effect, (3) a written statement that the Design-Builder knows of no substantial reason that the insurance will not be renewable to cover the period required by the Design-Build Documents, (4) consent of surety, if any, to final payment, (5) as-constructed record copy of the Construction Documents marked to indicate field changes and selections made during construction, (6) manufacturer's warranties, product data, and maintenance and operations manuals, and (7) if required by the Owner. other data establishing payment or satisfaction of obligations, such as receipts, or releases and waivers of liens, claims security interests, or encumbrances, arising out of the Contract, to the extent and in such form as may be designated by the Owner. If an Architect, a Consultant, or a Contractor, or other person or entity providing services or work for the Design-Builder, refuses to furnish a release or waiver required by the Owner, the Design-Builder may furnish a bond satisfactory to the Owner to indemnify the Owner against such liens, claims, security interests, or encumbrances. If such liens, claims, security interests, or encumbrances remains unsatisfied after payments are made, the Design-Builder shall refund to the Owner all money that the Owner may be compelled to pay in discharging such liens, claims, security interests, or encumbrances, including all costs and reasonable attorneys' fees.

§ 9.10.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Design-Builder or by issuance of Change Orders affecting final completion, the Owner shall, upon application by the Design-Builder, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance for Work not fully completed or corrected is less than retainage stipulated in the Design-Build Documents, and if bonds have been furnished, the written consent of surety to payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Design-Builder to the Owner prior to issuance of payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of claims.

§ 9.10.4 The making of final payment shall constitute a waiver of Claims by the Owner except those arising from

- .1 liens, Claims, security interests or encumbrances arising out of the Contract and unsettled;
- .2 failure of the Work to comply with the requirements of the Design-Build Documents; or
- .3 terms of special warranties required by the Design-Build Documents.

§ 9.10.5 Acceptance of final payment by the Design-Builder shall constitute a waiver of claims by the Design-Builder except those previously made in writing and identified by the Design-Builder as unsettled at the time of final Application for Payment.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 23 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes

#### ARTICLE 10 PROTECTION OF PERSONS AND PROPERTY

#### § 10.1 Safety Precautions and Programs

The Design-Builder shall be responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the performance of the Contract.

#### § 10.2 Safety of Persons and Property

§ 10.2.1 The Design-Builder shall be responsible for precautions for the safety of, and reasonable protection to prevent damage, injury or loss to

- .1 employees on the Work and other persons who may be affected thereby;
- .2 the Work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody or control of the Design-Builder or the Architect, Consultants, or Contractors, or other person or entity providing services or work for the Design-Builder; and
- .3 other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, or structures and utilities not designated for removal, relocation or replacement in the course of construction.

§ 10.2.2 The Design-Builder shall comply with, and give notices required by, applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities, bearing on safety of persons or property, or their protection from damage, injury or loss.

§ 10.2.3 The Design-Builder shall implement, erect, and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations, and notify owners and users of adjacent sites and utilities of the safeguards and protections.

§ 10.2.4 When use or storage of explosives or other hazardous materials or equipment, or unusual methods, are necessary for execution of the Work, the Design-Builder shall exercise utmost care, and carry on such activities under supervision of properly qualified personnel.

§ 10.2.5 The Design-Builder shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Design-Build Documents) to property referred to in Sections 10.2.1.2 and 10.2.1.3, caused in whole or in part by the Design-Builder, the Architect, a Consultant, a Contractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Design-Builder is responsible under Sections 10.2.1.2 and 10.2.1.3; except damage or loss attributable to acts or omissions of the Owner, or anyone directly or indirectly employed by the Owner, or by anyone for whose acts the Owner may be liable, and not attributable to the fault or negligence of the Design-Builder. The foregoing obligations of the Design-Builder are in addition to the Design-Builder's obligations under Section 3.1.14.

§ 10.2.6 The Design-Builder shall designate a responsible member of the Design-Builder's organization, at the site, whose duty shall be the prevention of accidents. This person shall be the Design-Builder's superintendent unless otherwise designated by the Design-Builder in writing to the Owner.

§ 10.2.7 The Design-Builder shall not permit any part of the construction or site to be loaded so as to cause damage or create an unsafe condition.

§ 10.2.8 Injury or Damage to Person or Property. If the Owner or Design-Builder suffers injury or damage to person or property because of an act or omission of the other, or of others for whose acts such party is legally responsible, written notice of the injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding 21 days after discovery. The notice shall provide sufficient detail to enable the other party to investigate the matter.

#### § 10.3 Hazardous Materials

§ 10.3.1 The Design-Builder is responsible for compliance with any requirements included in the Design-Build Documents regarding hazardous materials. If the Design-Builder encounters a hazardous material or substance not addressed in the Design-Build Documents and if reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated biphenyl (PCB), encountered on the site by the Design-Builder, the Design-Builder shall, upon

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 24 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes (2034726772)

recognizing the condition, immediately stop Work in the affected area and report the condition to the Owner in writing.

§ 10.3.2 Upon receipt of the Design-Builder's written notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Design-Builder and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Design-Build Documents, the Owner shall furnish in writing to the Design-Builder the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of such material or substance or who are to perform the task of removal or safe containment of such material or substance. The Design Builder will promptly reply to the Owner in writing stating whether or not the Design-Builder has reasonable objection to the persons or entities proposed by the Owner. If the Design-Builder has an objection to a person or entity proposed by the Owner, the Owner shall propose another to whom the Design-Builder has no reasonable objection. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Design-Builder. By Change Order, the Contract Time shall be extended appropriately and the Contract Sum shall be increased in the amount of the Design-Builder's reasonable additional costs of shut-down, delay and start-up.

§ 10.3.3 To the fullest extent permitted by law, the Owner shall indemnify and hold harmless the Design-Builder, the Architect, Consultants, and Contractors, and employees of any of them, from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work in the affected area, if in fact the material or substance presents the risk of bodily injury or death as described in Section 10.3.1 and has not been rendered harmless, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease or death, or to injury to, or destruction of, tangible property (other than the Work itself), except to the extent that such damage, loss or expense is due to the fault or negligence of the party seeking. indemnity.

§ 10.3.4 The Owner shall not be responsible under this Section 10.3 for materials or substances the Design-Builder brings to the site unless such materials or substances are required by the Owner's Criteria. The Owner shall be responsible for materials or substances required by the Owner's Criteria, except to the extent of the Design-Builder' fault or negligence in the use and handling of such materials or substances.

§ 10.3.5 The Design-Builder shall indemnify the Owner for the cost and expense the Owner incurs (1) for remediation of a material or substance the Design-Builder brings to the site and negligently handles, or (2) where the Design-Builder fails to perform its obligations under Section 10.3.1, except to the extent that the cost and expense are due to the Owner's fault or negligence.

§ 10.3.6 If, without negligence on the part of the Design-Builder, the Design-Builder is held liable by a government agency for the cost of remediation of a hazardous material or substance solely by reason of performing Work as required by the Design-Build Documents, the Owner shall indemnify the Design-Builder for all cost and expense thereby incurred.

#### § 10.4 Emergencies

In an emergency affecting safety of persons or property, the Design-Builder shall act, at the Design-Builder's discretion, to prevent threatened damage, injury or loss.

# ARTICLE 11 UNCOVERING AND CORRECTION OF WORK

#### § 11.1 Uncovering of Work

The Owner may request to examine a portion of the Work that the Design-Builder has covered to determine if the Work has been performed in accordance with the Design-Build Documents. If such Work is in accordance with the Design-Build Documents, the Owner and Design-Builder shall execute a Change Order to adjust the Contract Time and Contract Sum, as appropriate. If such Work is not in accordance with the Design-Build Documents, the costs of uncovering and correcting the Work shall be at the Design-Builder's expense and the Design-Builder shall not be entitled to a change in the Contract Time unless the condition was caused by the Owner or a separate contractor in which event the Owner shall be responsible for payment of such costs and the Contract Time will be adjusted as appropriate.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 25 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes

# § 11.2 Correction of Work

**§ 11.2.1 Before or After Substantial Completion.** The Design-Builder shall promptly correct Work rejected by the Owner or failing to conform to the requirements of the Design-Build Documents, whether discovered before or after Substantial Completion and whether or not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for any design consultant employed by the Owner whose expenses and compensation were made necessary thereby, shall be at the Design-Builder's expense.

#### § 11.2.2 After Substantial Completion

**§ 11.2.2.1** In addition to the Design-Builder's obligations under Section 3.1.12, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of an applicable special warranty required by the Design-Build Documents, any of the Work is found not to be in accordance with the requirements of the Design-Build Documents, the Design-Builder shall correct it promptly after receipt of written notice from the Owner to do so unless the Owner has previously given the Design-Builder a written acceptance of such condition. The Owner shall give such notice promptly after discovery of the condition. During the one-year period for correction of the Work, if the Owner fails to notify the Design-Builder and give the Design-Builder and to make a claim for breach of warranty. If the Design-Builder fails to correct nonconforming Work within a reasonable time during that period after receipt of notice from the Owner, the Owner may correct it in accordance with Section 7.9.

**§ 11.2.2.2** The one-year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual completion of that portion of the Work.

**§ 11.2.2.3** The one-year period for correction of Work shall not be extended by corrective Work performed by the Design-Builder pursuant to this Section 11.2.

**§ 11.2.3** The Design-Builder shall remove from the site portions of the Work that are not in accordance with the requirements of the Design-Build Documents and are neither corrected by the Design-Builder nor accepted by the Owner.

§ 11.2.4 The Design-Builder shall bear the cost of correcting destroyed or damaged construction of the Owner or separate contractors, whether completed or partially completed, caused by the Design-Builder's correction or removal of Work that is not in accordance with the requirements of the Design-Build Documents.

**§ 11.2.5** Nothing contained in this Section 11.2 shall be construed to establish a period of limitation with respect to other obligations the Design-Builder has under the Design-Build Documents. Establishment of the one-year period for correction of Work as described in Section 11.2.2 relates only to the specific obligation of the Design-Builder to correct the Work, and has no relationship to the time within which the obligation to comply with the Design-Build Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Design-Builder's liability with respect to the Design-Builder's obligations other than specifically to correct the Work.

#### § 11.3 Acceptance of Nonconforming Work

If the Owner prefers to accept Work that is not in accordance with the requirements of the Design-Build Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

#### ARTICLE 12 COPYRIGHTS AND LICENSES

**§ 12.1** Drawings, specifications, and other documents furnished by the Design-Builder, including those in electronic form, are Instruments of Service. The Design-Builder, and the Architect, Consultants, Contractors, and any other person or entity providing services or work for any of them, shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and shall retain all common law, statutory and other reserved rights, including copyrights. Submission or distribution of Instruments of Service to meet official regulatory requirements, or for similar purposes in connection with the Project, is not to be construed as publication in

derogation of the reserved rights of the Design-Builder and the Architect, Consultants, and Contractors, and any other person or entity providing services or work for any of them.

§ 12.2 The Design-Builder and the Owner warrant that in transmitting Instruments of Service, or any other information, the transmitting party is the copyright owner of such information or has permission from the copyright owner to transmit such information for its use on the Project.

§ 12.3 Upon execution of the Agreement, the Design-Builder grants to the Owner a limited, irrevocable and non-exclusive license to use the Instruments of Service solely and exclusively for purposes of constructing, using, maintaining, altering and adding to the Project, provided that the Owner substantially performs its obligations, including prompt payment of all sums when due, under the Design-Build Documents. The license granted under this section permits the Owner to authorize its consultants and separate contractors to reproduce applicable portions of the Instruments of Service solely and exclusively for use in performing services or construction for the Project. If the Design-Builder rightfully terminates this Agreement for cause as provided in Section 13.1.4 or 13.2.1 the license granted in this Section 12.3 shall terminate.

§ 12.3.1 The Design-Builder shall obtain non-exclusive licenses from the Architect, Consultants, and Contractors, that will allow the Design-Builder to satisfy its obligations to the Owner under this Article 12. The Design-Builder's licenses from the Architect and its Consultants and Contractors shall also allow the Owner, in the event this Agreement is terminated for any reason other than the default of the Owner or in the event the Design-Builder's Architect, Consultants, or Contractors terminate their agreements with the Design-Builder for cause, to obtain a limited, irrevocable and non-exclusive license solely and exclusively for purposes of constructing, using, maintaining, altering and adding to the Project, provided that the Owner (1) agrees to pay to the Architect, Consultant or Contractor all amounts due, and (2) provide the Architect, Consultant or Contractor with the Owner's written agreement to indemnify and hold harmless the Architect, Consultant or Contractor from all costs and expenses, including the cost of defense, related to claims and causes of action asserted by any third person or entity to the extent such costs and expenses arise from the Owner's alteration or use of the Instruments of Service.

§ 12.3.2 In the event the Owner alters the Instruments of Service without the author's written authorization or uses the Instruments of Service without retaining the authors of the Instruments of Service, the Owner releases the Design-Builder, Architect, Consultants, Contractors and any other person or entity providing services or work for any of them, from all claims and causes of action arising from or related to such uses. The Owner, to the extent permitted by law, further agrees to indemnify and hold harmless the Design-Builder, Architect, Consultants, Contractors and any other person or entity providing services or work for any of them, from all costs and expenses, including the cost of defense, related to claims and causes of action asserted by any third person or entity to the extent such costs and expenses arise from the Owner's alteration or use of the Instruments of Service under this Section 12.3.2. The terms of this Section 12.3.2 shall not apply if the Owner rightfully terminates this Agreement for cause under Sections 13.1.4 or 13.2.2.

#### **ARTICLE 13 TERMINATION OR SUSPENSION**

§ 13.1 Termination or Suspension Prior to Execution of the Design-Build Amendment

§ 13.1.1 If the Owner fails to make payments to the Design-Builder for Work prior to execution of the Design-Build Amendment in accordance with this Agreement, such failure shall be considered substantial nonperformance and cause for termination or, at the Design-Builder's option, cause for suspension of performance of services under this Agreement. If the Design-Builder elects to suspend the Work, the Design-Builder shall give seven days' written notice to the Owner before suspending the Work. In the event of a suspension of the Work, the Design-Builder shall have no liability to the Owner for delay or damage caused by the suspension of the Work. Before resuming the Work, the Design-Builder shall be paid all sums due prior to suspension and any expenses incurred in the interruption and resumption of the Design-Builder's Work. The Design-Builder's compensation for, and time to complete, the remaining Work shall be equitably adjusted.

§ 13.1.2 If the Owner suspends the Project, the Design-Builder shall be compensated for the Work performed prior to notice of such suspension. When the Project is resumed, the Design-Builder shall be compensated for expenses incurred in the interruption and resumption of the Design-Builder's Work. The Design-Builder's compensation for, and time to complete, the remaining Work shall be equitably adjusted.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," %ATA," the ATA Logo, and "ATA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by ATA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the ATA Contract Documents® Terms of Service. To report copyright violations, 27 e-mail copyright@aia.org. User Notes

§ 13.1.3 If the Owner suspends the Project for more than 90 cumulative days for reasons other than the fault of the Design-Builder, the Design-Builder may terminate this Agreement by giving not less than seven days' written notice.

§ 13.1.4 Either party may terminate this Agreement upon not less than seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.

§ 13.1.5 The Owner may terminate this Agreement upon not less than seven days' written notice to the Design-Builder for the Owner's convenience and without cause.

§ 13.1.6 In the event of termination not the fault of the Design-Builder, the Design-Builder shall be compensated for Work performed prior to termination, together with Reimbursable Expenses then due and any other expenses directly attributable to termination for which the Design-Builder is not otherwise compensated. In no event shall the Design-Builder's compensation under this Section 13.1.6 be greater than the compensation set forth in Section 2.1.

#### § 13.2 Termination or Suspension Following Execution of the Design-Build Amendment § 13.2.1 Termination by the Design-Builder

§ 13.2.1.1 The Design-Builder may terminate the Contract if the Work is stopped for a period of 30 consecutive days through no act or fault of the Design-Builder, the Architect, a Consultant, or a Contractor, or their agents or employees, or any other persons or entities performing portions of the Work under direct or indirect contract with the Design-Builder, for any of the following reasons:

- Issuance of an order of a court or other public authority having jurisdiction that requires all Work to be .1 stopped;
- .2 An act of government, such as a declaration of national emergency that requires all Work to be stopped;
- .3 Because the Owner has not issued a Certificate for Payment and has not notified the Design-Builder of the reason for withholding certification as provided in Section 9.5.1, or because the Owner has not made payment on a Certificate for Payment within the time stated in the Design-Build Documents; or
- .4 The Owner has failed to furnish to the Design-Builder promptly, upon the Design-Builder's request, reasonable evidence as required by Section 7.2.7.

§ 13.2.1.2 The Design-Builder may terminate the Contract if, through no act or fault of the Design-Builder, the Architect, a Consultant, a Contractor, or their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Design-Builder, repeated suspensions, delays or interruptions of the entire Work by the Owner as described in Section 13.2.3 constitute in the aggregate more than 100 percent of the total number of days scheduled for completion, or 120 days in any 365-day period, whichever is less.

§ 13.2.1.3 If one of the reasons described in Section 13.2.1.1 or 13.2.1.2 exists, the Design-Builder may, upon seven days' written notice to the Owner, terminate the Contract and recover from the Owner payment for Work executed, including reasonable overhead and profit, costs incurred by reason of such termination, and damages.

§ 13.2.1.4 If the Work is stopped for a period of 60 consecutive days through no act or fault of the Design-Builder or any other persons or entities performing portions of the Work under contract with the Design-Builder because the Owner has repeatedly failed to fulfill the Owner's obligations under the Design-Build Documents with respect to matters important to the progress of the Work, the Design-Builder may, upon seven additional days' written notice to the Owner, terminate the Contract and recover from the Owner as provided in Section 13.2.1.3.

# § 13.2.2 Termination by the Owner For Cause

§ 13.2.2.1 The Owner may terminate the Contract if the Design-Builder

- fails to submit the Proposal by the date required by this Agreement, or if no date is indicated, within a .1 reasonable time consistent with the date of Substantial Completion;
- .2 repeatedly refuses or fails to supply an Architect, or enough properly skilled Consultants, Contractors, or workers or proper materials;
- .3 fails to make payment to the Architect, Consultants, or Contractors for services, materials or labor in accordance with their respective agreements with the Design-Builder;
- .4 repeatedly disregards applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of a public authority; or
- .5 is otherwise guilty of substantial breach of a provision of the Design-Build Documents.

§ 13.2.2.2 When any of the above reasons exist, the Owner may without prejudice to any other rights or remedies of the Owner and after giving the Design-Builder and the Design-Builder's surety, if any, seven days' written notice, terminate employment of the Design-Builder and may, subject to any prior rights of the surety:

- Exclude the Design-Builder from the site and take possession of all materials, equipment, tools, and .1 construction equipment and machinery thereon owned by the Design-Builder;
- .2 Accept assignment of the Architect, Consultant and Contractor agreements pursuant to Section 3.1.15; and
- .3 Finish the Work by whatever reasonable method the Owner may deem expedient. Upon written request of the Design-Builder, the Owner shall furnish to the Design-Builder a detailed accounting of the costs incurred by the Owner in finishing the Work.

§ 13.2.2.3 When the Owner terminates the Contract for one of the reasons stated in Section 13.2.2.1, the Design-Builder shall not be entitled to receive further payment until the Work is finished.

§ 13.2.2.4 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Design-Builder. If such costs and damages exceed the unpaid balance, the Design-Builder shall pay the difference to the Owner. The obligation for such payments shall survive termination of the Contract.

# § 13.2.3 Suspension by the Owner for Convenience

§ 13.2.3.1 The Owner may, without cause, order the Design-Builder in writing to suspend, delay or interrupt the Work in whole or in part for such period of time as the Owner may determine.

§ 13.2.3.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay or interruption as described in Section 13.2.3.1. Adjustment of the Contract Sum shall include profit. No adjustment shall be made to the extent

- that performance is, was or would have been so suspended, delayed or interrupted by another cause for .1 which the Design-Builder is responsible; or
- .2 that an equitable adjustment is made or denied under another provision of the Contract.

#### § 13.2.4 Termination by the Owner for Convenience

§ 13.2.4.1 The Owner may, at any time, terminate the Contract for the Owner's convenience and without cause.

§ 13.2.4.2 Upon receipt of written notice from the Owner of such termination for the Owner's convenience, the Design-Builder shall

- cease operations as directed by the Owner in the notice; .1
- take actions necessary, or that the Owner may direct, for the protection and preservation of the Work; .2 and.
- except for Work directed to be performed prior to the effective date of termination stated in the notice, .3 terminate all existing Project agreements, including agreements with the Architect, Consultants, Contractors, and purchase orders, and enter into no further Project agreements and purchase orders.

§ 13.2.4.3 In case of such termination for the Owner's convenience, the Design-Builder shall be entitled to receive payment for Work executed, and costs incurred by reason of such termination, along with reasonable overhead and profit on the Work not executed.

# ARTICLE 14 CLAIMS AND DISPUTE RESOLUTION

#### § 14.1 Claims

§ 14.1.1 Definition. A Claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money, or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and Design-Builder arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim.

§ 14.1.2 Time Limits on Claims. The Owner and Design-Builder shall commence all claims and causes of action, whether in contract, tort, breach of warranty or otherwise, against the other, arising out of or related to the Contract in accordance with the requirements of the binding dispute resolution method selected in Section 1.3, within the time

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 29 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes (2034726772)

period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Design-Builder waive all claims and causes of action not commenced in accordance with this Section 14.1.2.

#### § 14.1.3 Notice of Claims

§ 14.1.3.1 Prior To Final Payment. Prior to Final Payment, Claims by either the Owner or Design-Builder must be initiated by written notice to the other party within 21 days after occurrence of the event giving rise to such Claim or within 21 days after the claimant first recognizes the condition giving rise to the Claim, whichever is later.

§ 14.1.3.2 Claims Arising After Final Payment. After Final Payment, Claims by either the Owner or Design-Builder that have not otherwise been waived pursuant to Sections 9.10.4 or 9.10.5, must be initiated by prompt written notice to the other party. The notice requirement in Section 14.1.3.1 and the Initial Decision requirement as a condition precedent to mediation in Section 14.2.1 shall not apply.

§ 14.1.4 Continuing Contract Performance. Pending final resolution of a Claim, except as otherwise agreed in writing or as provided in Section 9.7 and Article 13, the Design-Builder shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Design-Build Documents.

§ 14.1.5 Claims for Additional Cost. If the Design-Builder intends to make a Claim for an increase in the Contract Sum, written notice as provided herein shall be given before proceeding to execute the portion of the Work that relates to the Claim. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Section 10.4.

#### § 14.1.6 Claims for Additional Time

§ 14.1.6.1 If the Design-Builder intends to make a Claim for an increase in the Contract Time, written notice as provided herein shall be given. The Design-Builder's Claim shall include an estimate of cost and of probable effect of delay on progress of the Work. In the case of a continuing delay, only one Claim is necessary.

§ 14.1.6.2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time, could not have been reasonably anticipated, and had an adverse effect on the scheduled construction.

#### § 14.1.7 Claims for Consequential Damages

The Design-Builder and Owner waive Claims against each other for consequential damages arising out of or relating to this Contract. This mutual waiver includes

- damages incurred by the Owner for rental expenses, for losses of use, income, profit, financing, .1 business and reputation, and for loss of management or employee productivity or of the services of such persons; and
- .2 damages incurred by the Design-Builder for principal office expenses including the compensation of personnel stationed there, for losses of financing, business and reputation, and for loss of profit except anticipated profit arising directly from the Work.

This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination in accordance with Article 13. Nothing contained in this Section 14.1.7 shall be deemed to preclude an award of liquidated damages, when applicable, in accordance with the requirements of the Design-Build Documents.

#### § 14.2 Initial Decision

§ 14.2.1 An initial decision shall be required as a condition precedent to mediation of all Claims between the Owner and Design-Builder initiated prior to the date final payment is due, excluding those arising under Sections 10.3 and 10.4 of the Agreement and Sections B.3.2.9 and B.3.2.10 of Exhibit B to this Agreement, unless 30 days have passed after the Claim has been initiated with no decision having been rendered. Unless otherwise mutually agreed in writing, the Owner shall render the initial decision on Claims.

# § 14.2.2 Procedure

§ 14.2.2.1 Claims Initiated by the Owner. If the Owner initiates a Claim, the Design-Builder shall provide a written response to Owner within ten days after receipt of the notice required under Section 14.1.3.1. Thereafter, the Owner shall render an initial decision within ten days of receiving the Design-Builder's response: (1) withdrawing the Claim

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," %ATA," the ATA Logo, and "ATA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by ATA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the ATA Contract Documents® Terms of Service. To report copyright violations, 30 e-mail copyright@aia.org. User Notes:

in whole or in part, (2) approving the Claim in whole or in part, or (3) suggesting a compromise.

§ 14.2.2.2 Claims Initiated by the Design-Builder. If the Design-Builder initiates a Claim, the Owner will take one or more of the following actions within ten days after receipt of the notice required under Section 14.1.3.1: (1) request additional supporting data, (2) render an initial decision rejecting the Claim in whole or in part, (3) render an initial decision approving the Claim, (4) suggest a compromise or (5) indicate that it is unable to render an initial decision because the Owner lacks sufficient information to evaluate the merits of the Claim.

§ 14.2.3 In evaluating Claims, the Owner may, but shall not be obligated to, consult with or seek information from persons with special knowledge or expertise who may assist the Owner in rendering a decision. The retention of such persons shall be at the Owner's expense.

§ 14.2.4 If the Owner requests the Design-Builder to provide a response to a Claim or to furnish additional supporting data, the Design-Builder shall respond, within ten days after receipt of such request, and shall either (1) provide a response on the requested supporting data, (2) advise the Owner when the response or supporting data will be furnished or (3) advise the Owner that no supporting data will be furnished. Upon receipt of the response or supporting data, if any, the Owner will either reject or approve the Claim in whole or in part.

§ 14.2.5 The Owner's initial decision shall (1) be in writing; (2) state the reasons therefor; and (3) identify any change in the Contract Sum or Contract Time or both. The initial decision shall be final and binding on the parties but subject to mediation and, if the parties fail to resolve their dispute through mediation, to binding dispute resolution.

§ 14.2.6 Either party may file for mediation of an initial decision at any time, subject to the terms of Section 14.2.6.1.

§ 14.2.6.1 Either party may, within 30 days from the date of an initial decision, demand in writing that the other party file for mediation within 60 days of the initial decision. If such a demand is made and the party receiving the demand fails to file for mediation within the time required, then both parties waive their rights to mediate or pursue binding dispute resolution proceedings with respect to the initial decision.

§ 14.2.7 In the event of a Claim against the Design-Builder, the Owner may, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Design-Builder's default, the Owner may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy.

§ 14.2.8 If a Claim relates to or is the subject of a mechanic's lien, the party asserting such Claim may proceed in accordance with applicable law to comply with the lien notice or filing deadlines.

#### § 14.3 Mediation

§ 14.3.1 Claims, disputes, or other matters in controversy arising out of or related to the Contract, except those waived as provided for in Sections 9.10.4, 9.10.5, and 14.1.7, shall be subject to mediation as a condition precedent to binding dispute resolution.

§ 14.3.2 The parties shall endeavor to resolve their Claims by mediation which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of the Agreement. A request for mediation shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of binding dispute resolution proceedings but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration proceeding is stayed pursuant to this Section 14.3.2, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.

§ 14.3.3 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction.

AIA Document A141" - 2014. Copyright © 2004 and 2014 by The American Institute of Architects. All rights reserved. The "American Institute Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This draft was produced by AIA software at 12:51:21 ET on 04/23/2021 under Order No.4655459305 which expires on 01/25/2022, is not for resale, is licensed 31 for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes

# § 14.4 Arbitration

§ 14.4.1 If the parties have selected arbitration as the method for binding dispute resolution in Section 1.3, any Claim subject to, but not resolved by, mediation shall be subject to arbitration which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of the Agreement. A demand for arbitration shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the arbitration. The party filing a notice of demand for arbitration must assert in the demand all Claims then known to that party on which arbitration is permitted to be demanded.

§ 14.4.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the Claim would be barred by the applicable statute of limitations or statute of repose. For statute of limitations or statute of repose purposes, receipt of a written demand for arbitration by the person or entity administering the arbitration shall constitute the institution of legal or equitable proceedings based on the Claim.

§ 14.4.2 The award rendered by the arbitrator or arbitrators shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction.

§ 14.4.3 The foregoing agreement to arbitrate, and other agreements to arbitrate with an additional person or entity duly consented to by parties to the Agreement, shall be specifically enforceable under applicable law in any court having jurisdiction thereof.

#### § 14.4.4 Consolidation or Joinder

§ 14.4.1 Either party, at its sole discretion, may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation, (2) the arbitrations to be consolidated substantially involve common questions of law or fact, and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).

§ 14.4.4.2 Either party, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.

§ 14.4.4.3 The Owner and Design-Builder grant to any person or entity made a party to an arbitration conducted under this Section 14.4, whether by joinder or consolidation, the same rights of joinder and consolidation as the Owner and Design-Builder under this Agreement.

# ARTICLE 15 MISCELLANEOUS PROVISIONS

#### § 15.1 Governing Law

The Contract shall be governed by the law of the place where the Project is located except that, if the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 14.4

# § 15.2 Successors and Assigns

§ 15.2.1 The Owner and Design-Builder, respectively, bind themselves, their partners, successors, assigns and legal representatives to the covenants, agreements and obligations contained in the Design-Build Documents. Except as provided in Section 15.2.2, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.

§ 15.2.2 The Owner may, without consent of the Design-Builder, assign the Contract to a lender providing construction financing for the Project, if the lender assumes the Owner's rights and obligations under the Design-Build Documents. The Design-Builder shall execute all consents reasonably required to facilitate such assignment.

§ 15.2.3 If the Owner requests the Design-Builder, Architect, Consultants, or Contractors to execute certificates, other than those required by Section 3.1.10, the Owner shall submit the proposed language of such certificates for review at least 14 days prior to the requested dates of execution. If the Owner requests the Design-Builder, Architect,

Consultants, or Contractors to execute consents reasonably required to facilitate assignment to a lender, the Design-Builder, Architect, Consultants, or Contractors shall execute all such consents that are consistent with this Agreement, provided the proposed consent is submitted to them for review at least 14 days prior to execution. The Design-Builder, Architect, Consultants, and Contractors shall not be required to execute certificates or consents that would require knowledge, services or responsibilities beyond the scope of their services.

# § 15.3 Written Notice

Written notice shall be deemed to have been duly served if delivered in person to the individual, to a member of the firm or entity, or to an officer of the corporation for which it was intended; or if delivered at, or sent by registered or certified mail or by courier service providing proof of delivery to, the last business address known to the party giving notice.

#### § 15.4 Rights and Remedies

§ 15.4.1 Duties and obligations imposed by the Design-Build Documents, and rights and remedies available thereunder, shall be in addition to and not a limitation of duties, obligations, rights and remedies otherwise imposed or available by law.

§ 15.4.2 No action or failure to act by the Owner or Design-Builder shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach thereunder, except as may be specifically agreed in writing.

#### § 15.5 Tests and Inspections

§ 15.5.1 Tests, inspections and approvals of portions of the Work shall be made as required by the Design-Build Documents and by applicable laws, statutes, ordinances, codes, rules and regulations or lawful orders of public authorities. Unless otherwise provided, the Design-Builder shall make arrangements for such tests, inspections and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections and approvals. The Design-Builder shall give the Owner timely notice of when and where tests and inspections are to be made so that the Owner may be present for such procedures. The Owner shall bear costs of (1) tests, inspections or approvals that do not become requirements until after bids are received or negotiations concluded, and (2) tests, inspections or approvals where building codes or applicable laws or regulations prohibit the Owner from delegating their cost to the Design-Builder.

§ 15.5.2 If the Owner determines that portions of the Work require additional testing, inspection or approval not included under Section 15.5.1, the Owner will instruct the Design-Builder to make arrangements for such additional testing, inspection or approval by an entity acceptable to the Owner, and the Design-Builder shall give timely notice to the Owner of when and where tests and inspections are to be made so that the Owner may be present for such procedures. Such costs, except as provided in Section 15.5.3, shall be at the Owner's expense.

§ 15.5.3 If such procedures for testing, inspection or approval under Sections 15.5.1 and 15.5.2 reveal failure of the portions of the Work to comply with requirements established by the Design-Build Documents, all costs made necessary by such failure shall be at the Design-Builder's expense.

§ 15.5.4 Required certificates of testing, inspection or approval shall, unless otherwise required by the Design-Build Documents, be secured by the Design-Builder and promptly delivered to the Owner.

§ 15.5.5 If the Owner is to observe tests, inspections or approvals required by the Design-Build Documents, the Owner will do so promptly and, where practicable, at the normal place of testing.

§ 15.5.6 Tests or inspections conducted pursuant to the Design-Build Documents shall be made promptly to avoid unreasonable delay in the Work.

#### § 15.6 Confidential Information

If the Owner or Design-Builder transmits Confidential Information, the transmission of such Confidential Information constitutes a warranty to the party receiving such Confidential Information that the transmitting party is authorized to transmit the Confidential Information. If a party receives Confidential Information, the receiving party shall keep the Confidential Information strictly confidential and shall not disclose it to any other person or entity except as set forth in Section 15.6.1.

**§ 15.6.1** A party receiving Confidential Information may disclose the Confidential Information as required by law or court order, including a subpoena or other form of compulsory legal process issued by a court or governmental entity. A party receiving Confidential Information may also disclose the Confidential Information to its employees, consultants or contractors in order to perform services or work solely and exclusively for the Project, provided those employees, consultants and contractors are subject to the restrictions on the disclosure and use of Confidential Information as set forth in this Contract.

#### § 15.7 Capitalization

Terms capitalized in the Contract include those that are (1) specifically defined, (2) the titles of numbered articles or (3) the titles of other documents published by the American Institute of Architects.

#### § 15.8 Interpretation

**§ 15.8.1** In the interest of brevity the Design-Build Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

**§ 15.8.2** Unless otherwise stated in the Design-Build Documents, words which have well-known technical or construction industry meanings are used in the Design-Build Documents in accordance with such recognized meanings.

#### ARTICLE 16 SCOPE OF THE AGREEMENT

§ 16.1 This Agreement is comprised of the following documents listed below:

- .1 AIA Document A141<sup>™</sup>-2014, Standard Form of Agreement Between Owner and Design-Builder
- .2 AIA Document A141<sup>TM</sup>–2014, Exhibit A, Design-Build Amendment, if executed
- .3 AIA Document A141<sup>TM</sup>–2014, Exhibit B, Insurance and Bonds
- .4 AIA Document A141<sup>TM</sup>–2014, Exhibit C, Sustainable Projects, if completed
- .5 AIA Document E203<sup>™</sup>–2013, Building Information Modeling and Digital Data Exhibit, if completed, or the following:

« »

.6 Other:

« »

This Agreement entered into as of the day and year first written above.

**OWNER** (Signature)

« »« »

(Printed name and title)

#### **DESIGN-BUILDER** (Signature)

« »« »

(Printed name and title)



# Exhibit D

# Solar Shade Structure Design

# &

Proposal

# **BUDGETARY SOLAR PHOTOVOLTAIC PROPOSAL FOR:** MAGNOLIA SCIENCE ACADEMY

# 18220 Sherman Way Reseda, CA 91355



# THE STRENGTH BETWEEN SUN AND SHADE CONSTRUCTING A RENEWABLE FUTURE



CONSTRUCTING A RENEWABLE FUTURE

7/19/21





EXECUTIVE SUMMARY	3
M BAR C RENEWABLES	4
WHO WE ARE	5
CUSTOMER SATISFACTION	6
SAMPLE STRUCTURE TYPES	<u> </u>
PROJECT – SAMPLE LISTING	8
SCOPE OF WORK	9
CONCEPTUAL LAYOUT	10
CONTRACTOR'S LICENSES	11
PRICING: ASSUMPTIONS	12
PRICING: EXCLUSIONS	13

# EXECUTIVE SUMMARY

Magnolia Science Academy	
--------------------------	--

Installation Type	Non-Solar Full Coverage Canopy
Structure Size	10,069 SF
Total Price	\$340,164
Unit Price	\$33.78/SF



# **M BAR C RENEWABLES**



# Recycling our natural resources. Saving money. Providing for the future.

M Bar C's Renewable Energy division was created to meet the increasing demands for sustainable energy. From Solar PV to Advanced Energy Storage, we offer a variety of custom energy upgrades designed to meet your specific needs. M Bar C is now your one-stop-shop for everything under the sun.

# TAKE A DEEPER LOOK

M Bar C has vertically integrated. With in-house services we have effectively reduced multiple layers of communication and markup. We pride ourselves in efficient project management, coordination, and meeting deadlines. This efficiency in simplicity has been the key to success for small and large scale projects.

# WHO WE ARE



M Bar C Construction Inc., is the industry leader in the design and construction of custom solar carports. With over 60 Megawatts installed in 2017, we hold the largest market share for carports in the United States. M Bar C Construction Inc. is a recognized world-class steel contractor, dedicated to superior construction practices, innovative product development and the highest production safety standards.



# WHY CHOOSE US?

**INDUSTRY LEADER** We have the greatest market share of solar carports in the country. We are leading the industry with cutting edge designs and construction practices.

**QUALITY** Our reputation is built from quality. We build the best product, provide the best service and deliver what is promised. We maintain quality materials, construction & project management.

**SAFETY** Our crews are highly trained to handle the latest materials and equipment, and maintain Cal/OSHA compliance at all times. We set the bar higher than the industry standard.

ATTENTION TO DETAIL From the earliest planning stages to a clean work site at the end of each day, M Bar C is keenly aware of every aspect for each project.

**EFFICIENCY** Our extensive and diverse experience allows us to maximize productivity without cutting corners. This assures maximum value and on time project completion.

**VALUES** Our company is built on a culture of family, integrity, commitment, quality, growth & leadership. These principles manifest into a positive customer experience.

# **CUSTOMER SATISFACTION**



# WHAT DO CUSTOMERS SAY ABOUT US?

"M Bar C Construction is fast and professional. They take safety seriously and stick to the plan. Our Project Manager was impressed. We like working with them!"

Kevin M Clements, LEED® AP Director of Project Development, EMCOR Group, Inc. "Thanks for all the great work on site through this project as well. Your team is an example of what others should look to as how it should be done."

Joshua Goldberg Site Superintendent, Borrego Solar Systems, Inc. "I love having you guys on my projects because you show up with a small crew, but it's like you bring an army! You work faster and cleaner than anyone else l've worked with. Thank you!"

Alex Superintendent Solar City/Tesla



MBARC CONSTRUCTION

#### **CONSTRUCTING A RENEWABLE FUTURE**

# SAMPLE STRUCTURE TYPES

(CUSTOM DESIGNS AVAILABLE)



LOCATION Mountain Valley, CA STRUCTURE TYPE Ground Mount & Carports



LOCATION Los Angeles, CA STRUCTURE TYPE Rooftop



LOCATION Chula Vista, CA STRUCTURE TYPE Parking Garage – "T" Structure



LOCATION Vista, CA STRUCTURE TYPE "T" Structure



LOCATION Carlsbad, CA STRUCTURE TYPE Semi-Cantilever Structure



LOCATION San Jacinto, CA STRUCTURE TYPE Ground Mount

# **PROJECTS – SAMPLE LISTING**



**SCHOOLS/COLLEGES** Alliance College-Ready Public Schools: College Ready Middle Academy 4, College Ready Middle Academy 5, Renee & Meyer Luskin College Ready Academy, Patti & Peter Neuwirth Leadership Academy, Cindy & Bill Simon Technology High, Jack H. Skirball Middle, Ted K. Tajima High. SIATech Administrative Offices, Bright Star Valor Academy, Discovery Charter Prepatory, Vista Charter Middle

**GOVERNMENT/MILITARY/MUNICIPAL** Camp Pendleton *(multiple sites)*, NAS Miramar *(multiple sites)*, NAS North Island, MCRD, FAA TRACON, Grand Junction VA, Monterrey VA, West Los Angeles VA, March AFB, Carpinteria Valley Water District

**CORPORATE** Google, Bayer Health Care, Wells Fargo, Intel, Hunter Industries, Tustin Metro Link, Sierra Toyota, Palm Springs Motors, Novartis Pharmaceuticals, Edwards Life Sciences

**SPECIALTY DESIGN** San Diego Zoo, Tesla HQ, Galpin Ford, McDonalds (Riverside), Taco Bell (Riverside), Contra Costa Community College, Saddleback College, Alga Norte Park

**GROUND MOUNTS** Mojave/California City, US Gypsum, Taft, Cerro Coso College, Two Bunch Palms, Bubbling Wells Elementary, City of Riverside, Riverside County, Lucia Mar USD

NEW CONSTRUCTION Cypress Walk, Pulse Millenia Apartments, Quarry Creek Apartments, Skyline Elementary

NON-PROFITS Calvary Chapel, St. Patrick's Church, Mission Hills Church, Lighthouse Baptist Church, St Charles Borromeo Church & School, Corpus Christi Church, St. Pius X Church & School, St. Francis of Assisi Church & School



# **SCOPE OF WORK**

# **GENERAL OVERVIEW**

 M Bar C's custom designed solar-ready carports are made of the highest quality HSS tube steel columns and beams. This durable and structurally sound approach imbeds each column at the footing and uses a welded-side plate connection, eliminating bolted connections for an aesthetically pleasing look. The combination of site-specific engineering and a high quality M Bar C installation culminate in an efficient and reliable PV system.

# **PRICING INCLUDES**

- (1) solar ready two post long span canopy
- · Procurement of materials: Steel
- Structural design & engineering
- · Construction of system with all applicable building codes, plans, and specifications
- HSS columns and beams with light gauge "C" purlins
- · Installation of approximately 10,069 SF of two post long span solar ready canopies
- · Coring of asphalt for footings if no conflicts occur
- All sales tax for material purchases

# WARRANTIES

M Bar C Workmanship Warranty

1-year workmanship on all Columns, Beams, Purlins, and Concrete as per M Bar C Construction Inc., Drawings & Specifications for the period beginning at substantial completion and acceptance of M Bar C Construction Inc's scope of work and ending (1) year after that date. After the 1-year warranty period, M Bar C will pass through all individual supplier warranties with specific details on the coverage. Detailed manufacturer warranties can be provided upon request.


## **CONCEPTUAL LAYOUT: OPTION #1**





### **CONTRACTOR'S LICENSE**



### LICENSE # 869960

M Bar C Construction, Inc.

1770 La Costa Meadows Dr San Marcos, CA 92078 Business Phone Number: (760) 744-4131

> Entity: Corporation Issue Date: 12/22/2005 Expire Date: 12/31/2021

License Status: Current & Active

Classifications: B, C51

# **PRICING: ASSUMPTIONS**

- 1. All technology & equipment specified in this project will meet or exceed all current safety, environmental and interconnection standards, as well as all facility structural weight and fire safety requirements
- 2. Building department and utility inspections will be scheduled during normal business hours
- 3. Pricing assumes M Bar C employees will have free and clear access to array location during business hours (M-F 0700-1700)
- 4. Engineering included provides the structural framing package and the structural calculations package only for integration into the permit submittal package (by others). Specifically excluded are site plans, column layout plans, equipment pads, equipment attachment details, site visits, as-built drawings, submittal review, cad standard, military/government projects, site-specific architectural, electrical, mechanical, title-24, BMP, civil, SWPPP, and SUSUMP, etc. Quotes for these services and others are available upon request.
- 5. System layout is acceptable to AHJ in terms of clearances
- 6. Single mobilization and completion of project in one phase
- 7. Current/accurate as-builts in the form of CAD files will be provided by others as basis of design
- 8. Architectural, civil, and any surveying will be provided by architect of record
- 9. M Bar C has allocated \$3,500 for special inspection fees. M Bar C will adjust the amount billed to reflect actual values in the form of a change order.
- 10. On-site staging areas are available for storage of equipment and materials
- 11. No special safety requirements beyond M Bar C Construction's standard safety regulations will be enforced
- 12. 29'-0" bays assumed for all structures
- 13. 2-degree tilt assumed
- 14. 23'-0" clear height assumed for all structures
- 15. M Bar C assumes ground is fairly level. Change in grade should be + or 1/10th grading performance standard. Following grade assumed.
- 16. Pricing includes coring of asphalt for footings if no conflicts occur
- 17. Hot dipped galvanized columns and beams
- 18. Pricing is based on soils reports provided by client. If any conditions not specifically stated in the geotechnical information provided are experienced, additional costs for mitigation may be incurred.
- 19. Roof live load for solar-ready canopy with roof decking loading subject to change per site conditions
- 20. Site conditions unverified (no site walk performed; pricing subject to change)
- 21. Any underground conduits for future solar will be installed by 'Others'.
- 22. While USA will be called out on every project, some sites may not be accessible in which case it will be the owner's responsibility to engage a third party for utility mark out. M Bar C is not responsible for damage to unmarked underground utility lines including but not limited to electrical, gas, water, sewer, irrigation, fiber optic, television etc.
- 23. Ground penetrating radar (GPR) is accurate down to a depth of 3'-4'. There may be underground utilities below this level that are untraceable with GPR that would require the use of an underground camera to be located at additional cost.
- 24. Temporary facilities including dumpsters, toilets, handwash stations, etc. will be provided by 'Others'
- 25. Pricing is based on private wage rates
- 26. Final pricing is subject to change based on formal engineering
- 27. The cost of steel at time of execution of this contract is high and volatile. Contractor and Subcontractor will collaborate to minimize the impact of steel cost, but contract price is expected to require adjustment to accommodate steel costs at time of steel shipment.

# **PRICING: EXCLUSIONS**

- 1. Hard Rock Drilling, Underground Obstructions, Caving Soils, Casing, and Water Mitigation. \*NOTE: The client should consider carefully whether M Bar C should "attempt" to drill when conditions are present such that it can be reasonably assumed that subsurface conditions are likely to result in M Bar C's inability to drill successfully. Under this scenario, costs to the client will be lower if a sub driller is engaged from the beginning. No credit from M Bar C will be available, if an "ill-advised" attempt to have M Bar C drill is made.
- 2. Existing as built document creation and circuit tracing of existing electrical system due to inaccurate or missing as built information
- 3. Supplying, installing, and maintaining a storm water pollution prevention plan (SWPPP)
- 4. Any underground trenching or providing any underground conduits
- 5. Fire alarm, fire sprinkler, CCTV, security or communication work
- 6. Concrete coring, cutting, demolition, repair, or replacement
- 7. Redesigns after the 100% drawing submittal to the AHJ
- 8. Removal and/or disposal of hazardous materials
- 9. Union Wages, Skilled Workforce Requirements
- 10. Any excavation beyond column foundations
- 11. Pedestrian & vehicle traffic control
- 12. Landscaping and/or tree removal
- 13. Re-creation of building plan sets
- 14. Permitting and associated fees
- 15. Skilled workforce requirements
- 16. Design of low voltage systems
- 17. Lightening protection systems
- 18. Demolition of any light poles
- 19. ADA design/compliance
- 20. Painting of the structure
- 21. Prevailing wage rates
- 22. Concrete abutments
- 23. Painting of conduits
- 24. Sprinklers & gutters
- 25. Permanent fencing
- 26. Destructive testing
- 27. Survey services
- 28. Bonding



### Cover Sheet

### Approval of Two-Year Extension of Charter School Renewals (Renewal Relief)

Section:IV. Action ItemsItem:D. Approval of Two-Year Extension of Charter SchoolRenewals (Renewal Relief)Purpose:Purpose:VoteSubmitted by:Charter Term Extension Resolution.pdf



Board Agenda Item #	IV D: Action Item
Date:	August 12, 2021
То:	Magnolia Public Schools Board of Directors
From:	Alfredo Rubalcava, CEO & Superintendent
Staff Lead:	Alfredo Rubalcava, CEO & Superintendent
RE:	Charter Term Two-Year Extension Resolution

### Proposed Board Motion

I move that the MPS board approve the resolution that acknowledges the two-year extensions for all Magnolia Science Academies (MSA) and recognize the new charter term expiration dates as follows:

- MSA 1, 2, and 3 June 30, 2024;
- MSA 4 and 5 June 30, 2025;
- MSA 6, 7, and Santa Ana June 30, 2026; and
- MSA Bell and San Diego June 30, 2027.

### **Introduction**

The resolution brought forth memorializes the changes that have been made as part of Education Code Section 47607.4, which grants schools who have charters expiring on or between January 1, 2022, and June 30, 2025, a two-year extension.

This two-year extension affects all ten of our schools and pushes back the renewal dates to the new charter term expiration dates listed above.

### **Background**

Due to the impact of the COVID-19 pandemic, no state testing was conducted during the past two years which impacts the renewal in showing student growth since only one our two-years of data will be reflected in the renewals. In order to address the impact, the California State Legislature added Section 47607.4 to the Education Code which includes the two-year extension to charter schools.

Currently, Magnolia Public Schools has four authorizers: Los Angeles Unified School District, Los Angeles County Office of Education, San Diego Unified School District, and State Board of Education.

### <u>Analysis (If applicable)</u>

• NA

**Budget Implications** 

• NA

Exhibits (attachments):

• Resolution: Approval of Legal Extension of Charter Terms by Two Years

### RESOLUTION OF THE MAGNOLIA PUBLIC SCHOOLS BOARD OF DIRECTORS

### (A California Non-Profit Public Benefit Corporation)

### Board Resolution 20210812-1

### A RESOLUTION to recognize the legal extension of charter term by two years pursuant to Education Code Section 47607.4

WHEREAS, Magnolia Public Schools operates Magnolia Science Academy, Magnolia Science Academy 2, Magnolia Science Academy 3, Magnolia Science Academy 4, Magnolia Science Academy 5, Magnolia Science Academy 6, Magnolia Science Academy 7, Magnolia Science Academy Bell, Magnolia Science Academy San Diego, and Magnolia Science Academy Santa Ana (hereinafter, collectively the "Charter Schools"); and

WHEREAS, the Los Angeles Unified School District approved the renewal of the Magnolia Science Academy 4 charter petition for a term of five years, from July 1, 2018 to June 30, 2023; the renewal of the Magnolia Science Academy 6 charter petition and the Magnolia Science Academy 7 charter petition for a term of five years, from July 1, 2019 to June 30, 2024; and the renewal of the Magnolia Science Academy Bell charter petition for a term of five years, from July 1, 2019 to June 30, 2024; and the renewal of the Magnolia Science Academy Bell charter petition for a term of five years, from July 1, 2020 to June 30, 2025; and

WHEREAS, the Los Angeles County Office of Education approved the renewal of the Magnolia Science Academy charter petition, the Magnolia Science Academy 2 charter petition, and the Magnolia Science Academy 3 charter petition for a term of five years, from July 1, 2017 to June 30, 2022; and the renewal of the Magnolia Science Academy 5 charter petition for a term of five years, from July 1, 2018 to June 30, 2023; and

WHEREAS, the San Diego Unified School District approved the renewal of the Magnolia Science Academy San Diego charter petition for a term of five years, from July 1, 2020 to June 30, 2025; and

WHEREAS, the State Board of Education approved the renewal of the Magnolia Science Academy Santa Ana charter petition for a term of five years, from July 1, 2019 to June 30, 2024; and

WHEREAS, the California State Legislature added Section 47607.4 to the Education Code, which provides: "[n]otwithstanding the renewal process and criteria established in Sections 47605.9, 47607, and 47607.2 or any other law, effective July 1, 2021, <u>all charter schools whose term</u> <u>expires on or between January 1, 2022, and June 30, 2025, inclusive, shall have their term</u> <u>extended by two years</u>."

NOW, THEREFORE, BE IT RESOLVED, that the charter terms for Magnolia Science Academy, Magnolia Science Academy 2, and Magnolia Science Academy 3 have been extended by two years, to expire on June 30, 2024, by operation of law; and

BE IT FURTHER RESOLVED, that the charter terms for Magnolia Science Academy 4 and Magnolia Science Academy 5 have been extended by two years, to expire on June 30, 2025, by operation of law; and

BE IT FURTHER RESOLVED, that the charter terms for Magnolia Science Academy 6, Magnolia Science Academy 7, and Magnolia Science Academy Santa Ana have been extended by two years, to expire on June 30, 2026, by operation of law; and

BE IT FURTHER RESOLVED, that the charter terms for Magnolia Science Academy Bell and Magnolia Science Academy San Diego have been extended by two years, to expire on June 30, 2027, by operation of law.

\* \* \*

IN WITNESS WHEREOF, the Board of Directors has adopted the above resolution by the following vote at a regular Board meeting this 10 day of August, 2021.

Board Member	Vote
Ms. Sandra Covarrubias	
Dr. Umit Yapanel	
Mr. Mekan Muhammedov	
Ms. Diane Gonzalez	
Dr. Salih Dikbas	

AYES: NOS: ABSTENTIONS: ABSENT:

I, Sandra Covarrubias, am the Chair of the Magnolia Public Schools Board of Directors. I hereby certify that the foregoing is a true copy of a resolution duly and legally adopted by the Board of Directors on August 10, 2021, and that this resolution has not been revoked.

Date:

Sandra Covarrubias, Chair Magnolia Public Schools Board of Directors

## Cover Sheet

### Approval of Orange County Soccer Club Donation (Mini-Pitch)

Section: Item: Pitch) Purpose: Submitted by: Related Material: IV. Action Items E. Approval of Orange County Soccer Club Donation (Mini-

Vote

OC Soccer Club Donation.pdf



Board Agenda Item #	IV E: Action Item
Date:	August 12, 2021
То:	Magnolia Public Schools - Board of Directors
From:	Dr. Brenda D. Lopez, Chief External Officer
Staff Lead:	Dr. Brenda D. Lopez, Chief External Officer Steven Keskinturk, Principal MSA Santa Ana
RE:	Approval of grant received for MSA-SA from OC Soccer Club to build a Mini-Pitch Soccer Court on site

### PROPOSED BOARD MOTION

I move that the board conditionally approve the donation of a mini-pitch soccer court by the Orange County Soccer Club for MSA-Santa Ana; contingent upon the staff receiving a Memorandum of Understanding.

### BACKGROUND

Magnolia Public Schools is a tremendous resource for families in Southern California. As we devise innovative programs to meet our students' interests and challenges and share our accomplishments, we will attract more and more friends and supporters.

Magnolia Science Academy has led highly engaging work with their community especially during a Global pandemic. One of their effort's centers around today's proposal in partnerships with the Orange County Soccer Club. Soccer clinics have been hosted on the school site and have been extremely popular with community. Building a mini-pitch can open doors to a growing partnership that can benefit the entire community.

### Partnership Value Grows With Consistent Engagement



### **OCSC AIMS TO SERVE THOUSANDS OF KIDS BY 2026**

· Mini-pitches are an integral part of OCSC's plan to increase community engagement in underserved areas and focus on youth health and wellness

### LEGACY OF LASTING IMPACT

Pitches will stand for a minimum of six years and partners benefit from both recurring and revolving viewership of ad signage





Desert West, Phoenix, AZ



Santa Fe South High School, Oklahoma City, OK



### **INTRODUCTION**

About Orange County Soccer Club	Located in Irvine, CA OCSC plays in the USL Championship, the 2 <sup>nd</sup> Division of pro soccer in the U.S.
	<ul> <li>In 2020, OCSC had 20,000+ viewers per home match on ESPN+ and began broadcasting on Cox Cable throughout SoCal</li> </ul>
	OCSC has a partnership with Rangers Football Club of Scotland, the first of its kind for USL
	<ul> <li>The Club has invested significantly in young talent, signing many U.S. Youth National Team players with the intent to develop them and participate in the \$6 billion annual global transfer market</li> </ul>
	<ul> <li>In 2021, OCSC is adding elite boys and girls Academy teams as well as a Youth Soccer Network of 12 SoCal youth clubs</li> </ul>
About OCSC Community Foundation	<ul> <li>The OCSC Community Foundation was founded in 2020 to support the Greater O.C. community</li> </ul>
	<ul> <li>Its mission is to use soccer as a vehicle to promote health and wellness, develop youth leadership, advance social change, and create positive community impact</li> </ul>
	<ul> <li>In its inaugural year, the foundation received over \$55,000 in donations, including a \$15,000 donation from the Embassy of the State of Qatar in the United States to help alleviate food insecurity brought on by the COVID-19 pandemic</li> </ul>
	<ul> <li>From teaching kids via in-person soccer clinics to packing toiletry bags for veterans in need, over 200 hours of volunteer time was donated by OCSC players and staff</li> </ul>
	<ul> <li>In 2021, the foundation plans to continue educating kids by rolling out its Virtual Soccer Curriculum to more than 50 schools in the O.C.</li> </ul>
	<ul> <li>The OCSC Foundation will utilize USSF's Mini-Pitch System<sup>™</sup> to fulfill the mutual goals of helping children embrace an active and healthy lifestyle while nurturing their personal growth beyond sports under the thematic umbrella of soccer</li> </ul>

### **BUDGET IMPLICATIONS**

### The Mini-Pitch System™



The average American family participating in youth soccer pays **\$1,472 per child** and potentially upwards of **\$5k-10k to access** more elite levels of the sport. OCSC's introduction of mini-pitches in underserved O.C. communities will create opportunities for all to participate, increasing play equity throughout the county.

### Mini-Pitches Are Funded By Different Sources Public, private or corporate donors e.g., Target, Dick's

Sporting Goods

### Musco Lighting Is An Exclusive Partner

Provides state of the art LED lighting technology solutions – mini-pitches can be used during extended hours and in darker conditions (weather permitting)

#### **Landowner Obligations Are Minimal**

Court must not be removed within the first 6-8 years Landowner is responsible for general maintenance and upkeep



Sources: Mini-Pitch Installation System ™ Brochure; The Astronomical Cost of Kids' Sports, Time.com

### **EXHIBITS (attachments):**

Attachments included:

- 1. OC Soccer Club Slide Deck Proposal Details ...... Pg.
- 2. OC Soccer Club Slide Deck Proposal Details for MSA-SA ...... Pg.
- 3. OC Soccer Club Donation Letter ..... Pg. 32

Magnolia Public Schools - Regular Board Meeting - Agenda - Thursday August 12, 2021 at 6:00 PM



# ORANGE COUNTY SOCCER CLUB

**IN PARTNERSHIP WITH** 



Powered by BoardOnTrack

230 of 258



# **Executive Summary**



Orange County Soccer Club (OCSC) is the largest and most influential professional soccer club in the region and the <u>only</u> pro soccer club within an hour's drive from the nearly 7 million people of Orange and Riverside Counties.

A mini-pitch is a small, customizable, hard-court soccer space that is ideal for providing communities a safe place to play and revitalizing underutilized urban areas and municipal properties.

The OCSC Foundation, in conjunction with the U.S. Soccer Foundation (USSF), is leading the efforts to fund up to five minipitches throughout Orange County.

Outreach is currently being undertaken by OCSC with numerous local and municipal partners for Mini-Pitch System <sup>™</sup> builds and suitable locations across Orange County.



About Orange County Soccer Club

About OCSC Community Foundation Magnolia Public Schools - Regular Board Meeting - Agenda - Thursday August 12, 2021 at 6:00 PM

 Located in Irvine, CA OCSC plays in the USL Championship, the 2<sup>nd</sup> Division of pro soccer in the U.S.



- OCSC has a partnership with Rangers Football Club of Scotland, the first of its kind for USL
- The Club has invested significantly in young talent, signing many U.S. Youth National Team
  players with the intent to develop them and participate in the \$6 billion annual global transfer
  market
- In 2021, OCSC is adding elite boys and girls Academy teams as well as a Youth Soccer Network of 12 SoCal youth clubs
- The OCSC Community Foundation was founded in 2020 to support the Greater O.C. community
- Its mission is to use soccer as a vehicle to promote health and wellness, develop youth leadership, advance social change, and create positive community impact
- In its inaugural year, the foundation received over \$55,000 in donations, including a \$15,000 donation from the Embassy of the State of Qatar in the United States to help alleviate food insecurity brought on by the COVID-19 pandemic
- From teaching kids via in-person soccer clinics to packing toiletry bags for veterans in need, over 200 hours of volunteer time was donated by OCSC players and staff
- In 2021, the foundation plans to continue educating kids by rolling out its Virtual Soccer Curriculum to more than 50 schools in the O.C.
- The OCSC Foundation will utilize USSF's Mini-Pitch System<sup>™</sup> to fulfill the mutual goals of helping children embrace an active and healthy lifestyle while nurturing their personal growth beyond sports under the thematic umbrella of soccer

Ő

# **Mini-Pitches Create Value**



With funding for a first mini-pitch in O.C. already secured from **Dick's Sporting Goods**, OCSC is working to find 3-5 additional suitable locations throughout four prioritized communities in O.C. The Club is currently focused on the following cities: **Costa Mesa, Santa Ana, San Juan Capistrano and Anaheim**.

### COMMUNITY IMPACT

✓ Physical Activity

✓ Community Engagement

✓ Health & Wellness

✓ Safe Spaces

✓ Mentorship

### OCSC PATHWAY TO PRO

- ✓ New Athletes Welcome
- ✓ OCSC Coach Access

✓ Skill Building

✓ Weekly Local Programs

# PARTNER IMPACT

✓ Signage Opportunities

Design Input

✓ Localized Programs

Tax Incentives

# ACCORDING TO IMPACT SURVEY RESULTS, COMMUNITY PARTNERS REPORTED THE FOLLOWING ABOUT THEIR MINI-PITCHES:



Source: Mini-Pitch Impact Report Data

# The American Soccer Landscape Is Primed For Growth





Total number of registered U.S. Youth Soccer athletes







Major partners currently engaged

With demand for soccer on the rise in the U.S. and Orange County, OCSC aims to secure funding for 3-5 mini-pitches per year over the next five years.



# Mini-Pitch Concept

# The Mini-Pitch System<sup>™</sup>



The average American family participating in youth soccer pays **\$1,472 per child** and potentially upwards of **\$5k-10k to access** more elite levels of the sport. OCSC's introduction of mini-pitches in underserved O.C. communities will create opportunities for all to participate, **increasing play equity** throughout the county.

### Mini-Pitches Are Funded By Different Sources

Public, private or corporate donors e.g., Target, Dick's Sporting Goods

### **Musco Lighting Is An Exclusive Partner**

Provides state of the art LED lighting technology solutions – mini-pitches can be used during extended hours and in darker conditions (weather permitting)

### Landowner Obligations Are Minimal

Court must not be removed within the first 6-8 years Landowner is responsible for general maintenance and upkeep



Sources: Mini-Pitch Installation System ™ Brochure; The Astronomical Cost of Kids' Sports, Time.com

# **Currently Seeking Opportunities in Orange County**



#### Complete System

The Mini-Pitch System" modular sports solution includes lighting, structural, and electrical components designed and engineered to work together first structural installation and reliable operation.

- + Factory-built, wired, aimed, and tested.
- · Fast, trouble-free installation.
- Rugged seliability, designed to be long-lasting in all weather conditions and atop all surfaces.
- · 10-year warranty covers all parts and lobor.

#### TLC for LED\* Lighting

Enjoy the benefits of the LED sports lighting technology of choice at wenses around the world.

- Custom optics create a more uniform distribution of light for better visibility.
- Patented glare control preserves darkness around your pitch, keeping neighbors happy.
- Instant on/off and energy efficient operation.

#### Control-Link\* Service

Your system will be supported by Musco's Control Link control and monitoring service.

- Manage your lights instantly, from anywhere, with a touch of a smart phone.
- 34/7 support from the Masco Team includes proactive monitoring and scheduling assistance.
- Got a real, live voice on the other end of the phone anytime you call Control-Link Central" sension center;



### **Mini-Pitch Site Requirements**

- Current concrete or asphalt surface cannot construct on grass or dirt
- ✓ 4,000-8,000 square feet (typically 100-120' long and 50-60' wide)
- ✓ Current surface is open, flat, free of major obstructions
- ✓ Surface is in state of good repair and has minimal cracks (<1" in width) or deviations</p>
- ✓ No planned renovations for proposed mini-pitch area for at least 6 years

### **Mini-Pitch Site Preferences**

- Targeted mini-pitch space is currently unused or underutilized – common targets are defunct or underutilized tennis/basketball courts, parking lots, etc.
- Availability for branding/logo placement on mini-pitch surface and potential windscreens
- ✓ Potential to host youth development programming, such as the USSF Soccer for Success program
- Accessible to the public for year-round use

Sources: Mini-Pitch System™ Brochure



# Mini-Pitch Partner Opportunity



# **Growing Network Of OCSC Partners**





### THE MINI-PITCH ECOSYSTEM CREATES SIGNIFICANT VALUE FOR THE COMMUNITY, THE CLUB AND ADDITIONAL FOUNDATION PARTNERS

• With OCSC's growing network of partners throughout O.C., the Club is actively seeking opportunities to build 3-5 more mini-pitches in 2021

### LONG-TERM STRATEGIES FOR COMMUNITY ENGAGEMENT

### **OCSC** Programming

- · A safe place to exercise and socialize within local communities
- A chance for youth to focus their personal development on building skills beyond the traditional large field game, akin to methods being used in Europe and South America by top club academies
- OCSC intends to provide its human and intellectual capital to give youth a variety of chances to engage with OCSC coaches and staff

### **City and Partner Network Engagement**

- USSF has an established network of partners through its Soccer for Success program, which provides ample opportunities to leverage synergies in pitch programming and other ventures
- Cities and Community Partners are welcome and encouraged to look for innovative ways to use pitches for programming and events on a regular basis

# Partnership Value Grows With Consistent Engagement

### **OCSC AIMS TO SERVE THOUSANDS OF KIDS BY 2026**

 Mini-pitches are an integral part of OCSC's plan to increase community engagement in underserved areas and focus on youth health and wellness

### LEGACY OF LASTING IMPACT

• Pitches will stand for a minimum of six years and partners benefit from both recurring and revolving viewership of ad signage



Desert West, Phoenix, AZ



Santa Fe South High School, Oklahoma City, OK





Powered by BoardOnTrack









# Appendix









**OCSC Family of Partners** 



































# MAGNOLIA SCIENCE ACADEMY





# ORANGE COUNTY SOCCER CLUB

**IN PARTNERSHIP WITH** 



Powered by BoardOnTrack













# Magnolia Science Academy Santa Ana

# Access to the Public

# Usage by the School

- → PE Classes
  - The Mini-Pitch will be used for PE during the day. A fenced area like this can be used for small group instructions.
- → Nutrition Break and Lunch
  - The Mini-Pitch can be utilized during the nutrition break and lunch to separate the student groups by age and allow them to enjoy the break with a focused activity
- → Afterschool and Weekends
  - The Mini-Pitch can be used during the afterschool program by the soccer coach to give instruction to small groups without having the need to go to the field every time
  - The Mini-Pitch can be used during the weekends and Saturday Schools as an extracurricular activity

# **Access to the Public**

- → Youth in the Community
  - There is a large of youth population in the two mobile home parks. So far we are in contact with around 30 of them which we are doing different type of events.
  - Our goal is to open the Mini-Pitch to our youth in the community on Wednesdays and Fridays between 4:30pm and 5:30pm
  - Access to Mini-Pitch will be given by request and availability only
- → Hosting Tournaments and Youth Development Programs
  - Potential tournaments, clinic, youth development programs can be planned to be hosted on weekends

# Youth in the Community



# **Other Details**

- $\rightarrow$  Who owns the Mini-Pitch:
  - Mini-Pitch is a one time donation to the school, therefor MSA Santa Ana will be the owner of the field.
  - As the owner, MSA Santa Ana will be responsible for the maintenance, upkeep, and liability
- → Liability
  - Our CharterSafe insurance already covers activities for students who are not enrolled to our school and adults who are not our staff members since we have already activities on campus for a while.

Magnolia Public Schools - Regular Board Meeting - Agenda - Thursday August 12, 2021 at 6:00 PM



PROMOTING HEALTH AND WELLNESS, DEVELOPING YOUTH LEADERSHIP. ADVANCING SOCIAL CHANGE. AND CREATING A POSITIVE COMMUNITY IMPACT.



August 9, 2021

Steven Keskinturk Principal Magnolia Science Academy - Santa Ana 2840 W First Street Santa Ana, CA 92703

Dear Steven,

The Orange County Soccer Club Foundation (OCSCF) intends to offer Magnolia Science Academy – Santa Ana (MSASA) a donation of a "mini-pitch," a designated non-grass soccer playing space, to be located outdoors on MSASA school property. The United States Soccer Foundation (USSF) <u>https://ussoccerfoundation.org/about/mission/</u> will provide the mini-pitch materials and arrange for the installation of the mini-pitch on MSASA property.

This offer is pending approval from the Magnolia Science Academy board, site approval by the United States Soccer Foundation and a successful funding approval agreement with the OCSCF.

The Orange County Soccer Club Foundation looks forward to bringing this wonderful non-profit project and a state of the art soccer pitch to the MSASA and greater Santa Ana community.

Sincerely,

Lisa Keston Executive Director, OCSC Foundation

Make a donation at orangecountysoccer.com/donations Orange County Soccer Club 20 Fairbanks Irvine, CA Magnolia Public Schools - Regular Board Meeting - Agenda - Thursday August 12, 2021 at 6:00 PM



PROMOTING HEALTH AND WELLNESS. DEVELOPING YOUTH LEADERSHIP. ADVANCING SOCIAL CHANGE. AND CREATING A POSITIVE COMMUNITY IMPACT.



### About the Orange County Soccer Club Foundation

The OCSC Community Foundation is the charitable arm of Orange County Soccer Club. We are a 501(c)(3) organization founded in 2020 to serve the greater OC community. Our mission: To use soccer as a vehicle to promote health and wellness, develop youth leadership, advance social change, and create positive community impact. Website: <a href="https://www.orangecountysoccer.com/foundation">https://www.orangecountysoccer.com/foundation</a>

### About Orange County Soccer Club

Orange County Soccer Club is Orange County's only professional soccer team. Orange County SC is a part of the United Soccer League (USL), the fastest-growing Division II professional soccer league in the world comprised of 34+ teams across the United States. The team plays their home games at the Orange County Great Park's Championship Soccer Stadium located in Irvine, CA. Nestled right in the heart of the Great Park, the Championship Soccer Stadium has a capacity of over 5,000 for Orange County SC matches. Website: <a href="https://www.orangecountysoccer.com">https://www.orangecountysoccer.com</a>

### About United States Soccer Foundation

The U.S. Soccer Foundation's programs are the national model for sports-based youth development in underserved communities. Since its founding in 1994, the Foundation has established programs proven to help children embrace an active and healthy lifestyle while nurturing their personal growth beyond sports. Its cost-effective, high-impact initiatives offer safe environments where kids and communities thrive. Headquartered in Washington, D.C., the U.S. Soccer Foundation is a 501 (c)(3) organization.

Website: https://ussoccerfoundation.org/about/

Make a donation at orangecountysoccer.com/donations Orange County Soccer Club 20 Fairbanks Irvine, CA